DOI: 10.2903/j.efsa.2024.8755

SCIENTIFIC OPINION



Vaccination of poultry against highly pathogenic avian influenza – Part 2. Surveillance and mitigation measures

EFSA Panel on Animal Health and Animal Welfare (AHAW) | European Union Reference Laboratory for Avian Influenza | Søren Saxmose Nielsen | Julio Alvarez | Dominique Joseph Bicout | Paolo Calistri | Elisabetta Canali | Julian Ashley Drewe | Bruno Garin-Bastuji | Christian Gortázar | Mette S. Herskin | Virginie Michel | Miguel Ángel Miranda Chueca | Barbara Padalino | Helen Clare Roberts | Hans Spoolder | Karl Stahl | Antonio Velarde | Arvo Viltrop | Christoph Winckler | Alessio Bortolami | Claire Guinat | Timm Harder | Arjan Stegeman | Calogero Terregino | Barbara Lanfranchi | Ludovica Preite | Inma Aznar | Alessandro Broglia | Francesca Baldinelli |

Correspondence: biohaw@efsa.europa.eu

Abstract

Selecting appropriate diagnostic methods that take account of the type of vaccine used is important when implementing a vaccination programme against highly pathogenic avian influenza (HPAI). If vaccination is effective, a decreased viral load is expected in the samples used for diagnosis, making molecular methods with high sensitivity the best choice. Although serological methods can be reasonably sensitive, they may produce results that are difficult to interpret. In addition to routine molecular monitoring, it is recommended to conduct viral isolation, genetic sequencing and phenotypic characterisation of any HPAI virus detected in vaccinated flocks to detect escape mutants early. Following emergency vaccination, various surveillance options based on virological testing of dead birds ('bucket sampling') at defined intervals were assessed to be effective for early detection of HPAIV and prove disease freedom in vaccinated populations. For ducks, virological or serological testing of live birds was assessed as an effective strategy. This surveillance could be also applied in the peri-vaccination zone on vaccinated establishments, while maintaining passive surveillance in unvaccinated chicken layers and turkeys, and weekly bucket sampling in unvaccinated ducks. To demonstrate disease freedom with > 99% confidence and to detect HPAI virus sufficiently early following preventive vaccination, monthly virological testing of all dead birds up to 15 per flock, coupled with passive surveillance in both vaccinated and unvaccinated flocks, is recommended. Reducing the sampling intervals increases the sensitivity of early detection up to 100%. To enable the safe movement of vaccinated poultry during emergency vaccination, laboratory examinations in the 72 h prior to the movement can be considered as a risk mitigation measure, in addition to clinical inspection; sampling results from existing surveillance activities carried out in these 72 h could be used. In this Opinion, several schemes are recommended to enable the safe movement of vaccinated poultry following preventive vaccination.

K E Y W O R D S

avian influenza transmission, highly pathogenic avian influenza (HPAI), poultry, surveillance, surveillance strategies

This is an open access article under the terms of the Creative Commons Attribution-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited and no modifications or adaptations are made. © 2024 European Food Safety Authority, European Union Reference Laboratory for Avian Influenza. *EFSA Journal* published by Wiley-VCH GmbH on behalf of European Food Safety Authority.

CONTENTS

Ab	stract.				1
Su	mmary	y			4
1.	Intro	oductior	יייייי ו		11
	1.1.	Backg	round an	d Terms of Reference as provided by the requestor	11
	1.2.	Interp	retation	of the Terms of Reference	11
2.	Data	and Me	ethodolo	gies	11
	2.1.	ToR 3	– Surveill	ance in the vaccinated zone and/or vaccinated establishments	11
		2.1.1.	Surveill	ance as per Regulation (EU) 2023/361 during/after vaccination	11
			2.1.1.1.	Rules for implementation of surveillance during/after emergency protective vaccination	12
			2.1.1.2.	Rules for implementation of surveillance during/after preventive vaccination	12
		2.1.2.	Surveill	ance as per WOAH	13
		2.1.3.	Surveill	ance strategies and key attributes	13
			2.1.3.1.	Surveillance strategies	14
		2.1.4.	Diagno	stic methods	19
		2.1.5.	Method	ds to assess surveillance strategies	19
			2.1.5.1.	Methodology to assess surveillance for early detection (E1) and demonstration of freedom	
				(E3, E4) following emergency vaccination	19
			2.1.5.2.	Methodology to assess surveillance for early detection (P1) and demonstration of freedom	
				(P3, P4) following preventive vaccination	25
			2.1.5.3.	Methodology to assess vaccination effectiveness and level of immune response induced by	
				vaccination (surveillance strategies E2, P2)	28
	2.2.	ToR 4	– Risk mi	tigation measures	28
		2.2.1.	Risk mi	tigation measures as per delegated regulation (EU) No 2023/361 after vaccination	28
		2.2.2.	Definiti	on of risk mitigation measures scenarios	32
	2.3.	Uncer	tainty an	alysis	32
3.	Asse	ssment			32
	3.1.	ToR 3:	Surveilla	nce	32
		3.1.1.	Diagno	stic methods	32
			3.1.1.1.	Virological methods	32
			3.1.1.2.	Molecular methods	33
			3.1.1.3.	Rapid antigen detection systems	34
			3.1.1.4.	Serological methods	34
			3.1.1.5.	Monitoring of clinical signs and production parameters	43
			3.1.1.6.	Use of unvaccinated sentinel birds	43
			3.1.1.7.	Use of environmental samples for surveillance	43
		3.1.2.	Surveill	ance in vaccinated populations	44
			3.1.2.1.	Surveillance for early detection at the establishment level for emergency vaccination (strategy F1)	44
			3.1.2.2.	Surveillance for demonstrating disease freedom following emergency vaccination (strategy	/ 51
			3.1.2.3.	Surveillance for early detection and demonstrating freedom in preventive vaccination	
				populations (strategy P1, P3, P4)	51
			3.1.2.4.	Surveillance to assess vaccination effectiveness (E2)	54
			3.1.2.5.	Surveillance to assess the level of immune response induced by vaccination (P2)	55
			3.1.2.6.	Surveillance after cessation of vaccination (duration and type)	56
			3.1.2.7.	Surveillance in vaccinated captive birds	56
	3.2.	ToR 4:	Restrictio	ons and risk mitigation measures	57
		3.2.1.	Emerge	ency vaccination	57

		3.2.2.	Preventive vaccination	58
	3.3.	Source	es of uncertainty	58
4.	Cond	lusions	and Recommendations	59
	4.1.	ToR 3:	Diagnostic methods	59
	4.2.	ToR 3:	Surveillance	61
		4.2.1.	Surveillance for early detection at establishments for emergency vaccination	61
		4.2.2.	Surveillance for early detection and demonstrating disease freedom following preventive vaccination	۱63
		4.2.3.	Surveillance for estimation of vaccination effectiveness	64
		4.2.4.	Surveillance to assess level of immune response induced by vaccination	64
		4.2.5.	Surveillance in vaccinated captive birds	65
	4.3.	ToR 4:	Restriction and risk mitigation measures	65
		4.3.1.	Emergency vaccination	65
		4.3.2.	Preventive vaccination	66
Ab	brevia	tions		66
Glo	ossary			66
Ac	knowl	edgeme	nts	67
Со	nflict o	of intere	st	67
Re	questo	or		67
Qu	estion	numbe	r	67
Со	pyrigh	it for no	n-EFSA content	67
Pai	nel me	mbers.		67
Ret	ferenc	es		67
An	nendi	×Δ		69
· ·P	Period			

SUMMARY

Background and Terms of Reference

During the past three highly pathogenic avian influenza (HPAI) epidemiological years, multiple HPAI virus subtypes have co-circulated in Europe, particularly in the autumn and winter months. The virus has been increasingly detected in migratory waterfowl birds, resulting in an increased risk of infection in poultry. Therefore, it is necessary to explore the potential of the application of vaccination of poultry in addition to the already applied measures to prevent and control HPAI infections. Given this situation, the European Commission asked EFSA to:

ToR 3 – Assess suitability and effectiveness of the reinforced surveillance set out in the Delegated Regulation (EU) No 2023/361 and of alternative surveillance strategies to be implemented upon HPAI vaccination of poultry establishments. The assessments should target, in particular, surveillance for early detection of HPAI, demonstration of freedom from HPAI infection, also considering WOAH standards of surveillance.

ToR 4 – Assess suitability of the restriction and risk mitigation measures set out in the Delegated Regulation (EU) No 2023/361 and of alternatives to be applied in vaccinated establishment and zone, with the aim of preventing the spread of disease with the movements of vaccinated poultry and their products.

Data and Methodologies

ToR 3

Methods to assess surveillance strategies

An overview of the legal framework of the Delegated Regulation (EU) No 2023/361 was provided and alternatives were defined. This regulation provides rules for the implementation of surveillance during and after emergency protective vaccination (hereafter referred to as 'emergency vaccination') and preventive vaccination. The former requires laboratory surveillance with virological testing carried out every 2 weeks and reinforced clinical surveillance; the latter requires passive surveillance with weekly virological testing of a representative sample of dead birds and active surveillance with monthly clinical and serological/virological testing. The WOAH recommendations were also considered, which require all vaccinated establishments to be tested (virological and serological) to demonstrate the absence of HPAI circulation and the effectiveness of the vaccination programme.

Four different alternative surveillance strategies for each vaccination scenario (emergency and preventive) were considered and key attributes such as the performance indicator, sampling frequency, duration, poultry population to be sampled, diagnostic methods were defined according to the objective of the surveillance, i.e. in emergency vaccination (E): early detection (E1), assessment of vaccination effectiveness (E2), demonstration of freedom from HPAI in a vaccinated establishment (E3) and demonstration of freedom from HPAI in the vaccinated zone (E4); in preventive vaccination (P): early detection (P1), assessment of level of immunity induced by vaccination (P2), demonstration of freedom in a vaccinated establishment (P3) and in the vaccinated area (P4).

Strategy E1, E3 and E4 took into account the ability to detect and remove an infected establishment before it transmits the infection to, on average, one or more other establishments. The between-flock reproduction number given surveillance (R_s) was used as a measure of transmission and the surveillance was considered effective for early detection when leading to $R_{\rm s}$ < 1. Transmission parameters and case fatality for both vaccinated and unvaccinated populations (chicken layer, duck and turkey flocks) were previously determined in Part 1 of the Scientific Opinion on vaccination of poultry against HPAI; mortality at day of reporting suspicion was calculated based on daily mortality data recorded by farmers in the Netherlands and France. Mortality thresholds that could trigger reporting suspicion by farmers were defined based on available literature and the efficacy of using such thresholds for early detection was tested in this Opinion. By employing a deterministic compartmental SEIRD model, graphical representations were provided, illustrating the expected number of infectious birds over time during an outbreak and the expected daily mortality for unvaccinated and vaccinated flocks of chicken layers, ducks and turkeys, whereas, for the assessment, a stochastic model was used. It was assumed that an infected vaccinated flock would remain undetected by passive surveillance until the infectiousness level reaches that estimated for an unvaccinated flock and, the model considered that introduction of infection can take place at any moment between two adjacent sampling events. One thousand simulations of HPAI outbreaks in vaccinated flocks were carried out and used to quantify infectiousness of a flock given detection by passive surveillance only and the expected infectiousness of a flock that escapes detection by the evaluated alternative surveillance strategy (this detection should take place before passive surveillance). The two values were then related to the expected between-flock reproduction number R_h, which was derived from the analysis of the epidemics in France, Italy and the Netherlands, made in Part 1 of the Scientific Opinion on vaccination of poultry against HPAI, to finally calculate the expected between-flock reproduction number given surveillance (R_s). R_s was defined as the reference measure to compare different surveillance strategies, encompassing different sample types (live or dead birds), sizes and frequencies at flock level. A combination of sample size and sampling interval was considered effective only if the probability to escape detection reached levels lower than 0.01 for more than 95% of the outbreak simulations and R_s < 1.

Strategies P1, P3 and **P4** were assessed using scenario tree models to estimate the sensitivity of the surveillance system to demonstrate freedom and to early detect HPAI in a preventive vaccination scenario. This analysis provides an estimation of the probability that the surveillance in place is capable of detecting the infection in the unit of interest, given a certain prior HPAIV prevalence. The tree model traces all pathways by which a single unit can be detected as infected by the system. It includes a number of risk nodes and probability nodes, and the ultimate outcome is the probability that the outbreak is detected (P_{pos}) given the sensitivity of testing at flock level (F_{se}).

Surveillance strategies E2 and **P2** were addressed descriptively. Vaccine effectiveness, as the ability of a vaccine to protect from infection under field conditions, can be assessed by observation of the occurrence of HPAI outbreaks in a country where vaccination is implemented. Ways to assess vaccination effectiveness or determine the level of immune response after vaccination were proposed, specifying the data to be gathered and potential analyses for estimation. For emergency vaccination, an assessment of the radius and of the surveillance to be done in the **peri-vaccination zone** was carried out based on data and results from a previous EFSA Opinion. The assessment of the type and duration of surveillance to be conducted **after cessation of vaccination** was based on evidence previously collected in Part 1 of the Scientific Opinion on vaccination of poultry against HPAI and qualitatively assessed for the current Opinion.

Diagnostic methods

The different diagnostic tests that can be used as part of HPAIV surveillance were described: classical virological methods (virus isolation), molecular virological methods (conventional RT-PCR and real-time RT-PCR (RT-qPCR)), rapid antigen detection systems and serological methods (agar gel immunoprecipitation (AGID)), haemagglutination inhibition (HI) and ELISA technology. For each of the described methods, a qualitative assessment, according to the objective of the surveillance, was conducted. Serological data from experimental trials conducted on ducks, turkeys and chickens performed by AI-NRLs were obtained from France (ducks), Italy (turkeys), the Netherlands (chickens) and Belgium (chickens). Those data were analysed with the aim to assess the HI titres in vaccinated birds according to the vaccine(s) used for immunisation, the HI titres in experimentally vaccinated-infected birds and the performance of the ELISA assays used for the detection of antibodies against the nucleoprotein (NP) in birds vaccinated with DIVA compatible vaccines following infection. Also, the monitoring of clinical signs and the possible use of sentinels were qualitatively assessed.

ToR 4

Risk mitigation measures

According to Delegated Regulation (EU) No 2023/361, different risk mitigation measures can be implemented following emergency and preventive vaccination. Following vaccination, vaccinated animals and their products are subject to prohibition of movement. For emergency vaccination, the competent authority may allow the movement of animals if these are subject to compulsory killing or if conditions for moving set out in Delegated Regulation (EU) No 2020/687 are fulfilled. For preventive vaccination, by way of derogation, the competent authority may allow the movement of vaccinated animals and their products according to the conditions laid down in Delegated Regulation (EU) No 2023/361. EFSA assessed qualitatively those measures and identified and assessed the provision of demonstrating freedom as an alternative or complementary approach to guarantee a safe movement of poultry and their products.

Uncertainty analysis

The uncertainty analysis was performed following the procedure detailed in the EFSA guidance on uncertainty analysis in scientific assessments. The sources of uncertainty associated with the data available and the methodology used were listed and discussed, but their impact on the assessment was not quantified.

Assessment

ToR 3

Diagnostic methods

Sterile immunity state is the ideal outcome of an efficient vaccination campaign, but it is usually not achieved on a population base. However, vaccination should elevate the threshold of an individual animal to become infected and reduce virus excretion in amplitude and time compared to unvaccinated ones. To detect smaller amounts of virus at a lower prevalence, it is therefore necessary that during active surveillance of vaccinated establishments, highly sensitive diagnostic tests tailored to the epidemiological situation are employed.

Surveillance to detect HPAIV outbreaks or to prove freedom from disease in a vaccination zone can rely on a number of classical or molecular virological methods for virus detection, as well as serological methods for the detection of antibodies indicating previous exposure, which can be used alone or in combination depending on the specific objective of the surveillance activity. Since vaccination may reduce the window of viral shedding and lower viral replication titres, the ability of the virological tests to detect viral circulation may diminish. Therefore, due to inferior sensitivity compared to molecular methods and the slower turnaround time, **virus isolation** is not recommended for screening during active surveillance in vaccinated flocks. It can only be considered as an alternative to molecular methods in exceptional circumstances (e.g. unavailability of reagents for molecular testing); it would however be essential to allow phenotypic characterisation of the virus once the outbreak has been confirmed by a molecular method.

Molecular techniques are generally used for HPAIV diagnosis directly from clinical specimens of birds. RT-qPCR offers rapid turn-around time, added cross-contamination prevention and is applicable to high throughput analysis. The outstanding analytical sensitivity of RT-qPCR also allows pooling of clinical samples. In flocks vaccinated with live vectored H5 or H7 vaccines, interpretation of RT-qPCR may be compromised, since active replication of the vaccine would be taking place in different tissues and might return positive results to assays designed to detect the H5 or H7 specific molecular signatures shared between vaccines' inserts and field viruses. To overcome this issue, specific RT-qPCR stailored to avoid such interference may be required to distinguish residual vector vaccine inserts from HPAI field virus. Analysis of environmental samples employing the same assays could be carried out, as the effort of individual animal sampling is considerable and may itself provide an obstacle to vaccination, while collection of environmental samples may provide an easier handling. At the time of writing, however, evidence that environmental sampling could replace animal-based sampling in surveillance activities to guarantee early detection of HPAI outbreak at the design prevalence of 5% with 95% confidence currently foreseen in Delegated Regulation (EU) No 2023/361 is lacking due to a deficiency of independent systematic studies.

Next-generation sequencing (**NGS**) are high speed techniques that allow generation of large volumes of data and have many applications in research and diagnostic settings. In terms of surveillance purposes, NGS has a very high value in further characterisation of isolated AI viruses.

Rapid antigen detection systems are methods for the detection of viral antigen which are usually based on enzymatic immunoassays or immunochromatography (i.e. they use a monoclonal antibody against AIV nucleoprotein or other highly conserved regions of structural proteins). They produce results very rapidly and can be carried out on the poultry premises, but they have low sensitivity and, consequently, are prone to produce false-negative results with the smaller amounts of virus expected in vaccinated flocks. Therefore, they are currently not considered useful for active surveillance in vaccinated flocks.

The different vaccine technologies exhibit variances in their ability to stimulate preferentially humoral or cell-mediated immune response. The type of vaccine used for immunisation can therefore affect character and magnitude of the specific immune response, which in turn may affect its detectability via serological methods. At present, there are no scalable methods to measure cell-mediated immunity; instead, available tests are all based on antibody detection. For example, agar gel immunodiffusion (AGID) tests can detect the presence or absence of antibodies against any influenza A viruses (since they all share antigenically similar nucleoprotein and matrix antigens) in a subtype independent manner. However, several reasons (e.g. restriction to certain species only, generation of qualitative results, low sensitivity, virus proteins not involved in the protective immunity are targeted) limit their use for surveillance purposes. Thus, ELISA technology has replaced AGID as a generic serologic screening assay. Quantitative assessment of humoral immune response still relies on haemagglutination inhibition assays (HI). Clinical examination alone is not a reliable indicator in surveillance of vaccinated flocks, as it has been demonstrated that, despite the variations in protection after vaccination, absence of clinical signs and increase in mortality are poor indicators of viral circulation. However, in field conditions, a certain number of animals may show clinical signs and there may be an increase in mortality. So, these compromised animals deserve greater attention in surveillance investigations (i.e. bucket sampling strategies). The use of unvaccinated sentinels in vaccinated flocks is highly problematic, since, for correct and successful implementation, optimal mingling and identification of sentinels and vaccines is required, which has been proven difficult to establish in practice. In addition, fully susceptible sentinels may provide a door for viral entry.

Results from analysis of **serological data** indicate that monitoring of vaccine-induced immunity can be performed by HI from 3 to 7 weeks post-vaccination, if antigens antigenically close to the vaccine are used. Similarly, to assess past infection in immunised birds, antigens antigenically close to the field HPAI strains should be used. An increase in HI titres (up to 9 log₂) have been observed 14 days post-infection when sera from vaccinated and PCR-confirmed infected animals were tested using challenge strains.

ELISA assays targeting antibodies against the nucleoprotein were evaluated in vaccine trials performed by different NRLs to assess their suitability for a DIVA testing. Results have showed that antibodies against the NP of HPAI become detectable in vaccinated and infected animals starting from 7 to 14 days post-infection in all tested species and with all vaccines compatible with such DIVA strategy. Diagnostic sensitivity and specificity of indirect ELISA assays have been found to be higher compared to those of competitive ELISA assays, and thus should be favoured if compatible with the species that has to be tested.

Surveillance

Surveillance results for **early detection** at **establishment level** in case of **emergency vaccination (E1)** are reported grouped by poultry species and different surveillance strategies (passive reporting, 'mortality thresholds' and active surveillance). Different combinations of sampling type, sampling intervals and samples sizes were assessed and those leading to a very low probability of escaping detection, were considered effective. The surveillance strategies include sampling all

dead birds (up to a number of 5, 10, 15 and 20) collected within 48 h before sample submission at different sampling intervals. Consequently, there is no minimum sample size: If fewer dead birds are present, all the dead birds available are tested. For flocks ≥ **3000 chicken layers**, molecular testing all dead birds up to a number of 5 every week proved to be effective for early detection; longer sampling intervals can be considered if higher numbers of dead birds are sampled and tested (up to a number of 10 every 2 weeks or 15 for a 3-week interval). For flocks < **3000 chicken layers**, none of the surveillance strategies assessed resulted effective. In those flocks, the level of confidence of early detection can be increased combining the test of all up to a number of 10 dead birds weekly, with molecular testing of live birds every 2 weeks.

For flocks of **2500 ducks**, the strategy of molecular testing all dead birds up to a number of 5 every week showed to be effective for early detection. The strategy of virological or serological testing 60 live birds every 2 weeks would also be effective for early detection, together with the strategy based on 'mortality thresholds'. In flocks of \geq 6000 ducks, the strategies mentioned above would be effective, together with sampling all dead birds up to 10 every 2 weeks, or up to 15 dead birds every 3 weeks.

For flocks of \geq 6000 turkeys, the strategy of molecular testing all dead birds up to a number of 5, every week resulted effective to early detect HPAIV. Longer sampling intervals can be considered if higher numbers of dead birds are collected (up to a number of 15 every 2 weeks or up to 20 for a 3-week interval). For flocks of < 6000 turkeys, the only effective surveillance strategy was based on 'mortality thresholds' and it should be carried out combined with molecular testing of live birds every 2 weeks to increase the level of confidence of early detection.

After emergency vaccination is applied and **surveillance for disease freedom (E3, E4)** implemented, there is a phase where no further outbreaks are detected. With consistently negative results at both establishment and zone level and a high probability of detection (> 99%) of infected flocks within an establishment, a high level of confidence of disease freedom would be achieved.

According to Delegated Regulation (EU) No 2023/361 **surveillance in a peri-vaccination zone**, during emergency protective vaccination, should be applied to detect any possible new outbreak. A previous risk assessment on control measures for Category A disease has been here used to assess the radius of the peri-vaccination zone. Based on it, a 10-km peri-vaccination zone radius was identified as suitable to contain the spread of the disease and avoid any jump outside the area with 95% confidence. In the peri-vaccination zone, the type of surveillance depends on the vaccination status of the establishments. In preventively vaccinated farms, the surveillance strategy for early detection in emergency vaccination is to be applied. For non-vaccinated farms, passive surveillance for layer chickens and turkeys and weekly sample of available dead birds for ducks and geese applies.

Surveillance results for **early detection** and **demonstration of freedom** in preventive vaccinated populations (**P1**, **P3**, **P4**) are assessed based on simulated scenarios in the Netherlands, France and Italy. The focus is on geographical areas previously categorised as high-risk zones for HPAIV incursion. In these zones, coexistence of vaccinated and unvaccinated populations is present. The model tested active surveillance carried out in all vaccinated flocks based on monthly collection of all dead birds up to a number of 15, collected in the 48 h before sample submission, while passive surveillance is applied in unvaccinated flocks. The results indicated that the estimated probability of disease freedom (Pfree) would exceed 99% in the three evaluated high-risk regions, given that all test results are negative. The model outcomes also showed that early detection surveillance sensitivities (EDSe) in vaccinated flocks are higher (layer chickens and turkeys) or similar (ducks) to the EDSe of passive surveillance in unvaccinated flocks of the other poultry types in the regions. Scenarios with variations in sampling intervals (30, 15 and 7 days) and different proportions of vaccinated establishments (100%, 50% and 25%) were also explored, showing that EDSe increase with shorter sampling interval and decrease with smaller proportion of vaccinated establishments sampled. In contrast, these changes have a limited effect on the probability of freedom of the zone (Pfree).

Surveillance can also be used to **assess vaccination effectiveness** (VE) **(E2)**. Implementing vaccination at the regional level enables the observation of numerous vaccinated establishments, facilitating the assessment of vaccination effectiveness and its progression in time under natural exposure conditions to improve emergency preparedness. This approach also offers the opportunity to evaluate vaccination characteristics, such as the onset of protection after vaccination and the duration of protection. The data needed to estimate VE include vaccination status of an establishment, date of vaccination, HPAI infection status, date of detection. Data should be collected and grouped at the smallest administrative area. Probabilities of infection before vaccination and at different time intervals following vaccination (e.g. 7, 14 and \geq 21 days) can be then compared. During preventive vaccination, it is expected that all establishments of a specific poultry type have been vaccinated, so no comparison can be drawn with unvaccinated ones; nevertheless, an unexpected high number of outbreaks could point at vaccination failure.

Surveillance to assess the **immune response induced by vaccination (P2)**. In the absence of virus circulation, the level of immunity at establishment level induced by vaccination can be measured using different serological assays aimed at the detection of antibodies induced by vaccination. The type of vaccine applied determines the most appropriate diagnostic test to monitor the immune response at flock level. The animals should be sampled at a precise time for the detection of a specific correlate of protection. However, the identification of a correlate of protection remains elusive. So, to assess the percentage of a vaccinated flock showing an 'effective' immune response (as an indication of 'adequate protection' and possibly of the level of herd immunity reached), a pragmatic approach was suggested. First, it should be defined the desired level of herd immunity. Then, a sampling approach has to be chosen, it could be the one adopted in the emergency vaccination plans for the control of LPAI in Italy in 2000 (currently used in France), which is based on at least 20 random samples per establishment. For chicken layers, ducks and turkeys, the level of immunity to be targeted was specified and, based on different sample sizes, it was possible to obtain different results combinations.

After cessation of vaccination, when the last vaccine has been administered and no further vaccination plans are in place, a transition is expected from vaccinated flocks to unvaccinated ones at the end of each production cycle. These unvaccinated flocks will no longer be subject to HPAI surveillance implemented after vaccination. Due to limited knowledge on protective immunity duration and on protection threshold values, determining surveillance duration post-vaccination remains inconclusive. Therefore, vaccinated flocks should be under surveillance up to their replacement, with dead birds being preferred for testing. Meanwhile, passive surveillance should be applied on unvaccinated flocks present in the vaccination area. **Surveillance in vaccinated captive birds**. A surveillance system should be in place to early detect and intercept the presence of HPAIV in vaccinated kept birds such as those in zoo, ornamental birds or birds of high value. Passive surveillance is considered the only feasible option in most cases. It could be based on a frequent (e.g. weekly) observation and clinical examination of birds. In this setting, clinical signs can be immediately detected, coupled with a prompt analysis of any sick and dead birds, including, for zoo, wild birds found dead in the zoo premises. For kept ornamental birds, certain movements (i.e. those with a high risk of HPAIV spread such as for trade, exhibitions, shows, markets, etc.) may represent a risk, and testing should be carried out before it, since the premovement clinical examination alone may not be sufficient. For holding keeping captive birds above a certain number, the surveillance suggested for poultry might apply.

ToR 4

Risk mitigation measures

According to Delegated Regulation (EU) No 2020/687, during emergency vaccination authorisation for movement of animals and hatching eggs can be granted if the competent authorities deem the risk to be negligible. This risk assessment relies on a comprehensive evaluation that includes results of clinical examination, laboratory examination (if necessary) and an official veterinary visit. As per Delegated Regulation (EU) No 2023/361, for emergency vaccination, the vaccinated establishments are already under surveillance, so laboratory examination is ongoing. According to EFSA assessment, these emergency vaccinated establishments should be under surveillance as described in Sections 3.1.2.1 and 3.1.2.2 (Tables 10A-12B). Therefore, including the effectiveness of the combined clinical examination and surveillance established in this Opinion, the risk of movement of animals and eggs is considered negligible when both are performed within the 72 h before the movement. If the process of movement (e.g. loading) is not completed within 72 h, then testing should be repeated every 48 h until it ends. Given the already ongoing surveillance, testing could coincide with the sampling session of the surveillance in place after emergency vaccination (i.e. no additional sampling and testing should be done). Note that, differently from what stated in Delegated Regulation (EU) No 2020/687, suggesting that laboratory examinations should be conducted only if necessary, laboratory examinations have been assessed as an integral component of the risk mitigation measures to be implemented in case of emergency vaccination. For preventive vaccination, according to Delegated Regulation (EU) No 2023/361, safe movements of vaccinated birds and their products are guaranteed if the vaccinated animals were negative to the active and enhanced passive surveillance implemented after preventive vaccination against HPAI (Table 1). In addition, the implementation of other mitigation measures further contributes to this guarantee (see Figure 7). In the current Opinion, the surveillance to be applied after preventive vaccination is assessed in Section 3.1.2.3 and prescribes the monthly sampling of dead birds with a number of 15 collected in a flock during 48 h before the testing to assess a zone as free from HPAIV. Therefore, in the case of preventive vaccination, where no outbreaks are present, the surveillance as described in Section 3.1.2.3, that is already in place, provides a very high certainty that the area is free from infection (see Tables 15A–17B). Accordingly, if the birds are negative to the clinical inspection done not earlier than 72h before the movement and all establishments follow the surveillance protocol with no outbreak detection, the risk of HPAI spread via movement of such birds is considered negligible. If the establishment from where the vaccinated birds are to be moved is not under surveillance, such as in case

Sources of uncertainty

The current Opinion identifies sources of uncertainty across various assessment sections, including performance of diagnostic tests used in the context of HPAIV surveillance, and the effectiveness of surveillance strategies at the establishment level and of surveillance strategies at the area/zone level.

where less than 100% of vaccinated establishments are subjected to testing (such as 50% or 25%), then vaccinated birds are clinically inspected and dead birds up to a number of 15, should be collected not earlier than 72 h before the move-

ment. Negative test results should be provided before the movement and no clinical signs should be observed.

Conclusions and Recommendations

ToR 3

Diagnostic methods

There are a number of different diagnostic methods available that can be used according to vaccine type, strategy and surveillance objective, but lack of validation and field experience in the context of vaccinated poultry populations

of these methods may be an issue. Serology may increase confidence of virological surveillance results but, the ability to use serological tests may be compromised when vaccination is carried out with non-DIVA compatible vaccines. Samples collected from the environment could be a valid, time- and labour-saving alternative to animal samples, but there is currently a lack of validated protocols; research to extend and deepen knowledge here is needed. When implementing a vaccination programme, it is important that the most appropriate diagnostic method for surveillance is selected according to vaccine type; methods with high sensitivity are recommended when expected viral levels to be detected are low due to vaccination. In addition to routine molecular monitoring, viral isolation, genetic sequencing and phenotypic characterisation of HPAIVs eventually detected in vaccinated flocks are highly recommended to detect putative escape mutants as early as possible.

Surveillance

For the surveillance, some important **general considerations** and **conclusions** were drawn: The surveillance strategies modelled included the random sampling of the dead birds up to a number of 5, 10, 15 or 20 (increasing the sample size above this target does not increase the probability of virus detection); if fewer dead birds are present, then all the available ones should be tested, given that the probability of early detecting HPAIV is higher than testing live birds; to achieve the highest sensitivity, birds intended for testing, are those found dead in the last 48 h; the effectiveness of the above-mentioned approach is maximised by repeated sampling in time; the lack of knowledge regarding the correlate of protection and, consequently, the duration of protective immunity in vaccinated poultry, precludes any definitive conclusion on the surveillance duration following the cessation of vaccination. For this reason, vaccinated flocks should be under surveillance upon their replacement (sampling and testing of dead birds) and passive surveillance should be carried out in both unvaccinated and vaccinated flocks.

For **early detection** at **establishments in case of emergency vaccinations (E1)**, different options are proposed according to poultry species. For **chicken layers**, a number of effective options based on molecular testing of dead animals at a certain interval have been identified and should be selected according to country's specific circumstances and resources. For both flock sizes considered (\geq **3000** and **< 3000**), the surveillance strategy identified in Delegated Regulation (EU) No 2023/361 (virological testing of 60 live birds) was assessed as not effective, as well as the serological testing and the strategies based on passive reporting and 'mortality threshold'. However, for flocks **< 3000** birds, combining the molecular testing of dead birds with molecular testing of live birds would increase the level of confidence of early detection.

For **ducks**, a number of options based on the molecular testing of dead birds at a certain interval have been identified as effective and should be selected according to country's specific circumstances and resources. Other surveillance strategies can also be carried out (molecular or serological testing of 60 live ducks and 'mortality thresholds'), but they may present more challenges (both for welfare and logistical reasons) and their effectiveness may depend on various factors (availability and accuracy of information about mortality thresholds). The only strategy assessed as not effective for ducks is the one relying solely on passive reporting.

For flocks of \geq 6000 turkeys, a number of effective options based on the molecular testing of dead birds at a certain interval have been identified and should be selected according to country's specific circumstances and resources. The surveillance strategy identified in Delegated Regulation (EU) No 2023/361 (virological testing of 60 live birds) was assessed as not effective, as well as the serological testing and the strategies based exclusively on passive reporting and 'mortality thresholds'. However, for flocks < 6000 turkeys, the only valid strategy identified as effective was the one based on 'mortality thresholds' (0.21%), that should be combined with molecular testing of live birds to increase the level of confidence of early detection.

For the **surveillance in peri-vaccination zone**, it is recommended to establish a 10-km radius zone to increase the confidence that the probability of jump spread is lower than 0.04. In relation to the type of surveillance, it is recommended to apply either early detection surveillance in vaccinated establishments and passive surveillance for layer chickens and turkeys and bucket sampling for ducks and geese in unvaccinated farms.

For **disease freedom** at establishment and zone following **emergency vaccination** (E3, E4), the testing schemes as reported above by poultry species apply, given the sampling is repeated in time.

For the **surveillance to demonstrate disease freedom** following **preventive vaccination**, in the three high-risk zones assessed, it is recommended to carry out monthly molecular testing of all dead birds up to a number of 15, in all vaccinated establishments. Passive surveillance is to be maintained in both vaccinated and unvaccinated flocks. This testing scheme is also applied to **early detect** infections following **preventive vaccination**, but if the aim is to increase the early detection surveillance sensitivities (EDSe), then it is recommended to reduce the sampling intervals. Reducing the proportion of vaccinated establishments to be tested showed a limited impact on the Pfree, but a negative impact on the EDSe. However, for each of the targeted species, a number of effective options (maintaining high level of EDSe and Pfree) have been identified, testing lower proportions of establishments (25% or 50%) at different sampling intervals. These testing schemes should be selected according to country's specific circumstances and resources.

To estimate **vaccination effectiveness (E2)**, it is required a careful planning, sufficient human resources, follow-up of all flocks and a good data collection framework.

Surveillance to assess the **level of immune response induced by vaccination (P2)** can be performed when virus circulation is absent or expected to be low. To best detect how many birds have a measurable immune response, testing within a certain time window, depending on the vaccine used, is recommended. Since it is important to have accurate estimates of the minimum proportion of protected birds required to stop transmission, further studies quantifying R_0 values for the currently circulating HPAIV are recommended. Even if there is no established correlate of protection, an assessment of the prevalence of birds showing an immune response following vaccination is assumed to provide an indication of sufficient protection (for chickens, 20 blood samples per establishment, using appropriate diagnostic test, are appropriate to determine whether vaccines are correctly used in the field).

ToR 4

Restriction and risk mitigation measures

In case of **emergency vaccination**, HPAI outbreaks are present in the area, and vaccinated establishments are already under the surveillance applied after emergency vaccination (i.e. laboratory examination is already ongoing). In addition to ensuring negligible risk as set out in Delegated Regulation (EU) No 2020/687, the laboratory examination to demonstrate freedom from HPAI, conducted according to Sections 3.1.2.1 and 3.1.2.2, is considered as part of the risk mitigation measures to be implemented in addition to the clinical inspection in the 72 h before the movement. Negative test results should be provided before the movement and no clinical signs should be observed. Testing to guarantee safe movement of the birds could coincide with the sampling session of the surveillance in place (i.e. no additional birds/ samples to be tested).

In case of **preventive vaccination**, where no outbreaks are present, vaccinated establishments are already under the surveillance applied after preventive vaccination. To guarantee safe movement of vaccinated poultry or the day-old chickens and hatching eggs deriving from such poultry, negative surveillance test results, conducted according to Section 3.1.2.3, should be provided, and no clinical signs should be observed before the movement. If the vaccinated establishment is not under surveillance, such as in case where less than 100% of them are subjected to testing, then, in addition to the absence of clinical signs, also negative test results should be provided before the movement.

1 | INTRODUCTION

1.1 | Background and Terms of Reference as provided by the requestor

The background and the Terms of Reference (ToR) as provided by the requestor are reported in Section 1.1 of Part 1 of the Scientific Opinion on Vaccination of poultry against highly pathogenic avian influenza (HPAI) (EFSA AHAW Panel, 2023). This Scientific Opinion focuses on ToR 3 and ToR 4, which are reported in Appendix A.

1.2 | Interpretation of the Terms of Reference

The interpretation of ToRs 3 and 4 is reported in Sections 1.2.3 and 1.2.4 of Part 1 of the Scientific Opinion on Vaccination of poultry against highly pathogenic avian influenza (EFSA AHAW Panel, 2023).

2 | DATA AND METHODOLOGIES

The methodological approach adopted to address the ToRs 3 and 4 is described in the protocol reported in Appendix A of part 1 of the Scientific Opinion on Vaccination of poultry against HPAI (EFSA AHAW Panel, 2023). This protocol was developed upfront of the initiation of the risk assessment. In this section, a more detailed description of the specific data and methodology used for ToRs 3 and 4 is provided. A mapping of ToRs 3 and 4 as received from the European Commission and how they were addressed in the current Opinion is presented in Appendix A.

2.1 | ToR 3 – Surveillance in the vaccinated zone and/or vaccinated establishments

ToR 3 entails the assessment of different surveillance strategies (or surveillance approaches as mentioned in Table A1 APRIO elements for the assessment question and subquestions) to be implemented on poultry establishments vaccinated against HPAI. Vaccination scenarios include emergency protective and preventive vaccination and have been presented in Part 1 of the Scientific Opinion on Vaccination of poultry against HPAI (EFSA AHAW Panel, 2023). The surveillance assessed in this Opinion is the reinforced surveillance laid down in Delegated Regulation (EU) No 2023/361¹ (see Section 2.1.1); those described in the WOAH surveillance requirements (see Section 2.1.2); and alternative surveillance strategies as suggested by the EFSA Working Group (see Section 2.1.3).

Particularly, the current assessment was targeted at the surveillance objectives indicated in ToR 3 and in Delegated Regulation (EU) No 2023/361, namely:

- 1. Early detection of HPAIV infection in vaccinated establishments (ToRs 3.1, 3.2, Delegated Regulation (EU) No 2023/361 Article 9 and Part 5.2.b).
- 2. Demonstration of freedom from infection with HPAIV in vaccinated establishments and/or the vaccinated zone in order to guarantee movement of vaccinated and non-vaccinated animals and animal products within and from the vaccination zone during and after cessation of vaccination (ToR 3.1, 3.2); freedom from HPAIV in a vaccinated zone should be also demonstrated considering specifically the WOAH standards of surveillance (i.e. in all vaccinated establishments as provided in Chapter 10.4, Article 10.4.28) and possible alternatives to that (ToR 3.3).
- 3. Assessment of vaccination effectiveness (Delegated Regulation (EU) No 2023/361 Article 9).

The suitability and effectiveness of these surveillance strategies were assessed using mathematical models (see Sections 2.1.5 and 3.1.2) and the suitable diagnostic methods to be used according to each objective were described (Section 2.1.4) and assessed (Section 3.1.1). As stated in Opinion Part 1, backyard poultry were not considered due to data limitations. Also, previous studies in the Netherlands and France have shown that captive birds and backyard holdings likely played a limited role in disease spread (Bavinck et al., 2009; Souvestre et al., 2019).

2.1.1 | Surveillance as per Regulation (EU) 2023/361 during/after vaccination

To define surveillance strategies for the assessment, the Working Group initially provided an overview of the legal framework. In doing so, they considered Delegated Regulation (EU) No 2023/361, which sets rules for reinforced surveillance in the vaccination zones established as detailed in Art. 9 and in Annex XIII Part 2, Part 4 and Part 5.

¹Commission Delegated Regulation (EU) No 2023/361 of 28 November 2022 supplementing Regulation (EU) 2016/429 of the European Parliament and the Council as regards rules for the use of certain veterinary medicinal products for the purpose of prevention and control of certain listed diseases. *OJ L 52, 20.2.2023, pp. 1–42.*

2.1.1.1 | Rules for implementation of surveillance during/after emergency protective vaccination

According to Article 9, Delegated Regulation (EU) No 2023/361, when applying emergency protective vaccination, reinforced clinical and laboratory surveillance should be implemented in both the vaccination and peri-vaccination zones established. The objectives of such surveillance will be to assess **vaccination effectiveness** in the vaccination zone, and to **detect any possible new outbreaks** in the vaccination and peri-vaccination zones. According to Part 2, in all the vaccinated establishments, virological testing should be implemented every 2 weeks, and the sampling has to enable the detection of a minimum within-flock prevalence of HPAIV infection of 5% with a confidence level of 95% (Table 1). These measures should remain in place during the whole recovery period following emergency protective vaccination (as per Part 4). The recovery period lasts 28 days where vaccination is carried out in a zone that does not overlap with restricted zones due to HPAI outbreaks; where the vaccination zone overlaps with HPAI restricted zones, the recovery period ends when the restricted zones are lifted, if the date of lifting is later than 28 days after the last vaccine inoculation.² After the end of the recovery period, the rules for implementation of surveillance during/after preventive vaccination apply as long as an establishment keeps vaccinated birds.

Sampling methods of animals for laboratory examination are outlined in Annex I of Delegated Regulation (EU) No 2020/687³ and include the following requirements:

- Animals should be chosen randomly in a number large enough to detect the disease if present;
- Samples should be collected at random in each epidemiological unit of the establishment;
- Disease profile and level of exposure risk should be taken into account;
- Minimum number of animals to sample will depend on expected prevalence, level of confidence and performance of tests used.

The *diagnostic methods* chosen should aim at maximising surveillance sensitivity and may include using laboratory examinations to assess previous exposure to the disease.

2.1.1.2 | Rules for implementation of surveillance during/after preventive vaccination

For preventive vaccination, both enhanced passive and active surveillance should be implemented to early **detect occurrence of infection with HPAI**. Enhanced passive surveillance foresees weekly virological testing of a representative sample of dead birds collected within 1 week; according to the Diagnostic Manual for Avian Influenza (2006/437/EC),⁴ it is recommended to pool sampling of dead birds (with a maximum of 20) over the week equally subdivided from each flock⁵ of the establishment⁶: up to five carcasses/samples of birds deceased on the day of sampling should be collected by flock. Active surveillance should be carried out, by monthly clinical examination (clinical examination of poultry, check of production records, check of health records of each epidemiological unit⁷), and by monthly collection of representative samples from live birds for laboratory surveillance by virological and serological testing to enable detection of a minimum withinflock prevalence of HPAIV infection of 5% with 95% confidence level (Table 1). These measures will remain in place as long as the establishments keep vaccinated animals.

Requirements for *sampling of animals for laboratory examinations* and the *diagnostic methods* for surveillance for preventive vaccination against HPAIV in poultry are the same as in emergency vaccination and are summarised in Section 2.1.1.1.

²According to Annex XIII, Part 4, Delegated Regulation (EU) No 2023/361.

⁴https://eur-lex.europa.eu/legal-content/EN/TXT/PDF/?uri=CELEX:32006D0437&from=EN.

³Commission Delegated Regulation (EU) No 2020/687 of 17 December 2019 supplementing Regulation (EU) 2016/429 of the European Parliament and the Council, as regards rules for the prevention and control of certain listed diseases. OJ L 174, 3.6.2020, pp. 64–139.

⁵/Flock' means all poultry or captive birds of the same health status kept on the same premises or in the same enclosure and constituting a single epidemiological unit; in housed poultry includes all birds sharing the same airspace, Delegated Regulation (EU) No 2020/689.

⁶'Establishment' means any premises, structure, or, in the case of open-air farming, any environment or place, where animals or germinal products are kept, on a temporary or permanent basis, Regulation (EU) No 2016/429 on transmissible animal diseases ('Animal Health Law' – AHL).

⁷/Epidemiological unit' means a group of animals with the same likelihood of exposure to a disease agent, Regulation (EU) No 2016/429 on transmissible animal diseases ('Animal Health Law' – AHL).

 TABLE 1
 Surveillance activities according to Delegated Regulation (EU) No 2023/361 during/after implementing vaccination against HPAIV in poultry.

	Surveillance										
Type of vaccination	Surveillance category	Testing procedure	Frequency	Minimum detectable prevalence/type of information collected	Duration						
Emergency protective	Reinforced laboratory	Virological	2 weeks	5% prevalence with 95% confidence level ^a	According to the duration of the recovery period						
	Reinforced clinical	-	-	-	-						
Preventive	Enhanced passive	Virological	Weekly	Representative sample of dead birds	As long as there are vaccinated birds in						
	Active	Clinical examination	30 days	Clinical examination of poultry, check of production records, check of health records of each epidemiological unit	the establishment						
		Serological or virological	30 days	5% prevalence with 95% confidence level (representative sample) ^a							

^aTo detect HPAIV with 5% design prevalence with 95% confidence level, a sample size of 60 randomly selected live birds is required to be tested.

2.1.2 | Surveillance as per WOAH

To define surveillance strategies for the assessment, the Working Group also considered the WOAH recommendations for surveillance systems, in particular Chapters 1.4 and 10.4 of the Terrestrial Animal Code on Animal Health Surveillance and on surveillance for HPAI, respectively (WOAH, online-a).

Considering the diversity of situations regarding HPAI in the world, the WOAH approach is to provide generic recommendations on the approach and parameters to be considered when designing and implementing a surveillance system for early detection and for demonstration of freedom from infection with HPAIV; those principles are also necessary to support vaccination programmes. The WOAH does not have a standard operating procedure for official recognition of a country's status for HPAI, but Members Countries have the possibility to submit self-declaration of freedom from disease, under its sole responsibility and according to the conditions listed in Article 10.4.2 (Chapter 10.4) of the Terrestrial Code (WOAH, online-a). Self-declaration should include details of the surveillance system in place to sustain the health status. For areas that need to recover their freedom status following infection, the absence of infection should be demonstrated by the surveillance programme for a minimum period of 28 days starting after cleaning and disinfection of the last affected establishment if a stamping out policy is implemented. If the stamping out policy is not implemented, the absence of infection in poultry during a period of 12 months should be demonstrated by the surveillance programme.

In zones where vaccination is carried out, surveillance for demonstration of freedom from HPAI requires that all vaccinated flocks are tested to prove the absence of viral circulation, with a frequency that is proportional to the risk in that zone. Evidence of the effectiveness of the vaccination programme is also part of the requirements for the surveillance programme.

2.1.3 | Surveillance strategies and key attributes

To effectively define and compare surveillance strategies for the assessment, the following key attributes were considered:

- **Objective of surveillance**: (i) The early detection of HPAIV infection in a vaccinated establishment and in the vaccination/peri-vaccination zone; (ii) substantiating freedom from disease in an establishment/area in order to grant derogation to movements to animals and animal products; (iii) assessing vaccination effectiveness.
- Type of surveillance can be either defined as 'active' or 'passive' (Hoinville et al., 2013). Passive surveillance means reporting of suspect cases raised by increased mortality, other signs of serious disease or significantly reduced production rates. The health authorities are notified by veterinarians or farmers at their discretion. It is more likely to be useful for diseases that are characterised by typical clinical signs and for which subclinical infection is rare (Doherr & Audigé, 2001). Active surveillance is the detection of the disease through activities instigated following a plan. These activities can be carried out randomly or non-randomly and they can be based on clinical, serological or virological detection of disease or infection.
- **Performance indicator** describes concrete and measurable information serving as a measurable benchmark for assessing the effectiveness of a surveillance activity concerning the objective, i.e. for early detection that would be the time from virus introduction until detection of the disease, for freedom from disease that would be the confidence of (non) detecting the disease if it was present at a specified level.

- **Sampling frequency**: It characterises the surveillance activities i.e. how often samples are collected., being the sampling or collection of data carried out on a continuous basis (data are gathered continuously), on a regular basis (data are gathered at defined time intervals), or ad hoc (undertaken occasionally, when the need arises).
- **Duration and period** of surveillance refer to the period of time and the period of the year (e.g. season), in which the surveillance activities are carried out in order to meet the objective of the strategy.
- **Poultry population to be sampled** is the animal population from which the samples are collected; it refers both to the vaccinated establishments and the birds within each establishment and/or each flock within the establishment, considering species and production type, and immunity status dynamics, and the criteria for its selection (census, random, targeted, based on the risk, etc.).
- **Design prevalence** also known as the minimum detectable prevalence or target prevalence, it is defined as a threshold prevalence value associated with a statistical risk of error, at the animal or herd level, to test the hypothesis that disease is present in a population of interest and it is used to determine the needed sample size. It is the hypothetical proportion of animals/herds that are infected given the disease was present and it is used for evaluating the sensitivity of the surveillance systems. The error represents the probability of not detecting the disease among the sampled animals/herds if it was present at the minimum value set by the design prevalence.
- **Diagnostic methods**: It refers to the most suitable method with appropriate diagnostic tools to diagnose rapidly and accurately the disease or to detect the virus circulation in a population where evidence of those conditions (i.e. clinical signs and virus shedding) is modulated by the effectiveness of vaccination implemented and the field strain circulation. An overview of the benefits and limitations of diagnostic methods for HPAI is extensively provided in Section 3.1.1.
- **Epidemiological unit**: It refers to the establishment (any premises where animals are kept in a temporary or permanent basis); an establishment can comprise one or more flocks and if one flock is infected then the whole establishment is considered infected; for the number of birds to be sampled within an establishment, it always refers to the flock (all poultry of the same health status kept in the same enclosure (house/barn)).
- Sampling unit: flock(s) within the establishment

All those characteristics are described in Table 2 for both situations of emergency protective vaccination and preventive vaccination.

2.1.3.1 | Surveillance strategies

Table 2 provides an overview of the attributes associated with the different surveillance strategies considered in the assessment. The strategies encompass those specified in Delegated Regulation (EU) No 2023/361 as well as alternative surveillance strategies proposed by the Working Group and are separated by emergency protective (E) and preventive (P) vaccination scenarios. These alternative strategies explore variations in the sampled population, including differences in sample number (reduced vs. increased) or bird categories (dead vs. live), in sampling frequency (reduced vs. increased) and diagnostic methods while keeping the surveillance's objectives as they are in Delegated Regulation (EU) No 2023/361 and WOAH standard, with exception of objective in strategy P2 (assessment of level of immune response to vaccination) that was proposed by the Working Group (WG). The strategies are compared based on a performance indicator, which depends on the objective of the surveillance. An explanation of how each performance indicator is estimated is provided in Sections 2.1.5.1 and 2.1.5.2. For the surveillance set out in Delegated Regulation (EU) No 2023/361, for emergency vaccination, as the aim is to detect 5% prevalence and the animals to be sampled, according to Regulation as reported in Annex XIII part 2, should be chosen randomly in a number large enough, it is here interpreted as the sampling of live birds with a number of 60.

considered

Surveillance	Emergency protectiv	ve vaccination scenario	o – Surveillance within the va	Preventive vaccination scenario – Surveillance within vaccination zone				
strategy	Strategy E1	Strategy E2	Strategy E3	Strategy E4	Strategy P1	Strategy P2	Strategy P3	Strategy P4
Objective of surveillance	HPAIV early detection (to be implemented also in the peri-vaccination zone)	Assessment of vaccination effectiveness	Demonstrating freedom from HPAIV in the vaccinated establishment (to authorise the movement of birds from that establishment)	Demonstrating freedom from HPAIV in the vaccinated zone	Early detection in case of HPAIV introduction	Assessment of level of immune response induced by vaccination	Demonstrating freedom from HPAIV in the vaccinated establishment (to authorise the movement of birds from that establishment)	Demonstrating freedom from HPAIV in the vaccinated area (considering that also non- vaccinated establishments might be present)
Performance indicator	Between-flock reproduction number (R_h), i.e. the average number of farms infected by one infected farm in its infectious period. The surveillance was considered to be effective for early detection if it contributed to $R_h < 1$	Vaccination effectiveness, estimated as 1 minus a measure of relative risk (RR) such as: incidence or rates of infection, prevalence ratio (attack rates, cumulative incidence), the odds ratio (OR) or the hazard ratio (HR)	Probability of freedom Conditions required: no virus-positive samples	Probability of freedom within and between-flock/ establishment Required: no virus- positive flocks	Surveillance sensitivity for early detection in the vaccinated population	% of animals showing immune response following vaccination	Confidence of freedom	Confidence of freedom
Establishments to be sampled	Regulation (EU) 2023/361: all vaccinated establishments Alternative: Since this is emergency surveillance, no alternative strategy at this level is	Regulation (EU) 2023/361: all farms in the vaccination and peri-vaccination zones	Regulation (EU) 2023/361: all vaccinated establishments Alternative: since this is emergency surveillance, no alternative strategy at this level is considered	Regulation (EU) 2023/361: all vaccinated establishments	Regulation (EU) 2023/361: all vaccinated establishments Alternative: a sample of vaccinated establishments	All vaccinated establishments that completed the vaccination scheme	Regulation (EU) 2023/361: all vaccinated establishments	Regulation (EU) 2023/361: all vaccinated establishments Alternative: a sample of vaccinated establishments

TABLE 2 Attributes of the different surveillance strategies considered in the assessment described by emergency protective (E) and preventive vaccination (P) scenarios.

(Continues)

15 of 70

183

1732, 202

ş

on [15/11/2024]. See

app

TABLE 2 (Continued)

Surveillance	Emergency protecti	ve vaccination scer	nario – Surveillance within the va	accination zone	Preventive vaccination scenario – Surveillance within vaccination zone						
strategy	Strategy E1	Strategy E2	Strategy E3	Strategy E4	Strategy P1	Strategy P2	Strategy P3	Strategy P4			
Surveillance strategy S Birds to be sampled A	Regulation (EU) 2023/361: a representative sample of live birds to detect a design prevalence of 5% with 95% CL ^a Alternative: assessing different sample sizes based on sampling dead or live birds (see details in Section 3.1.2.1)	NR	Regulation (EU) 2023/361: a representative sample of live birds to detect a design prevalence of 5% with 95% CL ^a Alternative: assessing different sample sizes based on sampling dead or live birds (see details in Section 3.1.2.2)	Regulation (EU) 2023/361: a representative sample of live birds to detect a design prevalence of 5% with 95% CL ^a Alternative: assessing different sample sizes based on sampling dead or live birds (see details in Section 3.1.2.2)	Regulation (EU) 2023/361: representative sample of dead birds and a representative sample of live birds to detect design prevalence of 5% with 95% CL ^a Alternative: Assessing different sample sizes of dead or live birds to enable both high within flock sensitivity for early detection and high level of confidence in disease freedom (see details in Section 3.1.2.3)	≥ 20 randomly selected birds per establishment (to estimate the percentage of birds developing an immune response above predefined- thresholds)	Regulation (EU) 2023/361: representative sample of dead birds and a representative sample of live birds to detect design prevalence of 5% with 95% CL ^a Alternative: Assessing different sample sizes of dead or live birds to enable both high within flock sensitivity for early detection and high level of confidence in disease freedom (see details in Section 3.1.2.3)	Regulation (EU) 2023/361: representative sample of dead birds and a representative sample of live birds to detect design prevalence of 5% with 95% CL ^a Alternative: Assessing different sample sizes of dead or live birds to enable both high within flock sensitivity for early detection and high level of confidence in disease freedom (see details in Section 3.1.2.3)			

TABLE 2 (Continued)

Surveillance	Emergency protecti	ive vaccination scenario	o – Surveillance within the va	Preventive vaccination scenario – Surveillance within vaccination zone					
strategy	Strategy E1	Strategy E2	Strategy E3	Strategy E4	Strategy P1	Strategy P2	Strategy P3	Strategy P4	
Diagnostic methods	Regulation (EU) 2023/361: clinical examination and virological testing in live birds Alternative: molecular virological testing on live or dead birds (i.e. real time RT- PCR targeting M gene for screening. H5/ H7 subtyping by real-time RT-PCR or sequencing methods); Virus isolation followed by H5/ H7 subtyping by classical or molecular virological methods	Regulation (EU) 2023/361: clinical examination and virological testing in live birds Alternative: any virological method suitable for the detection and identification of HPAI infection in vaccinated farms	Regulation (EU) 2023/361: clinical examination and virological testing in live birds Alternative: molecular virological testing on live or dead birds (i.e. real time RT-PCR targeting M gene for screening. H5/H7 subtyping by real-time RT-PCR or sequencing methods); Virus isolation followed by H5/H7 subtyping by classical or molecular virological methods	Regulation (EU) 2023/361: clinical examination and virological testing in live birds Alternative: a combination of molecular virological (i.e. real time RT-PCRs) and serological tests such as NP-based ELISAs (if DIVA compatible vaccines are used). For NP- ELISA-positive and PCR- negative birds, HI with selected HPAI and LPAI antigens; real-time RT-qPCRs only (if inactivated vaccines are used or when the epidemiological situation makes impractical the use of serological tests)	Regulation (EU) 2023/361: clinical examination and virological testing of dead or live birds and serological testing in live birds Alternative: molecular virological testing by real-time RT- PCR targeting M gene for screening, confirmation of subtype and pathotype on positive samples at screening; Virus isolation followed by H5/ H7 subtyping by classical or molecular virological methods	Alternative: serological testing in live birds (e.g. H5-HI or H5-ELISA); molecular virological if an RT-qPCR tailored to the specific vaccine product is available	Regulation (EU) 2023/361: clinical examination and virological testing of dead or live birds and serological testing in live birds Alternative: molecular virological testing in live or dead birds (i.e. real time RT- PCR targeting M gene for screening, confirmation of subtype and pathotype on positive samples at screening); Virus isolation followed by H5/ H7 subtyping by classical or molecular virological methods	Regulation (EU) 2023/361: clinical examination and virological testing of dead or live birds and serological testing in live birds Alternative: a combination of molecular virological (i.e. real time RT-PCRs) and serological tests such as NP- based ELISAs (if DIVA compatible vaccines are used). For NP- ELISA-positive and PCR- negative birds, HI with selected HPAI and LPAI antigens; real-time RT-qPCRs only (if inactivated vaccines are used or when the epidemiological situation makes impractical the use of	

17 of 70

183

1732, 202

by

iley Onl

on [15/11/2024]

(Continues)

serological tests)

.8755 by C

Wiley Online Library on [15/11/2024]

TABLE 2 (Continued)

Sumoillance	Emergency protecti	ve vaccination scenario	o – Surveillance within the va	Preventive vaccination scenario – Surveillance within vaccination zone				
strategy	Strategy E1	Strategy E2	Strategy E3	Strategy E4	Strategy P1 Strategy P2		Strategy P3 Strategy P4	
Sampling frequency	Regulation (EU) 2023/361: every 14 days Alternative tested: 7, 14, 21 or 30 days	Regulation (EU) 2023/361: every 14 days Alternative tested: 7, 14, 21 or 30 days	Regulation (EU) 2023/361: every 14 days Alternative tested: Use of cumulative evidence from repeated sampling	Regulation (EU) 2023/361: 14 days Alternative tested: Use of cumulative evidence from repeated sampling	Regulation (EU) 2023/361: weekly virological testing of dead birds and monthly clinical examination and testing of live birds (virological or serological) Alternative tested: sampling frequency of 7, 14, 21 and 30 days	To be done once (from 2 to 4 weeks after completion of immunisation of after 7 weeks for vectored vaccines administered at 1 day of age)	Regulation (EU) 2023/361: weekly virological testing of dead birds and monthly clinical examination and testing of live birds (virological or serological) Alternative tested: sampling frequency of 7, 14, 21 and 30 days	Regulation (EU) 2023/361: weekly virological testing of dead birds and monthly clinical examination and testing of live birds (virological or serological) Alternative tested: sampling frequency of 7, 14, 21 and 30 days
Duration and period of surveillance	Regulation (EU) 2023/361: for the duration of the recovery period, after the surveillance for preventive vaccination applies Alternative: for the duration of the recovery period, after the surveillance for preventive vaccination applies (i.e. as long as vaccinated animals in the establishment have protective immunity)		Regulation (EU) 2023/361: for the duration of the recovery period, after the surveillance for preventive vaccination applies Alternative: for the duration of the recovery period, after the surveillance for preventive vaccination applies (i.e. as long as vaccinated animals in the establishment have protective immunity)	Regulation (EU) 2023/361: for the duration of the recovery period, after the surveillance for preventive vaccination applies Alternative: For the duration of the recovery period, after the surveillance for preventive vaccination applies (i.e. as long as vaccinated animals in the establishment have protective immunity)	Regulation (EU) 2023/361: as long as there are vaccinated animals in the establishments Alternatives: as long as vaccinated animals in the establishment have protective immunity (after that they should be under normal surveillance for non-vaccinated flocks)	As long as vaccination is performed	Regulation (EU) 2023/361: as long as there are vaccinated animals in the establishments Alternatives: as long as vaccinated animals in the establishment have protective immunity (after that they should be under normal surveillance for non-vaccinated flocks)	Regulation (EU) 2023/361: as long as there are vaccinated animals in the establishments Alternatives: as long as vaccinated animals in the establishment have detectable protective immunity (after that they should be under normal surveillance for non-vaccinated flocks)

Abbreviation: NR, not relevant.

^aAs the animals to be sampled, according to Delegated Regulation (EU) No 2023/361 in Annex XIII part 2, should be chosen randomly in a number large enough to detect 5% prevalence, it is here interpreted as the sampling of live birds with a number of 60; CL means confidence level.

2.1.4 | Diagnostic methods

Vaccination against HPAI in poultry requires the appropriate tailoring not only of surveillance strategies but also of the matching diagnostic methods to be applied in AI-vaccinated flocks to robustly verify or rule out the presence of AI virus circulation in such establishments. The effect of vaccination on clinical signs, on magnitude and duration of viral shedding following field virus exposure and on the serological profile of AI-vaccinated birds requires careful selection of the most appropriate assays according to the vaccination protocol implemented and the objective of surveillance as described in Table 2. In this Opinion, a description of the following different diagnostic tests that can be used as part of HPAIV surveillance is reported: classical virological methods (virus isolation), molecular virological methods (conventional RT-PCR and real-time RT-PCR (RT-qPCR)), rapid antigen detection systems and serological methods (agar gel immunoprecipitation (AGID), haemagglutination inhibition (HI) and ELISA technology). An updated list of diagnostic protocols for AIV detection and typing is maintained by the EURL and is available at the EURL website (EURL, online).

For each of the described methods, the WG conducted a qualitative assessment, according to the objective of the surveillance, whose results are reported in Section 3.1.1.

Serological data from experimental trials conducted in laboratory on ducks, turkeys and chickens performed by AI-NRLs were obtained from France (ducks), Italy (turkeys), the Netherlands (chicken layers) and Belgium (chicken layers). Those data were analysed with the aim to assess:

- Expected hemagglutination inhibition (HI) titres in vaccinated birds according to the vaccine(s) used for immunisation. Such values can be used to evaluate if immunisation has been obtained in the field.
- HI titres in experimentally vaccinated challenged birds. Irrespective of the severity of the clinical presentation and the magnitude of shedding, vaccinated infected birds, are expected to seroconvert following exposure to field virus. The magnitude of seroconversion is, however, strongly dependent on the effective replication of the virus in the host.
- Performances like sensitivity (Se) and specificity (Sp) of the ELISA assays used for the detection of antibodies against the nucleoprotein (NP) in birds vaccinated with DIVA compatible vaccines following infection.

Also, the monitoring of clinical signs and the possible use of sentinels are included in the assessment in Sections 3.1.1.5 and 3.1.1.6, respectively.

2.1.5 | Methods to assess surveillance strategies

To assess the effectiveness of the surveillance strategies in poultry after vaccination as described in Table 2, different methods were used at the animal and flock level (e.g. within establishment and between establishments). These methods were applied independently or combined, depending on whether the surveillance was conducted during emergency protective or preventive vaccination. Note that surveillance in captive birds was qualitatively assessed in Section 4.2.5.

2.1.5.1 | Methodology to assess surveillance for early detection (E1) and demonstration of freedom (E3, E4) following emergency vaccination

Early detection is here defined as the identification of infection with HPAIV in order to remove an infected poultry establishment before it transmits the infection on average to one or more establishments. Considering that a poultry establishment may consist of one or more flocks, in this Scientific Opinion, an establishment was considered infected if HPAIV was detected in one of its flocks and the flock was the sampling unit. We used the between-flock reproduction number (R_h) as a measure of transmission, and the surveillance was considered to be effective whenever it contributed to keeping R_h below 1 (meaning the epidemic would fade out) as performance indicator. The attributes that define surveillance effectiveness are the sample size at flock level, frequency of sampling, type of sample and diagnostic test sensitivity. For this assessment, we followed a methodology based on the approach previously described by Gonzales et al. (2014).

Transmission dynamics of HPAIV

To identify the critical surveillance parameters such as prevalence of infectious animals, the infectiousness of a flock at the time of detection and the expected efficacy of the surveillance strategy at the establishment level, it is key to understand the infection dynamics in vaccinated flocks where the protection conferred by the vaccine may not be sufficient to stop the disease transmission ($R_h > 1$).

Transmission parameters for unvaccinated and vaccinated chicken layer, duck and turkey flocks are summarised in Table 3. These data were collected as part of the systematic literature review (SLR) performed in Part 1 of the Scientific Opinion on vaccination of poultry against HPAI (EFSA AHAW Panel, 2023). There, it was assumed that in a vaccinated poultry population, 70% of the flocks were fully protected (R < 1) 3 weeks after vaccination, while 30% were partially protected (R > 1). Virus introduction into fully protected farms will result in infection of one or a few birds (minor outbreaks) and the probability of detecting such incursions is extremely low given that the number of infected birds is very small and that they will not show clear signs of infection. The probability of identifying them in a serological or virological survey (random

sample) are also extremely low given the very low number of infected/seroconverted birds relative to the sample size. In addition, such minor outbreaks do not pose a risk as the few infected birds will completely recover (no carriership of any kind) and the probability that they will infect birds on other farms is negligible due to lack of contact. Consequently, surveillance is here focused on detecting major HPAIV outbreaks in the vaccinated flocks given that the 30% of the vaccinated ones will be only partially protected; it is also assumed that the surveillance in unvaccinated flocks will not change and it will be based on detection of signs of infection via passive surveillance.

Parameters for within-flock transmission such as the transmission rate, latent and infectious periods were derived from transmission experiments performed in the Netherlands (Germeraad et al., 2023), France (Grasland et al., 2023) or identified during the previous Opinion (Ssematimba et al., 2019; Tatár-Kis et al., 2019) (Table 3).

Case fatality which is the mortality attributed to HPAI was derived from the results of SLR reported in Table 2 in Opinion Part 1 (EFSA AHAW Panel, 2023). The case fatality, in the reviewed experiments, was calculated as the number of dead birds due to HPAI over the total number of infected birds. Values for daily mortality not attributed to HPAI ('baseline mortality rate') were obtained from Gonzales and Elbers (2018), Elbers and Gonzales (2021) and Gonzales et al. (2023). These authors analysed daily mortality, during the production cycle, in a sample of not-infected flocks. During an outbreak of HPAI, the daily mortality will consist of uninfected and HPAIV-infected dead birds (Table 3). This information was used to assess the efficacy of testing dead birds. Of note, based on the SRL performed for the SO Part 1 (EFSA AHAW Panel, 2023), case fatality lower than 20% was associated with vaccines effective to stop sustained transmission (no outbreaks would take place in these flocks).

Mortality on the day of reporting suspicion parameter (Table 3) was calculated based on the daily mortalities recorded at the time of reporting suspicions of HPAI by farmers in the Netherlands and France (Unpublished data). This parameter was used as a performance indicator of passive surveillance and as reference for comparison of the infectiousness of a flock up to the time of detection when using enhanced passive surveillance or active surveillance.

Mortality thresholds for reporting suspicion (Table 3) were defined based on the studies by Gonzales and Elbers (2018), Elbers and Gonzales (2021) and Gonzales et al. (2023). They established mortality levels that could trigger farmers to report suspicions of HPAI. Whenever the observed mortality exceeds this predefined percentage, farmers are assumed to report their suspicions. Although these thresholds were defined using data from unvaccinated flocks., we assessed the efficacy of using such thresholds for early detection of vaccinated flock.

	Chicken layers		Ducks		Turkeys		
Parameter	Unvaccinated ^a	Vaccinated ^a	Unvaccinated ^{b,c}	Vaccinated ^c	Unvaccinated ^{d,e}	Vaccinated ^f	
Transmission rate (day ⁻¹)	1.13	0.47	4.02 10.8	1	3.2	0.64	
Latent period (days)	1	1	1 1	1	1		
Infectious period _{survive} (days)	3.2	6.8	7 8.1	2.7	-	4	
Infectious period _{die} (days)	3.2	4.5	4.9 -	-	4	4	
Case fatality (range)	0.95–1	0.2–1	0-0.8	0-0.1	0.9–1	0.37-0.62	
Daily mortality not attributed to HPAI (baseline mortality rate)	0.0002		0.0004		0.0007	0.0007	
Mortality at day of reporting suspicion. Mean (range)	1.66 (0.25–5.27)%		1.8 (0.46–15.34)%		1.50%		
Proposing reporting thresholds	0.08 (indoor layers)– (outdoor)%	0.13	0.17%		0.21% ^g	0.21% ⁹	

TABLE 3 Within-flock transmission parameters for unvaccinated and vaccinated partially protected bird flocks (i.e. 30% of the vaccinated flocks where R > 1).

^aGermeraad et al. (2023)

^bTatár-Kis et al. (2019). The value at the left of '|' come from this reference.

^cGrasland et al. (2023). Values for vaccinated are the upper confidence limits of the parameter estimates. Mean estimates resulted in *R* values lower than 1. These values are for mule ducks. These values are those provided at the right of '|'.

^dSsematimba et al. (2019).

^eReference laboratory EURL

^fThese values are assumed since no data on transmission parameters in vaccinated flocks could be found.

⁹No literature was found, hence we assumed a value three times higher than the 'normal' daily mortality as a potential threshold for evaluation.

For illustration purposes, graphical representations showing the expected number of infectious birds over time during an outbreak and the expected daily mortality for unvaccinated and vaccinated flocks are shown for chicken layers, ducks and turkeys in Figures 1–3, respectively. These figures have been built using a deterministic (frequency dependent) compartmental SEIRD model (S=Susceptible, E=exposed, I=infectious, R=recovered, D=dead) (Keeling & Rohani, 2008), which includes normal mortality affecting all compartments and case fatality affecting the I compartment only (Hobbelen et al., 2020). Dynamics were built using the parameters reported in Table 3: the transmission rate, the rate at which birds



FIGURE 1 Infection dynamics (deterministic modelling) of HPAI H5N1 in unvaccinated (left panels) (reproduction number = 3.6) or vaccinated (right panels) (partially protected, vaccine does not fully stop sustained transmission, R = 2.6) **chicken layers**. Dd, daily mortality; *I*, Infectious birds; R = recovered-seroconverting birds. The two lower panels provide a zoomed in view of the upper panel, facilitating the visualisation of the reported suspicion mortality level (dashed line) and how this level (1.66%, Table 3) intersects the infection dynamics curves. Case fatalities assumed were 0.95 (95%) and 0.4 (40%) for unvaccinated and vaccinated flocks, respectively. The two lower panels provide a zoomed-in view of the upper panel, facilitating the visualisation of the reported suspicion mortality level (dashed line) and how this level (1.66%, Table 3) intersects the infection dynamics curves. Case fatalities assumed were 0.95 (95%) and 0.4 (40%) for unvaccinated and vaccinated flocks, respectively. The two lower panels provide a zoomed-in view of the upper panel, facilitating the visualisation of the reported suspicion mortality level (dashed line) and how this level (1.66%, Table 3) intersects the infection dynamics curves. Case fatalities assumed were 0.95 (95%) and 0.4 (40%) for unvaccinated and vaccinated flocks, respectively.



FIGURE 2 Infection dynamics (deterministic modelling) of HPAI H5N1 in unvaccinated (left panels) (reproduction number = 24) or vaccinated (right panels) (partially protected, vaccine does not stop sustained transmission, R = 2.7) **ducks**. Dd, daily mortality; *I*, Infectious birds; *R*, recovered-seroconverting birds. The two lower panels provide a zoomed-in view of the upper panel, facilitating the visualisation of the reported suspicion mortality level (dashed line) and how this level (1.8%, see Table 3) intersects the infection dynamics curves. Case fatalities assumed were 0.4 (40%) and 0.05 (5%) for unvaccinated and vaccinated flocks, respectively.



FIGURE 3 Infection dynamics (deterministic modelling) of HPAI H5N1 in unvaccinated (left panels) (reproduction number = 12.8) or vaccinated (right panels) (partially protected, vaccine does not stop sustained transmission, R = 2.6) **turkeys**. Dd, daily mortality; *I*, Infectious birds; *R*, recovered-seroconverting birds. The two lower panels provide a zoomed-in view of the upper panel, facilitating the visualisation of the reported suspicion mortality level (dashed line) and how this level (1.5%, see Table 3) intersects the infection dynamics curves. Case fatalities assumed were 0.95 (95%) and 0.4 (40%) for unvaccinated and vaccinated flocks, respectively.

Definition of infectiousness of a flock and between-flock transmission

Reporting and control of HPAI is compulsory in the EU, and HPAI outbreaks in unvaccinated flocks are generally detected by passive surveillance before the outbreaks reach their completion (no infectious or susceptible birds left in the flock) (Figures 1–3). We assumed that a flock is infectious (i.e. the establishment is therefore infected) and can infect other establishments from the time the flock becomes infected and the first infected birds become infectious (generally 1 day post infection) until the daily mortality rate within the flock reaches the observed average mortality rates of the reported and confirmed outbreaks. At this stage, it is assumed that the flock would be detected and removed (Table 3). We defined infectiousness by function A(t), and assumed that infectiousness is dependent on the prevalence of infectious (shedding virus) birds in time, so that the higher this prevalence the higher the infectiousness. Therefore, A(t) was estimated as (Gonzales et al., 2014):

$$A(t) = \int_0^t l(\tau) d\tau, \tag{1}$$

where $l(\tau)$ is the prevalence of infectious birds at time t and the expected overall infectiousness was obtained by integrating over t from the time of introduction of infection (t = 0) to the time of detection by passive surveillance (t = ps). This overall infectiousness multiplied by a constant c, which contains information on the between-flock contact rate and the probability of transmission, defines the expected between-flock reproduction number R_{h} :

$$R_h = cA(ps). \tag{2}$$

Because the transmission dynamics in vaccinated flocks are different from the one in unvaccinated flocks and mortality levels are likely to be lower, thereby compromising detection by passive surveillance, we did not estimate infectiousness for vaccinated flocks, and we assumed instead that an infected vaccinated flock would remain undetected by passive surveillance until it reaches a level of infectiousness similar to the one we estimated for an unvaccinated flock. In other words, we used the estimated A(ps) for unvaccinated flocks as a reference to assess the efficacy of implementing active or enhanced passive surveillance (see below).

Active or enhanced passive surveillance and its effect on transmission

Even when surveillance is performed on a regular basis, there remains a probability that an infected flock escapes detection. This can occur, for example, when the prevalence of infected birds at the time of sampling is still lower than the predefined design prevalence. In such cases, there is a potential risk of transmission to flocks in other establishments. The probability of escaping detection (P_{esc}) at time t is defined as

$$P_{\rm esc}(t) = \prod_{\tau=0}^{\tau=t} \exp\left(-\frac{p(\tau)n\mathsf{Se}}{\Delta}\right),\tag{3}$$

where $p(\tau)$ is the prevalence of infected birds, which depending on the sampled population could be dead birds, infectious live birds or recovered-seroconverting live birds. Se is the diagnostic test sensitivity expressed in values between 0 and 1 (e.g. qPCR for swab samples from dead or infectious live chickens or ELISA for serum samples to detect seroconverting live birds) and *n* is the number of birds sampled (sample size). Finally, Δ is the sampling interval (in days) and refers to the repeated sampling performed on a flock at a determined time interval with the probability of escaping detection at time τ with $p(\tau)$ being conditional on the probability of having escaped detection at an earlier sampling time with $p(\tau - 1)$ within the sampling interval. A key parameter determining the probability of escaping detection is the prevalence of infected birds at the time of sampling p(t). However, in reality, it is not known when a flock could become infected. Therefore, this model considers that the introduction of infection takes place at a random moment between two adjacent sampling events. Following this consideration, the expected prevalence of infected birds can be calculated by averaging the prevalence of infected birds at the moment of sampling $p(\tau)$ over all possible introduction moments within Δ (Gonzales et al., 2012; Graat et al., 2001). In this analysis, we assumed perfect specificity, which means that any positive result, particularly if detection is based on a serological test, would need to be followed up to confirm or exclude infection.

Next, we estimated the expected infectiousness of a flock that escapes detection ($A_s(t)$). This infectiousness is a function of the fraction of infectious birds escaping detection at time (t) of sampling as

$$A_{\rm s}(t) = \int_0^t I(\tau) P_{\rm esc}(\tau) d\tau.$$

23 of 70

The expected infectiousness of a flock that escaped detection was then obtained by integrating over $t \in [0, ps]$. If surveillance is effective, we expect $A_s(ps)$ to be lower than A(ps) and this reduction in infectiousness would result in a proportional reduction in R_{hr} hence similar to Equation (2), the expected between-flock reproduction number given surveillance (R_s) (Gonzales et al., 2012) can be calculated as

$$R_{\rm s} = cA_{\rm s}({\rm ps}). \tag{5}$$

From Equation (2), we obtained $c = \frac{R_h}{A(ps)}$ and by substituting c into Equation (5) we got:

$$R_{\rm s} = \frac{A_{\rm s}(\rm ps)}{A(\rm ps)}R_{\rm h}.$$
(6)

When $R_s < 1$ a flock would be detected on average before it transmits infection to another flock in one or more other establishments ('early detection'). Under this requirement, large epidemics would not be expected, due to early detection and removal.

A key parameter to consider in the model is R_h . We have only estimated R_h for unvaccinated poultry populations in some European countries (EFSA AHAW Panel, 2023) and there are no specific estimates of R_h in populations where there is a combination of vaccinated-protected poultry flocks, vaccinated-partially protected flocks and unvaccinated flocks. Because of that, we had to assume a hypothetical value. In poultry-dense populations like those included in the Opinion Part 1, one could expect that R_h exceeds 1, but is lower than the R_h estimated in unvaccinated populations.

Model implementation

Using the parameters reported in Table 3, a stochastic compartmental SEIRD model (Hobbelen et al., 2020) was built using the library SimInf (Widgren et al., 2019) in the statistical software R (R Core Team, 2018). We refer the reader to Widgren et al. (2019) for a detailed overview of stochastic compartmental models within this package. A total of 1000 simulations of HPAI outbreaks (for each poultry species) in vaccinated flocks were performed. Building on these simulations, the abovementioned surveillance model was implemented to quantify A(t) and $A_s(t)$. These values were then related to a hypothetical R_h to calculate R_s as the reference measure to compare different surveillance strategies, encompassing different sample types, sizes and frequencies at the flock level. The methodological steps were as follows:

- We selected outbreak simulations in which a significant number of infected birds (/) occurred, i.e. those with a peak of I exceeding five birds (regardless of the assessed flock size). We considered these outbreaks 'major outbreaks'.
- For each of these major outbreaks, we estimated A(t) and $A_s(t)$. The latter depends on the surveillance strategy assessed.
- Surveillance strategies assessed were a combination of the type of sample, the sample size, sampling frequency and the diagnostic test used.
- Different sampling strategies were considered, involving the random selection of birds and sample sizes. Specifically, all dead birds up to a number of 5, 10 and 15 or more dead birds were randomly sampled and tested. If there were fewer dead birds than the required sample size (e.g. 5) on a given day (or a combination of 2 days), all available dead birds are sampled and tested. Note that in the model one additional day was assumed between this sampling/testing day and the diagnostic results. In addition, scenarios where live birds are randomly selected for sampling were also assessed. We assessed a sample size of 60 birds (as suggested in the EU Regulation), but also tested lower (*n*=30) or higher sample sizes (*n*=90 or 120). Note that sampling of birds refers to sampling birds in a flock, considering that an establishment can have more than one flock, this sample sizes would have to be taken for each flock within the establishment.
- Each considered sample size was assessed in combination with different sampling intervals: every 7, 14 and 30 days as recommended in the EU Regulation. In addition, we assessed longer sampling intervals. A combination of sample size and sampling interval was considered suitable only if the probability to escape detection reached levels lower than 0.01 (Figure 4) for more than 95% of the outbreak simulations. This means that at least 99% of outbreaks are detected in ≥ 95% of the simulations, given that surveillance strategy.
- To assess the effectiveness of surveillance for early detection, we defined the following hypothetical R_h values (see model description) in unvaccinated flocks ranging from 1.4 to 1.8, which are the mean R_h estimates obtained, from the transmission kernel analyses done for the Opinion Part 1 using data from epidemics in France, Italy and the Netherlands (EFSA AHAW Panel, 2023).
- We then derived the following performance indicators:
 - Reduction of infectiousness $A_s(t)$ and $R_{s'}$ given detection.
 - Median time from introduction to detection of an outbreak, which was derived as the time when the probability of escape at the farm level was 0.5 (Figure 4).
 - In addition, the prevalence of infectious and recovered (seroconverting) birds at the time of detection was also reported.



FIGURE 4 Probability of an infected vaccinated flock (one outbreak simulation is shown) escaping detection given three surveillance strategies based on sampling five dead birds at an interval of 7 days (black line), 14 days (red line) and 30 days (blue line). In this flock the probability of escaping detection when sampling at intervals of 14 and 30 days does not decrease below the required level of detection 0.01 (\geq 99% of probability of detection). This requirement could be reached by increasing the number of birds sampled. The x-axis is the time in days from the start of outbreak in the flock.

2.1.5.2 | Methodology to assess surveillance for early detection (P1) and demonstration of freedom (P3, P4) following preventive vaccination

Scenario tree models were used to estimate the sensitivity (Se) of the surveillance systems to demonstrate freedom from disease and to provide inputs for assessing early detection of HPAI, the latest only in a preventive vaccination scenario. A scenario tree model is a structured approach to describe the process of disease detection of a surveillance system or a component of that. A surveillance system component is a surveillance activity that, in itself, can contribute evidence to disease freedom; it has the capacity to detect disease if/when it occurs. This includes both general and targeted surveillance activities; both active and passive. A scenario tree graphically traces all pathways by which a single unit can be detected as infected by the system; it defines the conditional probabilities associated with each step in the pathway and it is useful for calculating surveillance components' sensitivity and the sensitivity of the surveillance system as a whole (Martin, Cameron, Barfod, et al., 2007; Martin, Cameron, & Greiner, 2007). This analysis provides an estimation of the probability that the surveillance would be able to detect an infection or a disease in the unit of interest (an animal, a sample, a pooled sample, a group of animals, a region), given that the condition is present in the population at a level specified by the design prevalence. Surveillance component sensitivity (CSe, see Glossary) is the average probability that a surveillance system component will return a positive surveillance outcome, given that the disease is present in the population at a level equal to or greater than the specified design prevalence.

The logical and practical structure of the surveillance strategy applied can be outlined by a scenario tree, in which the process of disease detection is quantitatively described by including all factors affecting the probability that a surveillance unit will be infected and all those affecting the probability that will be detected. In each branch of the tree, which usually identifies a surveillance component, proportions and probabilities are assigned to a node, one for each stage of the diagnostic process. We can identify different types of nodes: category nodes, which define groups into which units or group of units fall, risk nodes (the population into subsets with different risks of being infected at the design prevalence), infection nodes (infection status of a unit), detection nodes (any event, action or test which contributes to an infected unit's detection in the surveillance process). An example of the scenario tree model used for this assessment is provided in Figure 5. This tree illustrates the category of the population which is vaccinated and represents the target of surveillance (which is the focus of the assessment), commercial chicken layers in this example. The tree includes multiple risk nodes. The first risk node is associated with the relative risk (RR) of virus incursion in the zone, which describes the probability of virus incursion in a given geographical area relative to the probability of the same event in the other geographical areas (RRLow, RRMod, RRHigh). The proportion of poultry present in this incursion zone is also attributed to this risk node (Prpoultry). The second risk node refers to the RR of virus incursion in poultry, which describes the probability of virus incursion in a given poultry type, relative to the probability of the same event in the other poultry types (RRLayers&breeders, RRBroilers, RRTurkeys, RRDucks). The proportion of poultry type being present in this incursion zone is attributed to this risk node (Prlayers, Prbroilers, Prturkeys, Prducks). The tree also includes probability nodes, that are the probability that a vaccinated establishment is protected (Pvp) and the probability that sustained transmission is established following incursion (Pinf). Pinf was calculated based on a design prevalence of 0.01 multiplied by the probability of having major outbreaks (for simplicity, in

unvaccinated flocks, this probability is assumed to be 1 and for vaccinated flocks this probability is $1 - 1/R_{v'}$ where R_v is the within-flock transmission parameter).

As ultimate outcome of the tree there is the probability that an outbreak is detected (P_{pos}) given the Se of testing at flock level (F_{se}). Note that one or multiple flocks could be present within an establishment, and that the detection of one infected flock is enough to consider the whole establishment infected (see Table 4 for an explanation of these elements).



FIGURE 5 Illustrative diagram of a scenario tree model describing the surveillance process applied to the poultry population vaccinated. For this illustration, chicken layers represent the target for preventive vaccination in a given country and the unit where surveillance is applied is the establishment. A similar scenario can be derived for any other poultry sector/species targeted for vaccination (e.g. ducks in France or turkeys in Italy) and at other unit level (e.g. flock, zone). $F_{se'}$ sensitivity of testing at establishment level; Lay, layers; $P_{inf'}$ probability of infection; Pr, proportion; $P_{VP'}$ probability that a vaccinated establishment is protected; RR, relative risk.

The risk nodes contain information on the proportion of the population within each node and the RR attributed to that. The RR describes the probability of virus incursion in one group relative to the probability of the same event in the other group. These two are combined into adjusted RR (AR for the zone: ARLow, ARMod, ARHigh; AR for poultry: ARLayers&breeders, ARBroilers, ARTurkeys, ARDucks), which ensures that the average design prevalence is constant across the population (for more details, see Martin, Cameron, & Greiner, 2007).

The probability of detection of a truly infected flock/establishment (outbreak) was given by:

$$p_{\text{pos}} = \text{AR}_{\text{zone}} \times \text{AR}_{\text{pt}} \times (1 - P_{\text{vp}}) \times P_{\text{inf}} \times F_{\text{Se}}.$$
(7)

For unvaccinated flocks, $P_{vp} = 0$. The testing strategy at the establishment (flock) level and corresponding F_{Se} is derived from the early detection model described in the previous section. Then, the sensitivity of a surveillance component (CSe), which represents the probability that at least one infected flock (establishment) is detected and thus refers to a specific part of the overall surveillance system, was given by:

$$CSe_i = 1 - (1 - P_{pos})^n$$
. (8)

With *i* identifying each specific component of the system and *n* being the number of establishments sampled within the component. If all farms are sampled, then the exact method to calculate CSe was used as described by Martin, Cameron, Barfod, et al. (2007) and Martin, Cameron, and Greiner (2007). Note that, if an establishment has more than one flock, each flock needs to be sampled following the defined surveillance strategy. One infected flock is sufficient to consider the whole establishment infected. The WOAH standards state that all vaccinated establishments need to be sampled. Hence, *n* in our reference surveillance scenario includes the total number of vaccinated establishments within the target surveillance area/ zone. Additionally, we assessed scenarios based on random sampling of a representative number of establishments.

TABLE 4 Parameter values included in the model and the source from which these values were obtained.

Parameter	Description	Values	Reference
Relative Risk of virus incursion in the geographical zone	The relative risk of virus incursion in the zone describes the probability of virus incursion in one given geographical area, relative to the probability of the same event in other geographical areas (RRLow, RRMod, RRHigh). These relative risks were estimated by analysing data of primary introductions of HPAI in poultry in the Netherlands and were used for modelling surveillance in France and Italy as well	Mean (95% CI) ^a RRLow=1 (reference) RRModerate=2.98 (1.80–4.78) RRHigh=6.62 (4.40–9.90)	This Opinion. Estimates were derived from data in Gonzales et al. (2023)
Relative risk of virus incursion in poultry species	The relative risk of virus incursion in poultry describes the probability of virus incursion in one given poultry species, relative to the probability of the same event in the other poultry species (RRLayers&Breeders, RRbroilers, Rrturkeys, Rrducks). The relative risks estimated for the Netherlands were used for modelling surveillance in France and Italy as well	Mean (95% CI) ^a RRLayers&breeders = 1 (reference) RRBroilers = 0.3 (0.1–0.7) RRTurkeys = 2.8 (1.1–6.5) RRDucks = 3.7 (1.8–7.6)	This Opinion. Estimates were derived from data in Gonzales et al. (2022)
Proportion of poultry population (Pr)	The proportion of specific poultry type establishments (e.g. turkeys) within the area/ zone targeted for surveillance	Country specific	Opinion Part 1 (EFSA AHAW Panel, 2023)
Probability protected (P_v)	The proportion of establishments (flocks within the establishments) effectively protected following vaccination	0.7 (0.5–0.8) ^a	Opinion Part 1 (EFSA AHAW Panel, 2023)
Probability of infection (P _{inf})	This is the probability of having a major outbreak following virus incursion adjusted by the hypothetical design prevalence. $P_{inf} = DP \times (1 - 1/R_v)$, where R_v is the within-flock transmission parameter. For unvaccinated flocks, the probability of a major outbreaks is assumed to be 1	<i>R</i> _v values ^a for Chickens: 2.1 (1.03–4.0) Ducks: 2.7 (1.3–4.0) Turkeys: 2.6 (1.3–4.0) Design Prevalence (DP) = 0.01	Germeraad et al. (2023) Grasland et al. (2023). Table 3
Se sampling at flock level (F _{Se})	Combination of sample size and sampling interval derived using the methodology for early detection described in Section 2.1.5.1	Se values ^a [Median (lower – higher)] ^b when sampling 15 birds every 30 days were: Chickens: 97 (92–99) Ducks: 96 (89–98) Turkeys 96 (87–97)	This Opinion

^aFor all these parameters, a pert distribution was used in the stochastic simulations.

^bThese values are the percentage of outbreak simulation detected (Sensitivity) for probabilities of escaping early detection of < 0.05 (median). Lower and higher values used for the stochastic simulations around this parameter (using a pert distribution) were derived by using the escaping probabilities < 0.01 (lower value) and < 0.1 (upper value).

Then, to assess the early detection surveillance sensitivity of each SSC (EDSe_{*i*}), that is the probability that the surveillance activity is able to correctly detect, among the poultry establishments, the epidemiological unit(s) affected by a new incursion within a specific time frame, we followed the method proposed by Cameron et al. (2020):

$$EDSe_i = Cp_i \times Ct_i \times CSe_i$$

where Cp_i corresponds to the proportion of the population, in component *i*, targeted for surveillance, with Ct_i being the temporal coverage, indicating the conditional probability that an establishment will be tested within a predefined time frame, e.g. the time corresponding to median time from introduction to detection of an outbreak by passive surveillance divided by the given sampling interval. Time frame values for unvaccinated species and vaccinated species were extracted from results provided in Tables 15A–17B. CSe_i corresponds to the sensitivity of the surveillance component *i* (Equation. 8).

After the EDSe for each surveillance component had been calculated, the total surveillance system sensitivity (TotalSe, see Glossary) could be summarised as:

$$TotalSe = 1 - \prod_{i} (1 - EDSe_{i}).$$
(10)

With *i* being the number of component's EDSe.

(9)

Finally, the probability that the population is free from HPAI (Pfree), given that surveillance did not detect any infected establishment and assuming perfect specificity, as stated in the previous section, was based on Martin, Cameron, Barfod, et al. (2007) and Martin, Cameron, and Greiner (2007) and given by:

$$\mathsf{Pfree} = \frac{1 - \mathsf{priorInf}}{1 - \mathsf{priorInf} \times \mathsf{EDSe}'}$$
(11)

where priorInf is the prior probability that the poultry population at the establishment level would be infected (hypothetical prior prevalence in the population, see Glossary). To determine this prior probability, we used the hypothetical design prevalence of 1% times the risk (probability) for infection during the autumn–winter months relative to that in the spring–summer months. This relative risk was considered to be 5 (Gonzales et al., unpublished data). Consequently, priorInf was assumed to be 5%.

In Equation (11), we used either EDSe or TotalSe depending on the vaccination status of poultry establishments for which the probability of freedom was estimated. We used EDSe if we focused on the vaccinated establishment and TotalSe if we focused on all the establishments (vaccinated and unvaccinated) subjected to surveillance in all five components of the scenario tree.

2.1.5.3 | Methodology to assess vaccination effectiveness and level of immune response induced by vaccination (surveillance strategies E2, P2)

The surveillance strategies to assess vaccination effectiveness and the level of immunity response, E2 and P2, respectively, were addressed descriptively and the assessment performed is reported in Sections 3.1.2.4 and 3.1.2.5.

Vaccine effectiveness (VE) (E2) was considered as the ability of a vaccine to protect animals from a disease or an infection under field conditions. It differs from **vaccine efficacy**, which is measured under controlled conditions in laboratory settings typically through challenge experiments. The effectiveness of vaccination in real scenarios is likely to vary across countries and vaccination campaigns and can be assessed by observing the occurrence of HPAI outbreaks in a country where vaccination is implemented.

VE can be estimated as 1 minus a measure of RR in the vaccinated population compared with the unvaccinated population. These RR measures depending on the type of data collected and method of analysis could be for example: the RR, based on incidence or rates of infection, prevalence ratio (attack rates, cumulative incidence), the odds ratio (OR) or the hazard ratio (HR).

Level of immune response induced by vaccination (P2). Given the lack of knowledge on identifying a correlate of protection, e.g. a given HI titre, which can be used to monitor the level of immunity reached within a flock (EFSA AHAW Panel, 2023), possible pragmatic approaches to estimate the level of immune response induced by vaccination were given and general considerations, e.g. on when to apply the sampling, were discussed.

2.2 | ToR 4 – Risk mitigation measures

In ToR 4, the assessment focused on risk mitigation measures to be implemented in vaccinated poultry aiming at preventing the spread of HPAIV by moving poultry and their products following emergency and preventive vaccination. Firstly, the measures laid down in the Delegated Regulation (EU) No 2023/361 and in the general and specific conditions of Delegated Regulation (EU) No 2020/687⁸ for derogation were assessed (ToR 4.2). To enhance clarity, in Figures 6 and 7, these risk mitigation measures provided in the current legislation are described as two distinct groups: testing measures and other measures. It is important to highlight that these two categories of measures are not alternatives, but they complement each other. Secondly, alternative risk mitigation measures were defined by the Working Group, for both emergency and preventive vaccination scenarios, as requested in ToR 4.2.

2.2.1 | Risk mitigation measures as per delegated regulation (EU) No 2023/361 after vaccination

Delegated Regulation (EU) No 2023/361 envisage different risk mitigation measures to be implemented following vaccination in poultry, captive birds and wild birds.

According to Part 3 in Delegated Regulation (EU) No 2023/361, when implementing **emergency protective vaccination**, vaccinated animals and their products will be subject to the prohibition of movements within and outside the vaccination zone. The competent authority may allow the movement of animals if they are subject to compulsory killing or if their movement has been authorised. Movements of vaccinated poultry/captive birds/their product can be authorised if all the conditions for moving them from protection and surveillance zones provided in Delegated Regulation (EU) No 2020/687⁹ (Article 28, 29, 30, 31(1), 33, 34 and 37) are fulfilled (Figure 6). At the end of the recovery period (i.e. 28 after completion of emergency protective vaccination or when lifting the restricted zones), all the measures for granting a

⁸As per Articles 28, 29, 30, 31 (1), 33, 34 and 37.

⁹As per Articles 28, 29, 30, 31 (1), 33, 34 and 37.

derogation for moving that are provided for preventive vaccination applies (Figure 7), including the reinforced surveillance to be implemented for preventive vaccination (see Table 1).

According to Points 3 to 4 of Part 5 in Delegated Regulation (EU) No 2023/361, when implementing **preventive vaccination**, the movement of animals and their products from the establishment of vaccination is prohibited. The competent authority may allow the movement of animals and their products if they are not included in the list of animals and products subject to the prohibition of movements or if they comply with relevant conditions and their movement has been authorised. The conditions for granting a derogation for movement are described in Figure 7.

202

1/2024

RISK MITIGATION IN EMERGENCY VACCINATION

Vaccinated poultry and their products are subject to prohibition of movement within and outside of vaccination zone (Art.13 ${\rm DR}\,2023/361)$

DEROGATION OF PROHIBITION OF MOVEMENT (Art.28 DR 2020/687)



FIGURE 6 Provisions as in Delegated Regulation (EU) No 2020/687 to grant authorisation for movement of poultry/captive birds/their product within and from emergency vaccinated zones.

20

on [15/11/2024]

RISK MITIGATION IN PREVENTIVE VACCINATION

Vaccinated poultry and their products are subject to prohibition of movement outside of the establishment where they were vaccinated (Art.14 DR 2023/361)

DEROGATION OF PROHIBITION OF MOVEMENT (Annex XIII 2023/361)

SPECIFIC CONDITIONS



FIGURE 7 Provisions as in Delegated Regulation (EU) No 2023/361 to grant authorisation for movement of poultry/captive birds/their product within and from preventive vaccinated zones.

2.2.2 | Definition of risk mitigation measures scenarios

In addition to the mitigation measures described in Section 2.2.1, the Working Group identified and assessed the provision of demonstrating freedom from HPAIV as an alternative approach to enable safety movement of poultry and their products within and outside vaccinated zones and from a vaccinated establishment. The methodology used to demonstrate freedom from HPAIV was the scenario tree model as it is described in Section 2.1.5.2.

2.3 | Uncertainty analysis

The uncertainty analysis was performed following the procedure detailed in the EFSA guidance on uncertainty analysis in scientific assessments (EFSA Scientific Committee, 2018). The sources of uncertainty associated with the data available and methodology used are listed and discussed in Section 3.3, but their impact on the assessment was not quantified. For the assessment of the effectiveness of the surveillance strategies at the establishment level (except 'passive reporting' and 'mortality threshold'), the percentage of simulated outbreaks in which the desired detection probability threshold (>99%) was achieved was used as a proxy for the certainty regarding the effectiveness of a particular surveillance strategy, so that only strategies in which >95% certainty was achieved were considered effective. In the case of surveillance strategies based on 'passive reporting' and 'mortality threshold', the certainty on their effectiveness was quantified using the 97.5% confidence intervals for the between-flock transmission (R_s) obtained using the transmission models, so that these strategies were considered effective for HPAIV early detection if the 95% upper confidence limit was below 1 (i.e. their use stopped sustained between-flock transmission).

3 | ASSESSMENT

3.1 | ToR 3: Surveillance

3.1.1 | Diagnostic methods

Ideally, efficient vaccination results in sterile immunity, i.e. the vaccinated animals can no longer become infected upon virus exposure and therefore can neither fall ill nor replicate and excrete the virus. In reality, this state is usually not achieved on a population base. Thus, individual infections in a vaccinated flock that also lead to virus excretion by vaccinated animals cannot be fully excluded. However, the threshold to become infected should be higher and subsequent virus excretion should be significantly lower in amplitude and shorter in time than that of the unvaccinated animals. As a result, smaller amounts of virus at a lower prevalence compared with unvaccinated exposed flocks are available for detection. In order to detect these, the methods with the highest sensitivity should be used in the active surveillance of vaccinated flocks. When assessing various diagnostic methods in this respect, their analytical and diagnostic sensitivity and specificity must be considered along with other factors which can modify their performance characteristics: sample matrices (increased risk of the presence of PCR inhibitors in certain matrices e.g. faeces), enrichment effects (e.g. through skilful selection of target subjects – dead/diseased animals have a higher probability of being virus-positive) or dilution effects (pooling of samples). Vaccinated animals falling sick or dying due to HPAIV infection in vaccinated flocks, in contrast, in most cases are expected to produce amplitudes of virus similar to unvaccinated birds.

3.1.1.1 | Virological methods

Surveillance activities can rely on virological examination of samples collected in vaccinated flocks to obtain timely information about the active circulation of field viruses inside the flock. Vaccination can narrow the window of viral shedding of birds and lower viral replication titres; thus, the ability of virological tests to detect active viral circulation in flocks may diminish. Before molecular methods became widely available in veterinary laboratories, detection of AIV from clinical samples was performed by inoculation of embryonated specific pathogen free (SPF) chicken eggs (ECE). However, nowadays, inoculation of ECE is generally used only on PCR-positive samples as a confirmatory assay or to obtain viral isolates for phenotypic characterisation, since several factors can negatively affect viral isolation success and therefore the sensitivity of the method.

Virus isolation

The preferred method of growing influenza A viruses is by the inoculation of SPF ECE, or specific antibody-negative (SAN) eggs. The use of ECE in some EU Member States may be considered an animal experiment that is subject to authorisation by an ethics committee. Clarified supernatant preparations of clinical samples are inoculated in the chorioallantoic sac of three to five embryonated SPF or SAN ECE of 9–11 days of incubation. The eggs are then incubated at 37°C for 2–5 days and, if deemed necessary, a second passage can be performed. The chances of virus replication are determined by the infectivity retained in the sample. This, in turn, is influenced by characteristics of the sample and the way in which it is transported

and stored to preserve the infectivity of the virus. Virus isolation is not recommended for screening during vaccinationassisted surveillance due to the long duration of the test (embryos are monitored up to 5 days following inoculation by candling and a second passage of again up to 5 days may be necessary to prove the absence of replicating virus). In addition, the high number of samples to test generates high costs for the SPF eggs and binds staff for a considerable time. In the surveillance activities to be implemented in vaccination zones, considering these limitations, the use of virus isolation can only be considered as an alternative to molecular methods in exceptional circumstances limited in time (e.g. lack of reagents for molecular biology methods). However, following outbreak confirmation in vaccinated populations, it would be important to attempt viral isolation to allow further phenotypic characterisation that requires the availability of the live HPAI virus.

3.1.1.2 | Molecular methods

Conventional RT-PCR and Real-time RT-PCR (RT-qPCR)

Molecular techniques are now used preferentially for the diagnosis of HPAIV infection directly from clinical specimens of birds. The current definition of HPAIV according to the Manual of Diagnostic Tests and Vaccines for Terrestrial Animals Chapter 3.3.4 – Avian influenza (including infection with high pathogenicity avian influenza viruses) (WOAH, online-b), includes the molecular identification of virulence factors in the endoproteolytic cleavage site of the haemagglutinin (HA) protein of AIV of subtypes H5 and H7 and therefore enables the use of molecular techniques in the diagnosis of HPAI. Although conventional RT-PCR techniques could result in faster detection and subtype identification (at least of H5 and H7) at lower costs compared with virus isolation, and also provide a PCR amplicon product ready for nucleotide sequencing to define the pathotype, real-time amplification techniques have largely replaced conventional RT-PCR techniques. Real-time quantitative RT-PCR (RT-qPCR) offers increased turn-around time, added cross contamination prevention and is generally applicable to high-throughput analyses at all steps of the diagnostic algorithm, i.e. generic IAV detection, subtype specification and pathotype characterisation. Thus, these types of tests are now routinely used to process samples collected during surveillance activities in the EU. Clinical samples positive in tests specific for H5 or H7 must be further investigated for the identification of molecular signatures typical of HPAI or LPAI viruses. Due to its superior analytical sensitivity, pooling clinical samples is possible with RT-qPCR which further reduces costs. Pool size represents an important measure to reduce costs and allows the processing of larger numbers of samples, if carefully applied. Yet, pooling intrinsically decreases the analytical sensitivity of detection as samples are diluted. When pooling 5–10 samples this effect is usually negligible (Fereidouni et al., 2012; Ladman et al., 2012) but, nevertheless, arithmetically may lead to false-negative results for single samples with viral loads close to the limit of detection. Guidelines based on laboratory evidence on pooling recommendations in different species are available on the EURL website (EURL, online). Laboratories processing samples in pools should validate their internal procedures aiming to maximise efficiency without significantly affecting diagnostic sensitivity of the methods.

For screening purposes, generic M gene targeting RT-gPCR methods (e.g. SOP VIR018 available on the EURL website (EURL, online)) are preferred due to higher diagnostic sensitivity (99%–100%) and specificity (99%–100%) compared with other targets. Such methods can be used on clinical samples collected from live or dead birds for early detection or monitoring freedom from infection in vaccinated or unvaccinated poultry flocks. If vaccination reduces virus shedding effectively, the probability of any PCR to detect a positive sample will decrease. Also, the interpretation of subtype-specific RT-qPCRs in flocks vaccinated with modified live/vectored H5 or H7 vaccines could be affected: Organs/injection sites where active replication of the vaccine is taking place (e.g. spleen) would yield positive results to such assays even in the absence of field virus. Further investigations on recommended diagnostic methods to be applied in animals vaccinated with live vectored vaccines are currently ongoing at EURL. Specific RT-qPCRs tailored to the vector vaccine used may be required to distinguish residual vector vaccine inserts from HPAI field virus. The residual genetic material in birds vaccinated with inactivated vaccines is not to be expected unless injection sites are sampled and examined. Data available from vaccination trials performed in the EU against recent strains belonging to 2.3.4.4b clade indicate that samples collected from the respiratory tract, such as oropharyngeal/tracheal swabs, should be favoured in surveillance activities over cloacal swabs. The use of only oral swabs would be justified by higher and longer shedding compared with the ones observed in cloacal swabs obtained from chickens and turkeys. This would help save resources, labour and time for sampling in active surveillance programmes according to Delegated Regulation (EU) No 2023/361. Any positive result in clinical samples should be interpreted as potential indication of an AI infection and prompt immediate identification of the respective AIV sub- and pathotype by appropriate methods and time for sampling in active surveillance programmes according to Delegated Regulation (EU) No 2023/361.

The same assays used for surveillance in animals could also be applied to analyse sample matrices different from clinical specimens such as environmental samples collected for surveillance purposes or for verification of cleansing efficacy (e.g. swabs from surfaces, drinking water, air samples, etc.). Delegated Regulation (EU) No 2023/361 imposes the condition to use exclusively clinical sample material in vaccination surveillance programmes. However, the extent of sampling and investigating is considerable and might itself form an obstacle to the implementation of vaccination. Thus, the use of further informative matrices of the environment in poultry holdings could be a measure to mitigate those obstacles. Appropriate systematic validation studies of these matrixes and appropriate exogenous inhibition controls for environmental samples

are still missing. Also, based on farm infrastructure and poultry species, the usefulness of different environmental samples can vary considerably (Grasland et al., 2023).

Molecular methods can also be used for monitoring vaccination coverage in birds vaccinated with live vectored vaccines. For example, vector replication and therefore expression of the immunising antigen can be monitored by qPCR assays specifically designed to detect vaccine-specific genomic sections. For HVT, vector replication, and therefore vaccine uptake, can also be assessed in non-invasive samples such as feather pulp. However, the intermittent presence of vectorassociated genomic material in these types of samples is a major limitation for the accurate estimation of vaccination coverage in vaccinated poultry flocks (Cortes et al., 2011).

Next-generation sequencing (NGS) technologies

Novel DNA sequencing techniques, commonly referred to as 'next-generation' sequencing (NGS), are characterised by high speed and throughput allowing the production of an enormous volume of data with many applications in research and diagnostic settings. Several commercially available high-throughput sequencing platforms exist that vary by way of their sequencing principle, sequencing speed, expense and read length. The amount of total nucleic acids belonging to viral particles in clinical samples is often too low for sequencing and may require amplifying of viral nucleic acid using targeted or non-targeted approaches. The aim of these sequencing approaches is to (i) sequence all eight IAV gene segments, (ii) verify the IAV H gene subtype as reported by existing RT-PCR assays, (iii) determine the IAV N subtype and (iv) generate sequence data for analysis of the IAV hemagglutinin cleavage site, for pathotype assessment. In optimal scenarios and using real-time devices, this information can be obtained in less than 24 h, which is in many cases a longer turn-around time than RT-qPCRbased approaches. NGS technologies do not see a wide application for surveillance purposes mainly because of costs and time for analysis, while they are of high value in the further characterisation of detected AI viruses. Analysis of sequencing data can provide valuable information to monitor viral evolution and possible sources of introduction. Moreover, several studies have identified genetic markers associated with specific phenotypic behaviours that can be identified and used to infer the properties of the virus. In the context of HPAI vaccination, monitoring of viral genetic characteristics under the selective pressure exerted by immunity towards HPAI in vaccinated flocks allows the detection of mutant viruses and informs on the need for vaccine updates and the risks to public health.

Other amplification techniques

Several alternative amplification strategies have been developed and published including loop-mediated isothermal amplification (LAMP), strand displacement amplification (SDA) or Crispr amplification techniques. Some applications have been linked to point-of-care testing also. However, validation data for use in vaccinated poultry populations are scarce, and commercial products are not readily available at this moment. The potential of these methods has not been yet fully explored.

3.1.1.3 | Rapid antigen detection systems

Commercially available generic antigen detection kits are mostly based on enzymatic immunoassays or on immunochromatography (lateral flow devices) and generally use a monoclonal antibody against the nucleoprotein or other highly conserved regions of the structural proteins of AIVs. Some of these tests have been designed for the detection of influenza A viruses in poultry and should be able to detect any influenza A virus independent of its subtype. The main advantage of these tests is the rapid confirmation of the presence of an influenza A virus in a few (< 20) minutes. They do not require special laboratory equipment and therefore can be used directly on poultry premises without the need of highly trained personnel. However, they suffer from a substantial lack of sensitivity compared with RT-qPCR, which can produce falsenegative results. Given the intended and expected reduction of viral replication in exposed vaccinated birds and the reported lack of sensitivity, it is highly unlikely that such tests would be useful for active surveillance purposes in vaccinated flocks. However, birds that succumb to HPAIV infection in infected vaccinated flocks might still produce sufficient virus loads to be detected by rapid antigen assays (i.e. bucket sampling). Nevertheless, at the current stage, there is not enough information on the diagnostic sensitivity of these tests in field conditions to recommend their use in surveillance activities aimed at the identification of outbreaks or for pre-movement virological testing of vaccinated flocks. For similar reasons, any positive result obtained with such assays must be further investigated by validated methods characterised by higher specificity to confirm the results.

3.1.1.4 | Serological methods

Upon infection with HPAIV in birds, antibodies are mounted by the animals as part of the immunological response. The antibodies produced are generally directed towards all the viral proteins expressed although variabilities exist in terms of the relative quantities of antibodies produced. Neutralising antibodies, particularly those directed towards HA, are able to abrogate the infectivity of the virus. Different vaccine technologies vary in their ability to stimulate preferentially a humoral or cell-mediated immune response. For example, adjuvanted whole virus inactivated vaccines generally induce high levels of antibodies but with a poor stimulation of the cell-mediated immune response arm. On the opposite side,

replication-competent vectored vaccines are generally capable of more robust stimulation of cell-mediated immunity but might be slow to induce higher levels of circulating antibodies. The type of vaccine applied in a population can therefore affect the character and magnitude of the specific immune response, which in turn may affect its detectability by serological methods. Most of the available tests used for routine diagnostics aim at the detection of antibodies, while, at the time of writing, there are no scalable methods to assess cell-mediated immunity.

Assessment of immunity might be useful for different purposes:

- 1. Screening for vaccine coverage is done to assess the percentage of poultry that has mounted a measurable immune response upon vaccination. Depending on the diagnostic antigen used, the testing could be further stratified to extend the questioning to the protective efficacy of the mounted immune response against specific viruses that are in current circulation. However, defining thresholds of protection on basis of antibody titres has proven difficult (see Opinion Part 1).
- 2. Retrospective indirect screening for virus incursions into vaccinated flocks. Here, serological assays are part of the active surveillance in flocks that have received DIVA vaccination. The general drawback compared with the virological method-ology is the delay in time until antibodies are mounted (usually 7–14 days post infection).

Agar gel immunoprecipitation (AGID)

All influenza A viruses share antigenically similar nucleoproteins and matrix antigens. For this reason, AGID tests are able to detect the presence or absence of antibodies against any influenza A virus in a subtype-independent manner. Although AGID is relatively cheap and suitable for settings with limited resources, it is increasingly being replaced by species-independent ELISA, which can provide greater sensitivity and shorter turn-around times. Limitations of this method include the restriction of its use to species which develop precipitating antibodies following infection (i.e. not suitable for use in Anseriformes). Moreover, the result of AGID is qualitative only, and targets virus proteins not involved in mounting protective immunity. For all the above reasons, AGID would be of limited value for surveillance purposes in vaccinated flocks.

Haemagglutination inhibition (HI)

The HI test is used to identify AIV subtype-specific antibodies such as H5 and H7 or other H subtypes. HI antibodies are a surrogate of neutralising antibodies and therefore often are highly strain-specific. Variability within the same subtype can be extensive, so there is a considerable risk of obtaining false-negative results in case a mismatched antigen is used. Therefore, to ensure high diagnostic sensitivity, it is always necessary to use the most appropriate antigen for each purpose. For example, to evaluate vaccination coverage, it should be fine to use the vaccine antigen or antigens antigenically very similar to the vaccine strain. However, in order to assess protection efficacy against the circulating field virus, an HI test must be carried out with this antigen or with a closely related one. HI titres may be regarded as being positive if there is inhibition at a serum dilution of 1:16 or more against four haemagglutinating units (HAU) of antigen, according to international standards.

Comparing different vaccine challenge studies (Mo et al., 2023), the target values for antibody titres to the challenge virus were identified: a value of \geq 1:16 should be evaluated as a minimum titre that may be sufficient for survival and reduction of virus excretion of exposed poultry. In a field setting with physiological stress or immunosuppressive factors, survival and virus excretion may be accelerated and therefore the minimum positive titre in relation to protective efficacy should not be overinterpreted; it does not necessarily imply, e.g. that immunised birds with that titre will be protected against challenge or that birds with lower titres will be susceptible to challenge.

When compared with inactivated vaccines, HVT-vectored vaccines may be protective even if these threshold titres are not reached. This might be due to cellular immunity preferentially induced by vectored vaccines. Similarly, RNA vaccines are also known to elicit protective immunity through dendritic cell activation followed by stimulation of primary and memory T-cell responses and may induce protection even if HI titres remain below 1:16. Since a full evaluation of both arms of immunity, i.e. B- and T-cell-derived, is not possible at present, it is in any case recommended to use an antigen most closely related to the circulating virus in order to depict meaningful titres in vaccinated groups for an approximate evaluation of the supposed protective immunity in a vaccinated flock.

Onset of detectable immunity, that can be used to assess vaccination coverage in vaccinated poultry populations, is expected between 2 and 4 weeks following vaccination, in most cases. In birds (i.e. turkeys and chickens) vaccinated with HVT-vectored vaccines only, at 1 day of age, the development of a sufficient amount of antibodies to be detected by currently available serological tests could be delayed (i.e. 6–8 weeks following vaccination).

When longitudinal sero-surveillance is carried out using HI techniques, a 2–3 log₂ increase or more in HI titres, in the absence of a new vaccine booster, could be considered a sign of introduction of a field AIV with the same HA subtype as the vaccine in a vaccinated holding after reaching the plateau phase, and should prompt additional virological investigations aimed at the detection and typing of the AIV responsible of the incursion. However, under field conditions, the use of such thresholds require stable and homogenous antibody levels at flock size for a sufficient period of time and sampling intervals need to be short enough to avoid that natural waning of immunity would mask a HPAIV incursion. HI tests therefore should be mainly used to assess vaccine coverage and not for routine surveillance activities in vaccinated flocks. However,

in monitoring flocks to assess freedom from HPAI (E4 and P4 in Table 2), HI tests using a test antigen derived from circulating HPAIV is preferred and yields a more accurate interpretation of sera that tested positive in NP-ELISA method.

ELISA technology

Commercial ELISA kits have replaced AGID assays as a more sensitive and versatile alternative for the detection of influenza A group reactive antibodies in sera. Kits with an indirect and competitive/blocking format have been developed and validated and are routinely used in poultry surveillance activities. Indirect ELISA kits, where the primary antigen-specific antibody is recognised by a secondary conjugated antibody, requires validation in multiple species as the nature of the secondary antibody limits their use to the determined species. Competitive formats instead, allow their use in a species-independent manner. Similar to AGID, most of assays use the nucleoprotein as a target antigen. These assays are qualified for use in flocks vaccinated with DIVA vaccines with NP as a negative marker.

Alternatively, subtype-specific H5 or H7 kits are available, but these kits must be carefully validated to ensure that they identify with sufficiently high sensitivity and specificity the specific seroconversion caused by the circulating strain.

Specific kits for H5 or H7, as well as the generic anti-NP kits if whole inactivated vaccines are used, can also be used for the evaluation of vaccine coverage. In general, the assays are used to produce a qualitative result. Although equations are offered with some assays to define a titre-like result, these should be used with caution; usually, examination of sera by HI cannot be replaced if specific answers on protection efficacy are sought.

ELISAs aimed at the detection of antibodies against different proteins (e.g. NA, NS1, etc.) have been developed as alternatives, although currently there are no commercially available products with a validation dossier.

The following Table 5 presents an overview of some summarised characteristics of diagnostic methods currently available for surveillance purposes.

TABLE 5 Overview of some characteristics of diagnostic methods currently available for surveillance purposes.

Diagnostic test	Target	Follow-up	Sensitivity	Specificity	Time window for detection (days post-infection)	Surveillance purposes	Time to results ^a	Susceptibility to viral mutations affecting the method	Labour intensity	Costs	Positive result interpretation
Genome detectio	on				-						-
Conventional RT-PCR	H5/H7	Sequencing of the PCR product for confirmation and HP/LP differentiation	++	+++	From 2 to 5–7	Confirmation of clinical cases; population freedom from infection	4-6h	+	++	+	Presence of the targeted genetic material of H5/H7
RT-qPCR	M gene	End point	+++	+++	from 1–2 to 7–10	Confirmation of clinical cases; population freedom from infection; individual animal freedom from infection before movement	3h	+	+	++	Presence of the targeted genetic material of AIV
	HA cleavage site	End point	++	+++	from 1–2 to 7–10	Confirmation of viral pathotype ^b	3 h	+++	+	++	Presence of the targeted genetic material of HPAI/LPAI
	H5/H7	End point	+++	+++	from 1–2 to 7–10	Confirmation of HA subtype	3 h	++	+	+	Presence of the targeted genetic material of H5/H7
	NA (1–9)	End point	+++	+++	from 1–2 to 7–10	Confirmation of NA subtype	3 h	++	+	++	Presence of the targeted genetic material of NA(X)
	Vaccine vector (e.g. HVT-H5)	End point	+	+++	Since day 21–28 post- vaccination	Vaccination coverage	3 h	+	+	++	Presence of the targeted vector- specific genetic material
Antigen detectio	n										
Antigen-ELISA	NP	End point	+	++	3-6	Confirmation of clinical cases	<1h	+	+	+++	AIV
Lateral flow devices	NP	End point	+	++	3–6	Confirmation of clinical cases	<1h	+	(+)	++	AIV

(Continues)

TABLE 5 (Continued)

Diagnostic test	Target	Follow-up	Sensitivity	Specificity	Time window for detection (days post-infection)	Surveillance purposes	Time to results ^a	Susceptibility to viral mutations affecting the method	Labour intensity	Costs	Positive result interpretation
Detection of infe	ctious virus										
Embryonated chicken eggs	Whole virus	Typing by HI or RT-qPCR	+	+++	2–8	Confirmation of clinical cases	2–14 days	_	+++	++	Infectious virus presence
Cell cultures	Whole virus	Typing by HI or RT-qPCR	+	+++	2–6	Confirmation of clinical cases	2–10 days	-	++	+	Infectious virus presence
Antibody detecti	on										
ELISA	NP	Typing by HI	+++	++	>7–10 days to 3 months	Immune status in individual animals or populations; Prevalence of infection – DIVA surveillance	3 h	(+)	+	++	Presence of antibodies against the NP of AIVs
ELISA	H5/H7	End point	++	+++	> 10–14 days to 3 months	Immune status in individual animals or populations post-vaccination (Vaccination coverage)	3h	+	+	++	Presence of antibodies against the HA of H5/ H7 (including vaccine-derived immunity)
AGID ^c	Whole virus	Typing by HI	+	++	> 14 days to 3 months	Immune status in individual animals or populations	2 days	(+)	++	+	Presence of antibodies against AIVs
н	HA	Titration	++	+++	> 14 days to 3 months	Immune status in individual animals or populations post- vaccination (vaccine coverage) using vaccine homologous antigen; detection of field HPAIv exposure using field antigen	4 h	+	+++	+	Presence of antibodies against the HA of AIVs (subtype- specific)
V-NT	(HA) ^d	Titration	+++	+++	> 14 days to 3 months	Immune status in individual animals	3 days	+	+++	++	H5/H7

Abbreviation: I, Scaling equivalents; +++, high; ++, moderate; +, low; (+), very low.

^aMeant as time from the arrival of the sample in the lab to the availability of the results (i.e. time for sampling and transport are not included).

^bUse of RT-qPCR for pathotype identification should be used when epidemiological scenario is well known (e.g. circulation of 2.3.4.4b).

^cGalliform poultry only.

^dMost of neutralising epitopes of the AIV are located on the HA protein. Antibodies against the NA protein can also have neutralising ability.

Serological data from challenge experiments in vaccinated chickens, turkeys and ducks

Haemagglutination inhibition (HI)

Homologous HI titres represent values obtained with an antigen which is identical to the vaccine strain or which has a very short antigenic distance to the vaccine antigen (see Section 2.1.3.2 in Opinion Part 1 for estimation of the antigenic distance). Heterologous HI titres represent values obtained with any other test antigen with a larger antigenic distance. For the purpose of the analysis presented below if different antigens were tested by laboratories to analyse the same set of sera only the best performing one has been considered.

Chicken layers vaccinated with recombinant vectored vaccines showed robust immune response 42 days after vaccination at 1 day of age with mean homologous HI titres around 6–7 log₂ before exposure to challenge virus and when tested against an antigen antigenically similar to the H5 HA expressed by the vector (Table 6). When tested with the heterologous challenge virus, sera of vaccinated animals show little reactivity with the majority of titres being <4 log₂ (Table 6). Sera obtained from vaccinated birds after challenge showed increased heterologous HI titres against the challenge virus antigen, while lower increase in HI titres was evident against the vaccine virus antigen (Table 6). Similar results have been obtained with nucleic acid-based vaccines (i.e. RNA and DNA-based vaccines) in birds vaccinated at 1 day or 2 weeks of age, although variable prechallenge HI titres have been observed which could compromise sensitivity of surveillance activities aimed at the evaluation of level of immune response in vaccinated flocks (Table 6).

In vaccinated **turkeys**, immune responses clearly detectable by HI are reached at approximately 7 weeks of age irrespective of the different immunisation schemes and vaccines used. Higher HI titres can be obtained using antigens homologous to the vaccines used, and therefore should be preferred for assessing immunisation of birds (Table 7). Wide variability in immune response elicited has been observed, with higher antibody titres obtained in groups immunised with vaccination schemes that included an heterologous vaccine as boost (Table 7). Seroconversion (evaluated as the Delta-Log of HI titres in sera collected before or after challenge in vaccinated and infected birds) measured with the challenge antigen (field 2.3.4.4b strain) is generally a more reliable indicator of past infection with Delta values ranging from 5 to 7 log₂ compared to antigen homologous to the vaccine (Table 7).

Ducks vaccinated (at 10 and 28 days) with a subunit vaccine showed higher HI titres compared to ducks vaccinated (at 1 and 28 days) with an RNA-based vaccine in sera collected before challenge (i.e. 49 and 42 days, respectively) when tested with an antigen homologous to the vaccines (Table 8). Seroconversion (evaluated as the HI Delta-Log) was more evident in samples collected from convalescent birds, using the challenge strain as the antigen in HI assays (Table 8).

byC

Wiley Onl

on [15/11/2024]

app

TABLE 6 Antibody titres in **chicken layers** vaccinated (pre-challenge), and vaccinated and challenged (post-challenge) based on HI test with **homologous** and **heterologous** antigens.

	Vaccine administered						
	ExactVac – Vaxliant ENABLE ac	ljuvant (14, 28) ^a	RESPONS AI H5 (1, 28) ^a		Vectormune AI (1) ^a		
	(median and 95% CI) ^b		(median and 95% CI) ^b		(median and 95% CI) ^c		
HI titres	Homologous Heterologous		Homologous	Heterologous	Homologous	Heterologous	
'pre-challenge'	6.00 (2.72;10.55) ^d 3.00 (2.17;4.00) ^e	6.00 (6.00–7.00) ^d	6.00 (2.22;6.77) ^d	3.00 (0.00;3.77) ^d	7.75 (6.24;9.26) ^h	0.00 (0.00;1) ^h	
'delta log post-challenge'	0.00 (-4.27;5.50) ^f 0.00 (-1.00;2.55) ^f 6.50 (2.17;7.82) ^g		0.00 (-1.75;4.10) ^f	1.5 (-0.77;3.00) ^f	0.50 (–1.26;2.26) ⁱ	2.00 (0.00;7.52) ⁱ	
				λa			
	HVI-IBD-AIV-H5 haemagglutin	IN COBRA (I)	Nobilis Influenza H5N2 (8)	Exactivac – Vaxilant ENABL	E adjuvant (14)	
	(median and 95% CI) ^c		(median and 95% CI) ^c		(median and 95% CI) ^c		
HI titres	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous	
'pre-challenge'	6.00 (1.19;9.55) ^h	0.00 (0.00;2.52) ^h	0.00 (0;1) ^h	0 ^h	4.75 (0.0;7.5) ^h	0 ^h	
'delta log post-challenge'	0.25 (-0.5;7.1) ⁱ	2.00 (-0.26;6.76) ⁱ	6.00 (0.00;10.4) ⁱ	5.75 (0.00;10.91) ⁱ	-0.25 (-2.36;4.00) ⁱ	5.50 (0.00;7.85) ⁱ	

^aAge in days at time of vaccination.

^bData source: Belgium – NRL for AI.

^cData source: The Netherlands – NRL for Al.

 d,e Days at testing: d = 41 days, e = 174 days.

^{fg}Delta-Log between HI titres in sera collected before (41 or 174 days of age) and after (56 or 189 days of age) the challenge in vaccinated animals f = 56–41 days, g = 189–174 days. Challenge performed at 42 or 175 days of age.

^hDays at testing: 42 days.

¹Delta-Log between HI titres in sera collected before (42 days of age) and after (77 days of age) the challenge in vaccinated animals. Challenge performed at 56 days of age.

TABLE 7 Antibody titres in turkeys vaccinated (pre-challenge), and vaccinated and challenged (post-challenge) based on HI test with homologous and heterologous antigens.

	Vaccine(s) admin	istered								
	Volvac B.E.S.T. Al+ND (1,28) ^a		Vectormune AI (1) ^a	HVT-IBD-AIV-H 'COBRA' (1) ^a	5 haemagglutinin	ExactVac – Vaxliant	ENABLE adjuvant (1) ^a	Vectormune AI (1) ^a ENABLE adjuvant (2	+ ExactVac – Vaxliant (8) ^a
								ExactVac – Vaxliant ENABLE adjuvant (28) ^a		
	(median and 95%	CI)	(median and 95% CI)		(median and 95% CI)		(median and 95% CI)		(median and 95% CI)	
HI titres	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous
'pre-challenges'	5.00 (3.17;7.00) ^b 4.50 (3.00;6.82) ^c	1.00 (1.00;2.00) ^{b,c}	4.50 (3.17;6.82) ^c	2.00 (1.00;3.82) ^c	na	3.00 (1.17;4.00) ^c	1.50 (1.00;5.47) ^b 2.00 (1.00;5.47) ^c	1.00 (1.00;4.30) ^b 1.00 (1.00;3.47) ^c	3.50 (2.00;4.00) ^b 5.00 (3.00;5.00) ^c	1.00 (1.00;1.82) ^b 1.50 (1.00;2.82) ^c
'delta log post-challenges'	5.00 (5.00;5.95) ^d 6.00 (5.05;6.00) ^e	5.00 (4.05;7.85) ^{d,e}	3.00 (3.00;5.00) ^d	5.00 (2.3;6.0) ^e	na	8.00 (7.00;9.65) ^e	5.50 (4.07;6.92) ^d 5.00 (4.05;5.95) ^e	5.00 (–) ^d 5.50 (5.02;5.97) ^e	4.50 (3.00;7.65) ^d 3.50 (1.17;6.65) ^e	5.50 (1.35;8.82) ^d 5.00 (1.35;7.82) ^e
	Vectormune AI (1) ^a AI + ND (28) ^a	+ Volvac B.E.S.T.	HVT-IBD-AIV-H5 haemagglutinin 'COBRA' (1) ^a + Volvac B.E.S.T. AI + ND (36) ^a		HVT-IBD-AIV-H5 haemagglutinin 'COBRA' (1) ^a + ExactVac – Vaxliant ENABLE adjuvant (36) ^a		Vectormune AI (1) ^a + Vaxigen Flu H5N8 (36) ^a		HVT-IBD-AIV-H5 haemagglutinin * 'COBRA' (1) ^a + Vaxigen Flu H5N8 (36) ^a	
	(median and 95% C	:1)	(median and 95%	o CI)	(median and 95%	o CI)	(median and 95% C	:1)	(median and 95	% CI)
HI titres	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous	Homologous	Heterologous
'pre-challenges'	7.00 (5.35;8.82) ^b 7.00 (5.17;8.82) ^c	1.50 (1.00;3.00) ^b 2.00 (1.00;3.82) ^c	8.00 (7.17;10.65) ^c	4.00 (2.17;5.00) ^b 5.00 (4.00;6.82) ^c	na	2.00 (2.00;3.82) ^b 3.00 (2.00;4.00) ^c	4.00 (3.00;5.00) ^b 6.00 (4.35;7.00) ^c	2.00 (–) ^b 6.00 (4.35;7.00) ^c	5.50 (4.00;6.00) ^c	2.50 (2.00;4.82) ^b 5.00 (4.00;6.00) ^c
'delta log post- challenges'	3.00 (2.00;4.82) ^d 3.00 (3.00;4.00) ^e	7.00 (3.52;7.00) ^d 6.00 (3.35;7.00) ^e	2.00 (1.17;4.00) ^e	5.50 (4.17;7.65) ^d 3.50 (3.00;6.00) ^e	na	7.00 (5.15;7.85) ^d 6.00 (5.00;7.00) ^e	4.50 (4.00;6.00) ^d 3.00 (2.00;3.00) ^e	6.00 (4.35;7.00) ^d 4.50 (3.17–6.82) ^e	4.00 (3.00;5.82) ^e	6.00 (4.17;7.82) ^d 4.00 (3.00;5.82) ^e

Data source: Italy – EURL for AI.

^aAge in days at time of vaccination.

^{b,c}Days at testing: b=43 days, c=50 days.

d,eDelta-Log between HI titres of sera collected before (42 or 50 days of age) and after (64 days of age) the challenge in vaccinated animals d=64-43 days, e=64-50 days. For all groups, challenge was performed at 50 days of age.

on [15/11/2024]

TABLE 8 Antibody titres in **ducks** vaccinated (pre-challenge), and vaccinated and challenged (post-challenge) based on HI test with **homologous** and **heterologous** antigens. Groups vaccinated with Duck H5-SRV vaccine[®] were challenged at 46 or 75 days of age. Groups vaccinated with Volvac B.E.S.T. Al+ND were challenged at 53 or 81 days of age.

	Vaccine administered							
	Duck H5-SRV vaccine® (1, 2	28) ^a	Volvac B.E.S.T. AI + ND (10, 28) ^a					
	(median and 95% CI)		(median and 95% CI)					
HI titres	Homologous	Heterologous	Homologous	Heterologous				
'pre-challenge'	1.00 (–) ^{b,c}	1.00 (1.00;2.77) ^b 1.00 (–) ^c	7.00 (5.00;7.00) ^d 5.00 (1.52;5.00) ^e	1.00 (–) ^{d,e}				
'delta log post-challenge'	4.00 (3;5) ^f 6.0 (3.45;7.00) ^g	5.00 (2.22;6.00) ^f 6.5 (4.45;7.00) ^g	1.50 (0.22;3.77) ^h 5.00 (4.00;8.47) ⁱ	3.50 (2.22;4.77) ^h 4.00 (3.00;4.82) ⁱ				

Data source: France - NRL for AI.

^aAge in days at time of vaccination.

 b,c,d,e Days at testing: b=42 days, c=71 days, d=49 days, e=77 days.

 $f_{g,h,i}$ Delta-Log between HI titres of sera collected before (42, 49, 71 or 75 days of age) and after (60, 67, 89, 95 days of age) the challenge in vaccinated animals f=60-42 days, g=89-71 days, h=67-49 days, i=95-77 days. Groups vaccinated with Duck H5-SRV vaccine[®] were challenged at 46 or 75 days of age. Groups vaccinated with Volvac B.E.S.T. Al+ND were challenged at 53 or 75 days of age.

<u>ELISA</u>

NP-based ELISA in either an indirect or competitive format will be of importance in surveillance activities aimed at the detection of past AI-infection in flocks vaccinated with DIVA compatible vaccines. Regardless of the species tested, an indirect ELISA assay showed higher Se and Sp when compared to a competitive ELISA assay. Antibodies against the NP were elicited by HPAI replication in 100% of the infected birds, 14 days after infection in chickens and turkeys and 10 days after infection in mule ducks (Table 9). Earlier time points after exposure to the virus have been assessed for turkeys and ducks (Table 9), but showed lower Se and Sp values.

Competitive NP-ELISA assays evaluated under experimental conditions showed lower Se and Sp values than indirect ELISAs at all tested time points following infection in both chickens and ducks (Table 9). No values for turkeys were available for this type of ELISA at the time of writing.

TABLE 9	ensitivity (Se) and specificity (Sp) of the indirect and competitive NP-ELISA at 4, 7, 10, 14 and 21 days post
infection ([Pl) in chicken layers , turkeys , ducks and mule ducks vaccinated with a DIVA compatible vaccine.

	Poultry species												
	Chicken layers		Turkeys		Ducks		Mule ducks						
DPI	Indirect NP- ELISA (exact 95% CI)	Competitive NP-ELISA (exact 95% CI)	Indirect NP- ELISA (exact 95% Cl)	Competitive NP-ELISA (exact 95% CI)	Indirect NP-ELISA (exact 95% CI)	Competitive NP-ELISA (exact 95% CI)	Indirect NP-ELISA (exact 95% CI)	Competitive NP-ELISA (exact 95% Cl)					
4	na	na	na	na	na	Se = 10% (0.25%-44.5%) Sp = 100% (71%-100%)	Se = 28.95% (15.42%-45.9%) Sp = 100% (90.7%-100%)	na					
7	na	na	Se=60% (41%-77%) Sp=100% (63%-100%)	na	na	Se=60% (26.2%-87.8%) Sp=100% (71%-100%)	Se=97.36% (86.2%-99.9%) Sp=100% (90.7%-100%)	na					
10	na	na	na	na	na	Se=70% (34.7%-93.3%) Sp=100% (71%-100%)	Se=100% (90.5%-100%) Sp=100% (90.7%-100%)	na					
14	Se = 100% (59%-100%) ^a Sp = 92.3% (64%-99.8%) ^a	Se = 75% (53.2%-90.2%) ^a $Sp = 79.5%$ (64.7%-90.2%) ^a	Se=100% (88%-100%) Sp=100% (63%-100%)	na	na	Se=70% (34.7%-93.3%) Sp=100% (71%-100%)	Se = 100% (90.5%-100%) Sp = 100% (90.7%-100%)	na					
21	na	Se = 85% (62.10%-96.7%) ^b Sp = 100% (88.43%-100%) ^b	na	na	na	na	na	na					

Data source: NRL for AI in Belgium, France, Italy, the Netherlands.

Abbreviation: na, not available.

^aData from Belgium.

^bData from the Netherlands.

3.1.1.5 | Monitoring of clinical signs and production parameters

The absence of mortality and/or overt clinical disease is the basic effect of HPAI vaccination, although variations in vaccine efficacy/effectiveness and the characteristics of the circulating viruses may cause different degrees of clinical protection (Table 3). As shown in the previous Opinion, absence of clinical signs and an increase in mortality in vaccinated flocks are poor indicators of the absence of viral circulation (EFSA AHAW Panel, 2023). Similarly, more refined parameters, which have been proven effective as indicators of ongoing disease in unvaccinated poultry flocks, such as reduced weight gain, reduced consumption of water or feed, drop in egg production, etc. bear a low predictive value of HPAIV incursions in vaccinated flocks. Clinical examination is therefore not a reliable indicator in surveillance activities aimed at early identification of HPAIV incursions in vaccinated flocks both in emergency and in preventive vaccination scenarios.

However, it must be considered that under realistic field conditions, as it has been extensively demonstrated, a certain percentage of vaccinated animals may fail to reach expected protective immunity levels. This might be due to a plethora of factors including vaccination failures or problems associated with the individual vaccinated bird such as the presence of immunosuppressive agents, nutritional diseases, genetic characteristics, etc. Therefore, there may be a certain percentage of birds in vaccinated flocks, which would show clinical signs of the disease and there may be even an increase in mortality associated with the HPAIV infection (Huynh et al., 2019; Tarigan et al., 2018). These 'fallen' animals deserve special attention in further surveillance investigations with the use of sensitive laboratory diagnostic methods (i.e. 'bucket sampling'). Similarly, if an escape-mutant arises or is introduced from outside this would be expected to spread despite vaccination, and clinical signs and increased mortality are the sequelae.

In peri-vaccination zones or in unvaccinated flocks within a vaccination zone, passive 'syndrome' surveillance based on clinical signs and production parameters would still represent a valuable method for the early detection of HPAI outbreaks.

3.1.1.6 | Use of unvaccinated sentinel birds

The use of unvaccinated sentinels in vaccinated flocks to determine if flocks have been exposed to infections has been a common strategy of disease detection (Bouma et al., 2008). In theory, sentinels could provide a sensitive measure of exposure to infection with HPAIV of a vaccinated flock, since the birds are completely susceptible to infection and would show clinical signs and mortality typical of HPAI. Sentinels can then be routinely tested for HPAIV infection with appropriate serological or virological methods (Suarez, 2005). However, several practical problems have been found to hamper the use of sentinels in the surveillance strategy in HPAI-vaccinated flocks and areas, and therefore, the sentinel bird approach has been abandoned in countries with long-standing vaccination experience such as Indonesia and Hongkong (Harder et al., 2023):

- 1. Without good compliance of farmers and a proper management system, the presence of unvaccinated sentinel birds in the flock may in fact function as the 'gap in the door of protection' since if exposed to the virus they can get infected and spread HPAIV in the flock and to other farms.
- 2. It is logistically challenging to identify sentinel animals in large flocks if sentinels are of the same species (conspecific) as the vaccinated part of the flock. Conspecific sentinels would enable an appropriate diffusion in the vaccinated flock. If they are from a different species (e.g. sentinel chickens in a duck flock) problems arise with compatibility of the different species leading to a separation of sentinels and vaccinees rather than observing them mixing in the flock.
- 3. In certain species, lateral transmission of replication-competent live-vectored HPAI vaccines would eventually immunise unvaccinated sentinels (e.g. turkeys vaccinated with a live HVT-vectored vaccine).

In addition, where LPAI viruses are circulating in the vaccination area, infection in sentinels will compromise the interpretation of virological and serological monitoring results.

The current scenarios evaluated as part of this Scientific Opinion therefore do not envisage the use of sentinels due to the above-mentioned problems and due to the increased risk of flock infection associated with the use of fully susceptible unvaccinated sentinels.

3.1.1.7 | Use of environmental samples for surveillance

The current sampling schemes laid down in Delegated Regulation (EU) No 2023/361 Annex XIII require surveillance of vaccinated flocks to exclude a prevalence of HPAIV infection in the flock of 5% with 95% confidence. This amounts to at least 60 samples for the average commercial holding when oropharyngeal and cloacal are taken as prescribed. Such samples are to be obtained every 2 weeks (emergency protective vaccination) or every 4 weeks (preventive vaccination) until the last vaccinated bird leaves the holding. In practice, several problems arise with these prescriptions:

- Catching the birds repeatedly troubles the flock and may create animal welfare problems.
- In weighty, long-lived poultry such as fattening turkeys or geese parents the catching and restraining of animals up to 20 kg body weight requires several stable hands and means a heavy physical workload.
- Obtaining the samples by an official veterinarian will be costly and time and human resources are often limited.
- Transporting of 60 samples per round to a laboratory requires a cool chain and appropriate logistics.
- RNA extraction and RT-qPCR analysis of samples is costly, even if pools of up to 10 samples could be formed.

Therefore, alternative sampling schemes and sample matrices are sought that would contribute to reducing workload during sampling, reduce costs for analysis and reduces stress and strains for the individual animals. However, such alternatives must have been validated to prove similar efficacy (i.e. detecting a design prevalence of 5/95) like the 60-swab scheme.

Recent investigations of environmental samples from AIV infected flocks have documented that several matrices are promising with respect to detection of viral genomes by RT-qPCR. In particular, water and biofilm samples from drinkers in turkey flocks (Muñoz-Aguayo et al., 2019), dust samples from walls and air samples (Filaire et al., 2022) have been analysed. Data showed that virus detection in such environmental samples was as sensitive as in conventional swab samples from animals. However, these investigations were all carried out in unvaccinated infected flocks. Systematic validation should clarify whether these promising data can be transposed to vaccinated flocks, since HPAI virus excretion in infected vaccinated animals is shortened in time and reduced in load. This may reduce virus loads in environmental sample matrices to a stronger extent than in infected non-vaccinated herds.

3.1.2 | Surveillance in vaccinated populations

Building upon the context of the previous Opinion (ToRs 1 and 2) we focused here on evaluating the effectiveness of surveillance in the three productive poultry groups targeted for vaccination, namely chicken layers (the Netherlands), ducks (France) and turkeys (Italy), according to the surveillance strategies as indicated in Table 2.

3.1.2.1 | Surveillance for early detection at the establishment level for emergency vaccination (strategy E1)

In the context of surveillance for early detection, $R_h < 1$ was chosen as indicating effectiveness. The reason was that it takes time to vaccinate flocks, so in the initial stages of an epidemic, despite the start of vaccination, R_h remains above 1 for a certain time lag and between-farm transmission may occur as shown in the previous Opinion (EFSA AHAW Panel, 2023). Surveillance is key to detecting infected flocks and the removal of those will contribute together with vaccination to reduce the number of susceptible animals/establishments and consequently to bring faster R_h below 1. Because of the methodology used in this assessment R_h is replaced by R_s , but the interpretation of R_s is equivalent to R_h .

The results of the assessment on surveillance strategies for early detection of HPAIV for emergency vaccination are shown in Tables 10A–12B, grouped by poultry species (chicken layers, ducks and turkeys) and for different surveillance strategies: passive reporting (i.e. the level of mortality at which farmers tend to report their suspicions calculated based on mortality data and reported suspicion dates, see Table 3), mortality thresholds (i.e. inferred level of mortality beyond which farmers have to report their suspicions) based on the studies by Gonzales and Elbers (2018), Elbers and Gonzales (2021) and Gonzales et al. (2023) (see Table 3), and active surveillance based on sampling dead or live birds to be tested by qPCR, or live birds to be tested by serology.

Different combinations of sampling intervals and samples sizes are presented as results; only those surveillance strategies whose combination of sampling interval and sample sizes resulted in a probability to escape detection below 0.01 at the flock level for more than 95% of the simulated within-flock outbreaks are considered effective to early detect HPAIV. The reason is that undetected outbreaks led to an underestimation of R_h and of the time between introduction and detection. Note that surveillance strategies for testing dead birds included randomly sampling all up to a number of 5, 10 and 15 dead birds. If fewer dead birds were available than these required sample sizes, all available dead birds were tested, which is why a minimum sample size was not provided.

Surveillance in chicken layers

Table 10A reports results for flocks of 20,000 chicken layers. However, the model outcomes remained consistent for flock sizes of 6000 and 60,000 birds; simulations for these flock sizes produced similar results as those presented in Table 10A. The results reported (Table 10A) indicate that sampling live birds for PCR testing or serology did not result in detecting 99% of the outbreaks in \ge 95% of the simulations. Therefore, the strategy of 'sampling live birds (n = 60)' every 14 days as recommended by the EU Regulation, when emergency vaccination is applied (scenario E1) would not be effective for early detection. The proposed strategy of weekly sampling all dead birds up to a number of five achieved early detection at a median (95% confidence intervals (Cl)) time of 20 (14–31) days post-introduction and a reduction of R_s to 0.13 (95% CI: 0.10–0.16). The median (95% CI) number of dead birds likely to be found at the time of sampling and detection of an infected flock was 6 (4–10). With longer sampling intervals, such as every 14 or 21 days, all dead birds up to a number of 10 or 15, respectively, need to be sampled to enable that >99% of the outbreaks would be detected in 95% of the simulations, with median times of detection of 21 and 22 days post-introduction, respectively. The median number of dead birds likely to be found at the time of sampling and detection than testing live birds and also fewer resources are needed. Furthermore, longer sampling intervals compared with the weekly sampling can be considered if more dead birds are collected (up to a number of 15 for a 3-week interval) (Table 10A).

The strategies based on 'passive reporting' or on 'mortality thresholds' would not be effective for early detection, with median detection days of 31 (25–43) and 28 (22–41), respectively, and $R_{h/s} > 1$.

TABLE 10A Efficacy of different combination of sampling types, sample size, sampling frequency and diagnostic test for early detection (defined as reduction of $R_{s} < 1$) of vaccinated-infected flocks, strategy E1. These simulations were done for a flock of **20,000^a chicken layers** and assuming a case fatality equal to 0.2, a qPCR Se=0.99 and a Serology (ELISA) Se=0.9. Results are reported only for effective surveillance strategies, i.e. where the proportion of outbreak simulations with the probabilities of escaping detection below 1% is above 95%; otherwise, these are left empty.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^b	Detection time as days post introduction (median (2.5–97.5 Cl))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 Cl))	R _h /R _s (reproduction number) (median (2.5–97.5 Cl))
Passive reporting (reference)				31 (25–43)	3.93 (3.44–4.5)	2.16 (1.86–2.46)	1.4
Mortality threshold (0.13%)				28 (22–39)	2.35 (2.01–2.75)	1.26 (1.06–1.49)	1.09 (1.04–1.1)
Dead birds (qPCR)	≤5	7 14 21 30	99% 90% 51% 0%	20 (14–31)	0.34 (0.25–0.43)	0.18 (0.11–0.24)	0.13 (0.1–0.16)
	≤10	7 14 21 30	99% 98% 94% 84%	18 (13–30) 21 (15–33)	0.26 (0.19–0.34) 0.44 (0.35–0.56)	0.14 (0.08–0.19) 0.23 (0.15–0.31)	0.1 (0.08–0.13) 0.17 (0.15–0.2)
	≤ 15	7 14 21 30	99% 99% 97% 92%	18 (13–30) 20 (15–32) 22 (16–34)	0.26 (0.19–0.33) 0.41 (0.32–0.52) 0.56 (0.45–0.71)	0.13 (0.08–0.19) 0.21 (0.15–0.29) 0.3 (0.21–0.39)	0.1 (0.08–0.13) 0.16 (0.14–0.19) 0.22 (0.19–0.26)
Live birds (qPCR)	60 120	14 30 14 30	72% 30% 89% 69%				
Live birds (serology)	60	14 30	47% 9%				

^aFlock sizes of 6000 and 60,000 chicken layers were also assessed. No differences in the results regarding reduction of $R_s < 1$ were observed.

^bAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99).

The results were different when a flock size of 2500 birds was used to represent small flocks. For these flocks, none of the strategies evaluated were effective, since the requirement to detect 99% of the outbreaks in 95% of the simulations was not met (Table 10B); in these flock size, sampling all birds found dead up to a number of 10 birds weekly gave detection of 99% of the outbreaks in 90% of the simulations. For flocks \geq 3000 and < 6000 weekly sampling of all up to a number of 10 or 15 dead birds resulted in an effective early detection in \geq 95% of the simulations (Data not shown).

TABLE 10B Efficacy of different combination of sampling types, sample size, sampling frequency and diagnostic test for early detection (defined as reduction of $R_s < 1$) of vaccinated-infected flocks, strategy E1. These simulations were done for a flock of **2500 chicken layers** and assuming a case fatality equal to 0.2, a qPCR Se = 0.99 and a Serology (ELISA) Se = 0.9. Results are reported only for effective surveillance strategies, i.e. where the proportion of outbreak simulations with the probabilities of escaping detection below 1% is above 95%; otherwise, these are left empty.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^a	Detection time as days post introduction (median (2.5–97.5 Cl))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 Cl))	R _h /R _s (reproduction number) (median (2.5–97.5 Cl))
Passive reporting (reference)				21 (16–33)	4.09 (3.01–5.35)	2.19 (1.59–2.91)	1.4
Mortality threshold (0.13%)				15 (10–26)	1.01 (0.6–1.56)	0.48 (0.24–0.81)	1.27 (0.44–1.41)
Dead birds (qPCR)	≤ 5 ≤ 10 ≤ 15	7 14 21 30 7 14 21 30 7 14 21 30	87% 50% 2% 0% 90% 59% 18% 90% 80% 69%				
Live birds (qPCR)	60 120	14 30 14 30	69% 25% 85% 67%				
Live birds (serology)	60	14 30	42% 7%				

^aAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99).

Surveillance in ducks

In flocks of 6000 ducks or larger, the results reported in Table 11A indicate that the strategy based on 'mortality thresholds' achieved the earliest detection, at a median of 17 (13–26) days post-introduction, with a R_s < 1. Note that the strategy based on 'passive reporting' is ineffective for early detection, with a median detection day of 23 (19–32) and R_h > 1. Sampling dead birds at weekly intervals showed in more than 95% of the simulations that they would be effective for early detection, with a detection time of 14–15 days and R_s < 1 irrespective of collecting all up to a number of 5, 10 or 15 dead birds. Similarly, biweekly sampling of all up to a number of 10, or every 3 weeks up to 15 dead birds would be effective for early detection. The median number of dead birds likely to be found at the time of sampling and detection of an infected flock would be 4 (2–9) when sampling weekly, 6 (4–10) when sampling every 2 weeks and 8 (5–12) when sampling every 3 weeks.

The strategy of 'sampling live birds (n = 60)' every 14 days as recommended by the EU Regulation for emergency vaccination (E1), or testing 60 birds serologically, would be effective for early detection with a median detection time of 17 days (13–25) and $R_s < 1$. This is also in contrast to chicken layers (both small and larger flocks), likely because of the higher transmission rate in duck flocks. Of note, sampling live ducks implies challenges in terms of welfare consequences as the animals undergo catching and handling, as well as in term of logistics and resources needed as more samples would be collected and tested compared with those required if dead birds are sampled. **TABLE 11A** Efficacy of different sample types, sample size, sampling frequency and diagnostic test combination for early detection (defined as reduction of $R_s < 1$) of vaccinated-infected flocks, strategy E1. These simulations were done for a flock \ge **6000 ducks**^a and assuming a case fatality equal to 0.05 and qPCR sensitivity of 0.99 and ELISA sensitivity of 0.90. Results are reported only for effectiveness surveillance strategies, i.e. those where the proportion of outbreak simulations with the probabilities of escaping detection below 1% is above 95% otherwise these are left empty.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^b	Detection time as days post- introduction (median (2.5–97.5 CI))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 CI))	R _h /R _s (reproduction number) (median (2.5−97.5 Cl))
Passive reporting (reference)				23 (19–32)	20.5 (18.9–22.3)	33.31 (29.26–37.26)	1.8
Mortality threshold (0.17%)				17 (13–26)	5.84 (4.83–7.22)	6.13 (4.9–7.28)	0.62 (0.49–0.63)
Dead birds (qPCR)	≤5	7 14 21 30	98% 70% 0 ^b 0%	15 (11–24)	3.09 (2.51–3.86)	3.09 (2.33–3.88)	0.21 (0.18–0.26)
	≤10	7 14 21 30	99% 97% 89% 36%	14 (11–23) 16 (12–25)	2.35 (1.87–3.02) 4.29 (3.48–5.19)	2.35 (1.74–3) 4.37 (3.34–5.38)	0.17 (0.14–0.19) 0.3 (0.26–0.34)
	≤15	7 14 21 30	99% 98% 96% 89%	14 (10–23) 16 (12–24) 17 (13–26)	2.33 (1.86–2.96) 3.93 (3.18–4.8) 5.5 (4.54–6.68)	2.3 (1.72–2.99) 3.99 (2.97–4.94) 5.75 (4.43–6.97)	0.16 (0.14–0.19) 0.28 (0.24–0.32) 0.39 (0.35–0.44)
Live birds (qPCR)	60 90	14 30 14 30	97% 44% 98% 93%	17 (13–25) 15 (12–24)	4.95 (4.03–5.98) 3.34 (2.64–4.11)	5.05 (3.9–6.34) 3.33 (2.49–4.17)	0.35 (0.32–0.39) 0.23 (0.21–0.26)
Live birds (serology)	60	14 30	97% 93%	17 (13–25)	5.16 (4.25–6.28)	5.34 (4.17–6.62)	0.36 (0.29–0.44)

^aFlock size of 15,000 ducks was also assessed. No differences in the results were observed.

^bAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99). Results for scenarios where this probability is 0 are results for flocks where the probability of escaping detection was > 0.01 and hence not considered an effective surveillance scenario for early detection.

In flocks of 2500 ducks, the results reported in Table 11B indicate that the strategy based on 'mortality thresholds' achieved the earliest detection, at a median of 10 (7–18) days post-introduction, with an $R_s < 1$. Sampling dead birds at weekly intervals showed in more than 95% of the simulations that they would be effective for early detection, with a detection time of 14–15 days and $R_s < 1$ irrespective of collecting all up to a number of 5, 10 or 15 dead birds. Different from the simulations made for flocks with ≥ 6000 birds, sampling intervals longer than 7 days would not be effective (<95% of the simulations detected > 99% of the outbreaks). The median number of dead ducks likely to be found at the time of sampling and detection of an infected flock would be around 3 (2–6) when sampling up to five dead birds, or 3 (2–5) when sampling up to 10 (sampling up to 15 birds does not improve further the efficacy). This is in contrast to surveillance in small flocks of chicken layers, where weekly sampling of dead birds could only reach detection of 90% of the simulations. This contrast is likely to be due to the faster transmission rate expected in ducks than in chickens.

TABLE 11B Efficacy of different sample types, sample size, sampling frequency and diagnostic test combination for early detection (defined as reduction of $R_s < 1$) of vaccinated infected flocks, strategy E1. These simulations were done for a flock of **2500 ducks^a** and assuming a case fatality equal to 0.05 and qPCR sensitivity of 0.99 and ELISA sensitivity of 0.90. Results are reported only for effectiveness surveillance strategies, i.e. those where the proportion of outbreak simulations with the probabilities of escaping detection below 1% is above 95% otherwise these are left empty.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^b	Detection time as days post introduction (median (2.5–97.5 Cl))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 Cl))	R _h /R _s (reproduction number) (median (2.5–97.5 Cl))
Passive reporting (reference)				21 (17–30)	20.76 (18.37–23.21)	34.5 (29.9–38.74)	1.8
Mortality threshold (0.17%)				10 (7–18)	1.29 (0.77–1.97)	1.21 (0.72–1.73)	0.63 (0.1–0.76)
Dead birds (qPCR)	≤5	7 14 21 30	96% 33% 0% 0%	15 (11–24)	5.83 (4.64–7.25)	6.13 (4.4–7.56)	0.4 (0.34–0.48)
	≤10	7 14 21 30	97% 92% 12% 0%	14 (11–24)	5.42 (4.37–6.81)	5.63 (4.12–7.11)	0.37 (0.31–0.44)
	≤ 15	7 14 21 30	97% 92% 20% 0%	14 (11–24)	5.42 (4.37–6.81)	5.63 (4.12–7.11)	0.37 (0.31–0.44)
Live birds (qPCR)	60	14 30	97% 48%	14 (11–23)	4.91 (3.98–6.16)	5.04 (3.67–6.72)	0.34 (0.31–0.37)
	90	14 30	98% 93%	13 (9–22)	3.33 (2.51–4.34)	3.3 (2.34–4.38)	0.23 (0.2–0.25)
Live birds (Serology)	60	14 30	97% 93%	14 (11–24)	5.22 (4.2–6.61)	5.36 (3.9–7.02)	0.36 (0.27–0.45)

^aThis was the median size for flocks smaller than 4000 in France.

^bAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99).

Surveillance in turkeys

Table 12A reports results for flocks of 18,000 turkeys. However, the model outcomes remained consistent for flock sizes of 6000 and 60,000 birds; simulations for these flock sizes produced similar results as those presented in Table 12A. The results reported in Table 12A indicate that sampling all dead birds up to a number of five every week is effective to early detect HPAIV. The earliest detection is achieved using all up to a number of 15 dead birds at weekly intervals, with a median of 15 (10–27) days post-introduction and with the lowest R_s at 0.11 (0.09–0.13). The median number of dead birds likely to be found at the time of sampling and detection of an infected flock would be 14 (7–22). Less frequent sampling intervals, such as every 14 days with a sample size of all up to a number of 15 dead birds and every 21 days with a sample size of up to a number of 20 birds, remains effective for early detection and limited transmission ($R_s < 1$). The median number of dead birds likely to be found at the time of sampling and detection of an infected flock would be 16 (11–23) when sampling birds likely to be found at the time of sampling and detection of an infected flock would be 16 (11–23) when sampling birds likely to be found at the time of sampling every 3 weeks.

The strategies involving 'sampling live birds' (qPCR or serology) (n = 60 or 90) would not be effective, including the one recommended by the EU Regulation for emergency vaccination (E1), e.g. 60 live birds every 14 days.

The strategies based on 'passive reporting' or on 'mortality thresholds' would not be effective for early detection, with median detection days of 27 (21–38) and 24 (19–34), respectively, and $R_s > 1$.

TABLE 12A Efficacy of different sample types, sample size, sampling frequency and diagnostic test combination for early detection (defined as reduction of $R_s < 1$) of vaccinated partially protected infected flocks, strategy E1. These simulations were done for a flock of **18,000 turkeys**^a and assuming a case fatality equal to 0.2. A qPCR sensitivity of 0.99 and an ELISA sensitivity of 0.9. Results are reported only for effective surveillance strategies, i.e. those where the proportion of outbreak simulations with the probabilities of escaping detection below 1% is above 95% otherwise these are left empty.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^b	Detection time as days post introduction (median (2.5–97.5 CI))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 Cl))	R _h /R _s (reproduction number) (median (2.5–97.5 Cl))
Passive reporting (reference))			27 (21–38)	4.1 (3.51–4.79)	3.31 (2.83–3.83)	1.6
Mortality threshold (0.21%)				24 (19–34)	2.5 (2.07–2.96)	1.97 (1.64–2.34)	1.42 (1.68–1.04)
Dead birds (qPCR)	≤5	7 14 21 30	95% 66% 5% 0 ^b	19 (14–31)	0.66 (0.53–0.81)	0.52 (0.38–0.66)	0.28 (0.23–0.34)
	≤10	7 14 21 30	99% 94% 77% 59%	16 (11–28)	0.33 (0.25–0.44)	0.25 (0.16–0.36)	0.14 (0.11–0.18)
	≤ 15	7 14 21 30	99% 98% 94% 87%	15 (10–27) 18 (12–29)	0.26 (0.18–0.35) 0.46 (0.35–0.6)	0.2 (0.12–0.29) 0.35 (0.25–0.48)	0.11 (0.09–0.14) 0.2 (0.17–0.24)
	≤20	7 14 21 30	99% 98% 96% 93%	15 (10–27) 17 (12–29) 19 (13–30)	0.25 (0.17–0.33) 0.42 (0.31–0.53) 0.59 (0.48–0.75)	0.18 (0.12–0.27) 0.31 (0.22–0.43) 0.45 (0.33–0.6)	0.1 (0.08–0.13) 0.18 (0.15–0.21) 0.26 (0.22–0.3)
Live birds (qPCR)	60 90	14 30 14 30	70% 21% 82% 55%				
Live birds (serology)	60	14 30	61% 22%				

^aFlock sizes of 6000 and 60,000 turkeys were also assessed. No differences in the results were observed.

^bAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99).

In flocks < 6000 neither testing dead birds, nor testing live birds would be effective for early detection. Different to chicken layers but similar to small duck flocks, using a daily mortality threshold to raise suspicions and issuing guidelines for sampling and detection could result in an effective early detection (Table 12B).

TABLE 12B Efficacy of different sample types, sample size, sampling frequency and diagnostic test combination for early detection (defined as reduction of $R_s < 1$) of vaccinated partially protected infected flocks, strategy E1. These simulations were done for a flock of **2500 turkeys** and assuming a case fatality equal to 0.2. A qPCR sensitivity of 0.99 and an ELISA sensitivity of 0.9. Results are reported only for effective surveillance strategies, i.e. those where the proportion of outbreak simulations with the probabilities of escaping detection below 1% is above 95% otherwise these are left empty.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^a	Detection time as days post- introduction (median (2.5–97.5 Cl))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 CI))	R _h /R _s (reproduction number) (median (2.5–97.5 Cl))
Passive reporting (reference)				19 (13–29)	4.24 (3.18–5.56)	3.37 (2.52–4.29)	1.6 (1.6–1.6)
Mortality threshold (0.21%)				5 (3–9)	0.16 (0.04–0.28)	0.04 (0–0.15)	0.89 (0.06–0.98)
Dead birds (qPCR)	≤5	7	11%				
		14	11%				
		21	9%				
		30	0%				
	≤10	7	11%				
		14	11%				
		21	11%				
		30	10%				
	≤ 15	7	11%				
		14	11%				
		21	11%				
		30	11%				
	≤20	7	11%				
		14	11%				
		21	11%				
		30	11%				
Live birds (qPCR)	60	14	64%				
		30	12%				
	90	14	77%				
		30	43%				
Live birds	60	14	44%				
(serology)		30	6%				

^aAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99).

Surveillance in peri-vaccination zone (only for emergency protective vaccination)

According to Article 9 1(c)(ii) Delegated Regulation (EU) No 2023/361, HPAIV surveillance should be applied in a perivaccination zone established for emergency vaccination to detect any possible new outbreaks. The risk assessment carried out by EFSA in the opinion on control measures for Category A diseases (EFSA AHAW Panel, 2021) (pp. 35–38) has been used to assess the **radius of the peri-vaccination zone**. Given that the 3-km recommended radius for an emergency vaccination zone (EFSA AHAW Panel, 2023) corresponds to the 3-km radius of a classical 'protection zone' following HPAIV outbreaks in poultry, the peri-vaccination zone always overlaps with the 'surveillance zone' that, according to Delegated Regulation (EU) No 2020/687 should have a 10-km minimum radius.

In the EFSA Opinion on Category A diseases, a simple kernel based on three parameters was used on unvaccinated population to estimate the radius of a surveillance zone based on the probability of HPAIV transmission beyond given distances (jump spread), as reported in Table 13 for scenarios of the Netherlands (epidemic in 2003) and Italy (epidemic 1999). In particular, if the aim of the peri-vaccination zone is to contain the spread of the disease avoiding any jump outside the area with 95% confidence, a zone radius of at least 10 km would be needed (probability of jump spread equal to 0.04, meaning probability of containment equal to 0.96, so 96%). For emergency vaccination, the establishments within the 3-km protection zone would be vaccinated and therefore should have a reduced risk of HPAIV transmission compared with that observed in the absence of vaccination; therefore, emergency vaccination applied in the 3-km radius around the outbreak increases the confidence that the probability of transmission beyond 10 km is lower than 0.04. Given this consideration, a 10-km peri-vaccination zone radius can be considered as a worst-case scenario estimate.

TABLE 13Probability of transmission of highly pathogenic avian influenza virus beyond different distances in the scenario of the Netherlands(epidemic in 2003) and Italy (epidemic 1999) as estimated in EFSA AHAW Panel (2023).

	Distance (km)								
	3	5	10	15	20	25	50		
Netherlands, 2003 epidemic	0.52	0.22	0.04	0.01	0.007	0.004	< 0.001		
Italy, 1999 epidemic	0.33	0.15	0.04	0.02	0.01	0.006	0.001		

In relation to the **type of surveillance** to detect any new outbreaks in the peri-vaccination zone, the options depend on the vaccination status of the establishments within that area. If the establishments are vaccinated (e.g. when preventive vaccination was applied before the outbreak), the scenarios presented in Tables 10A–12B would also apply here.

If establishments in the peri-vaccination zone are not vaccinated, the recommendations provided in the EFSA Opinion on control measures for Category A diseases (EFSA AHAW Panel, 2021) are applicable, i.e. passive surveillance would be applied for gallinaceous poultry species (chickens and turkeys) and bucket sampling for Anseriformes (ducks and geese). The reason for the latter is due to the risk that some HPAIV strains could cause limited mortality (case fatality) in these species. Therefore, we have assessed scenarios where we assumed a low case fatality of 0.2 (20%) (Table 14) in unvaccinated duck flocks. Weekly sampling of all available dead birds (preferably dead within the last 48 h), as recommended by the EFSA opinion on Category A diseases, would be effective to early detect the infection.

TABLE 14 Efficacy of different sample size, sampling frequency and diagnostic test combination for early detection (defined as reduction of $R_s < 1$) of unvaccinated infected flocks, strategy E1. These simulations were done for a flock of **2500 ducks**^a and assuming a case fatality equal to 0.2, a qPCR sensitivity of 0.99 and ELISA sensitivity of 0.90.

Sample type (diagnostic test)	Sample size	Sampling interval (days)	Percentage of outbreak simulations with the probabilities of escaping detection below 1% ^b	Detection time as days post- introduction (median (2.5–97.5 Cl))	Prevalence (%) infectious birds (median (2.5–97.5 Cl))	Prevalence (%) recovered birds (median (2.5–97.5 Cl))	R _h /R _s (reproduction number) (median (2.5–97.5 Cl))
Passive reporting (reference)				6 (5–8)	59.81 (45.26–68.1)	7.66 (4.4–11.5)	1.8
Mortality threshold (0.13%)				4 (3–5)	5.41 (2.85–11.43)	0.44 (0.12–1.07)	0.2 (0.1–0.33)
Dead birds (qPCR)	≤15	7	96%	3 (2–5)	2.33 (1.04-4.5)	0.16 (0-0.48)	0.08 (0.06-0.12)
		14	95%	3 (3–5)	3.41 (1.68–6.9)	0.24 (0.04–0.67)	0.13 (0.1–0.18)
		21					
		30					

^aThis was the median size for flocks smaller than 4000 in France.

^bAn individual infected flock is considered detected when the probability of escaping detection < 0.01 (probability of detection > 0.99).

3.1.2.2 | Surveillance for demonstrating disease freedom following emergency vaccination (strategy E3, E4)

During the process of controlling an epidemic where emergency vaccination is applied and surveillance is implemented, following the strategies identified and described in Tables 10A–12B, there is a moment in time when no more outbreaks are detected. This will result in only negative results at both establishment (flock(s) within an establishment) and zone levels. In some cases (and for some establishments), multiple samplings would be done in time and all would be negative. Given these repeated negative results in time and the high probability of detection (> 99%) of infected flocks within an establishment, one could expect a high level of confidence in freedom. If the estimation of the probability of freedom is required, then Equation (10) (Section 2.1.5.2) could be used.

3.1.2.3 | Surveillance for early detection and demonstrating freedom in preventive vaccination populations (strategy P1, P3, P4)

In this section, results are presented based on simulated scenarios in the Netherlands, France and Italy, focusing specifically on areas previously identified as high-risk zones for the incursion of HPAIV (see Opinion Part 1). We assessed scenarios for demonstrating freedom from disease within these high-risk zones, where specific poultry species are vaccinated (chicken layers in the Netherlands, ducks in France and turkeys in Italy), leading to the co-existence of vaccinated and unvaccinated

51 of 70

poultry populations. This situation could occur if, for instance, only certain poultry species are vaccinated during a campaign of preventive vaccination as described in Opinion part 1 (EFSA AHAW Panel, 2023). In this high-risk zone, it is assumed that active surveillance is conducted in all vaccinated flocks by collecting every 30 days in a 48-h period all dead birds up to a number of 15 to be tested by qPCR, while passive surveillance is applied in unvaccinated flocks (Tables 15A, 16A and 17A). As mentioned in Table 4, this testing scheme corresponded to a median of 97% of outbreak simulations being detected (Sensitivity) in chicken layers, 96% in ducks and 96% (CL: 87–97) in turkeys. Note that passive surveillance is by default applied also on vaccinated flocks; therefore, if unusual increased mortality is observed between two consecutive sampling events, additional testing should be considered to exclude the presence of HPAIV.

In addition, we explored scenarios with variations in sampling intervals (30, 15 and 7 days) and proportions of vaccinated flocks (100%, 50% and 25%) (Tables 15B, 16B and 17B). For definitions of the measures used in the tables, see Glossary.

As indicated in the methods, the probability of freedom from the disease is here provided and, considering that this probability is highly dependent on the prior chosen for the prevalence of the disease in a certain area (Equation 11 and see Glossary), we also provided the early detection sensitivity (EDSe, i.e. the probability that at least one infected establishment is detected by the surveillance component in the selected time frame for a given design prevalence) and the surveillance system sensitivity (TotalSe), which combines the sensitivity of all surveillance components in the Tables below.

Results show that by testing all dead birds up to a number of 15 collected in a flock during 48 h by qPCR every 30 days in all vaccinated flocks and by applying passive surveillance in unvaccinated flocks, the estimated probability of disease freedom (Pfree) would exceed 99% in the three high-risk regions assessed, if all results show negative. The overall sensitivity of active and passive surveillance in all the poultry (Surveillance system sensitivity) was 95% in the high-risk regions of the Netherlands and 99% in the high-risk regions of France and Italy. The probability of detecting at least one infected vaccinated establishment (EDse) in the time frame of 30 days was 92% for vaccinated chickens in the Netherlands, 74% for vaccinated ducks in France and 93% for vaccinated turkeys in Italy. This indicates that the early detection sensitivity in vaccinated flocks is higher (chicken layers and turkeys) or similar (ducks) to the early detection sensitivity of passive surveillance in unvaccinated flocks of other poultry types in the regions (Tables 15A, 16A and 17A)

Results show that the early detection surveillance sensitivities (EDSe) vary with different sampling frequencies and proportions of vaccinated establishments sampled (Tables 15B, 16B and 17B). For example, reducing the proportion of tested turkey flocks from 100% to 50% reduces the EDSe from an average of 93% to 49% (Table 17B). Similarly, reducing it to 25% further reduces the early detection surveillance sensitivity to 25% (Table 17B). Conversely, the early detection surveillance sensitivities are increased by reducing the sampling intervals. For example, reducing the sampling intervals from 30 to 7 days in the turkey flocks that are tested increases the early detection sensitivity from an average of 93% to 100% (Table 17B).

However, reducing the proportion of the establishments tested or reducing the sampling intervals had a more limited effect on the probability of freedom (Pfree), given all tests were negative. This is likely due to the low hypothetical prior probability of infection (priorInf) chosen (5%) and the inclusion of passive surveillance in non-vaccinated poultry types. For example, sampling every 30 days and reducing the proportion of vaccinated turkey flocks tested to 25% resulted in a probability of freedom of 96% (compared with 99% if 100% of vaccinated turkey flocks are tested).

The monthly clinical examination of poultry, check of production records, check of health records of each epidemiological unit, as set out in Delegated Regulation (EU) No 2023/361 for surveillance in preventive vaccination, when applied in addition to the surveillance scheme here proposed, further contributes at increasing the estimated confidence for early detection and disease freedom.

Farm type	Number of establishments in the high-risk zone	Time frame (days) ^a	Sensitivity of surveillance component (CSe) ^b	Early detection sensitivity (EDSe) ^c	Surveillance system sensitivity (TotalSe) ^d	Probability of freedom (Pfree) ^e
Unvaccinated breeder flocks	1	16 (14–23)	0.02 (0.01–0.03)	0.02 (0.01–0.03)	0.95 (0.70–0.99)	0.997 (0.985–0.999)
Unvaccinated broiler flocks	33	16 (14–23)	0.20 (0.08–0.34)	0.19 (0.07–0.32)		
Unvaccinated duck flocks	2	6 (5–8)	0.16 (0.07–0.28)	0.15 (0.07–0.25)		
Vaccinated Layer flocks	242	31 (25–43)	0.93 (0.56–0.99)	0.92 (0.56–0.99)		
Unvaccinated turkey flocks	1	7 (6–9)	0.07 (0.03–0.13)	0.06 (0.03–0.12)		

TABLE 15A Early detection sensitivity and probability of freedom from disease in a high-risk zone for both introduction and transmission in the Netherlands. Here, the active surveillance is performed in vaccinated **chicken layer** flocks (testing all dead birds up to a number of 15 and up to 30 days) and passive surveillance in unvaccinated flocks.

^aTime period between virus introduction and detection by passive surveillance.

^cEDSe as in Equation (9). For passive surveillance of unvaccinated flocks, the cumulative EDSe is calculated (assuming no outbreaks was reported) up to the end of the sampling interval.

^dTotalSe as in Equation (10).

^ePfree as in Equation (11).

^bCse as in Equation (8)

TABLE 15B Early detection sensitivity and probability of freedom from disease (if all sampled establishments are negative) when considering different sampling frequency and proportion of vaccinated **chicken layer** establishments tested.

Sampling interval	Proportion of vaccinated chicken layer establishment sampled (%)	Number of establishments	Early detection sensitivity (EDSe) ^a	Probability of freedom (Pfree)
30	100	242	0.92 (0.56–0.99)	0.996 (0.977–0.999)
30	50	121	0.46 (0.28–0.50)	0.973 (0.963–0.974)
30	25	61	0.23 (0.14–0.25)	0.961 (0.957–0.962)
14	100	242	0.986 (0.80–0.99)	0.999 (0.989–1)
14	50	121	0.71 (0.48–0.75)	0.985 (0.973–0.987)
14	25	61	0.41 (0.26–0.44)	0.970 (0.962–0.971)
7	100	242	1	1
7	50	121	0.99 (0.99–1)	1
7	25	61	0.99 (0.99–0.99)	0.999 (0.999–1)

^aFor sampling intervals shorter than the time frame, the cumulative EDSe up to the end of the time frame (expected detection by passive surveillance) is calculated.

TABLE 16A Early detection sensitivity and probability of freedom from disease in a high-risk zone for both introduction and transmission in France. Here, active surveillance is performed in vaccinated **duck** flocks (testing all dead birds up to 15 and every 30 days) and passive surveillance in unvaccinated establishments.

Farm type	Number of establishments in the high-risk zone	Time frame (days) ^a	Sensitivity of surveillance component (CSe) ^b	Early detection sensitivity (EDSe) ^c	Surveillance system sensitivity (TotalSe) ^d	Probability of freedom (Pfree) ^e
Unvaccinated breeder flocks	625	16 (14–23)	0.99 (0.99–1)	0.77 (0.74–0.81)	0.99 (0.99–1)	0.999 (0.999–1)
Unvaccinated broiler flocks	3194	16 (14–23)	1	0.77 (0.74–0.81)		
Vaccinated duck flocks	1907	21 (17–30)	1	0.74 (0.60–0.90)		
Unvaccinated layer flocks	1000	16 (14–23)	1	0.77 (0.74–0.81)		
Unvaccinated turkey flocks	154	7 (6–9)	0.99 (0.99–1)	0.68 (0.67–0.69)		

^aTime period between virus introduction and detection by passive surveillance.

^bCse as in Equation (8).

^cEDSe as in Equation (9). For passive surveillance of unvaccinated flocks, the cumulative EDSe is calculated (assuming no outbreaks was reported) up to the end of the sampling interval.

^dTotalSe as in Equation (10).

^ePfree as in Equation (11).

TABLE 16B Early detection sensitivity and probability of freedom from disease (if all sampled establishments are negative) when considering different sampling frequency and proportion of vaccinated **duck** establishments tested.

Sampling interval	Proportion of vaccinated duck establishment sampled (%)	Number of establishments	Early detection sensitivity (EDSe) ^a	Probability of freedom (Pfree)
30	100	1907	0.74 (0.60–0.90)	0.999 (0.998–0.999)
30	50	954	0.37 (0.30–0.45)	0.968 (0.964–0.972)
30	25	478	0.18 (0.15–0.22)	0.959 (0.957–0.961)
14	100	1907	1	1
14	50	954	0.75	0.987 (0.987–0.999)
14	25	478	0.44	0.971 (0.970–0.999)
7	100	1907	1	1
7	50	954	0.95	0.997
7	25	478	0.71	0.985

^aFor sampling intervals shorter than the time frame, the cumulative EDSe up to the end of the time frame (expected detection by passive surveillance) is calculated.

TABLE 17AEarly detection surveillance sensitivity, and probability of freedom from disease in a high-risk zone for both introduction and
transmission in Italy. Here, active surveillance is performed in vaccinated **turkey** flocks (testing all dead birds up to a maximum of 15 every 30 days)
and passive surveillance in unvaccinated establishments.

Farm type	Number of establishments in the high-risk zone	Time frame ^a (days)	Sensitivity of surveillance component (CSe) ^b	Early detection sensitivity (EDSe) ^c	Surveillance system sensitivity (TotalSe) ^d	Posterior probability free (Pfree) ^e
Unvaccinated breeder flocks	120	16 (14–23)	0.89 (0.68–0.99)	0.71 (0.58–0.71)	0.99 (0.99–1.00)	0.999 (0.999–1.00)
Unvaccinated broiler flocks	501	16 (14–23)	0.93 (0.36–0.99)	0.72 (0.34–0.80)		
Unvaccinated duck flocks	10	6 (5–8)	0.35 (0.13–0.79)	0.30 (0.12–0.58)		
Unvaccinated layer flocks	300	16 (14–23)	0.99 (0.98–1)	0.77 (0.74–0.81)		
Vaccinated turkey flocks	385	27 (21–38)	0.99 (0.97–1)	0.93 (0.83–1)		

^aTime period between virus introduction and detection by passive surveillance.

^bCse as in Equation (8).

^cEDSe as in Equation (9). For passive surveillance of unvaccinated flocks, the cumulative EDSe is calculated (assuming no outbreaks was reported) up to the end of the sampling interval.

^dTotalSe as in Equation (10).

^ePfree as in Equation (11).

TABLE 17B Early detection sensitivity and probability of freedom from disease (if all sampled establishments are negative) when considering different sampling frequency and proportion of vaccinated **turkey** establishments tested.

Sampling frequency	Proportion of vaccinated turkey establishment sampled (%)	Number of establishment	Early detection sensitivity (EDSe) ^a	Probability free (Pfree)
30	100	385	0.93 (0.83–1)	0.988 (0.986-0.990)
30	50	193	0.49 (0.46–0.50)	0.974 (0.973–0.975)
30	25	97	0.25 (0.24–0.25)	0.962 (0.961–0.962)
14	100	385	0.99 (0.99–1)	0.999 (0.999–1)
14	50	193	0.75 (0.74–0.75)	0.987 (0.986–0.987)
14	25	97	0.44 (0.43–0.44)	0.971 (0.970–0.971)
7	100	385	1	1
7	50	193	1	1
7	25	97	0.99 (0.99–1)	0.999 (0.999–1)

^aFor sampling intervals shorter than the time frame, the cumulative EDSe up to the end of the time frame (expected detection by passive surveillance) is calculated.

3.1.2.4 | Surveillance to assess vaccination effectiveness (E2)

Estimation of vaccine effectiveness – data needed and possible analysis

Surveillance can also be used to assess vaccination effectiveness (VE), e.g. the ability of vaccination to prevent outbreaks upon exposure. The regional implementation of vaccination allows the observation of many vaccinated establishments to assess vaccination effectiveness following natural exposure and it may also be possible to assess vaccination characteristics like the onset of protection upon vaccination (e.g. following emergency vaccination) or the duration of protection. In addition, the effectiveness of different vaccines or vaccination schemes can be compared. However, some challenges can be foreseen: (i) Natural virus circulation has to occur in the population, (ii) an effective surveillance system has to be in place to early detect the virus, (iii) the probability of exposure to the pathogen is not necessarily equal across the territory or across an epidemic (i.e. among study groups). To estimate the effectiveness of vaccination, the probability of exposure of vaccinated and non-vaccinated flocks to the field virus should be similar on average (EFSA, 2018; Klement et al., 2020). Given that vaccination is expected to be applied in those regions most at risk, it would be difficult to have all those conditions in place.

During an epidemic, where emergency vaccination is applied, it will be of interest to assess VE at different time points following vaccination. During emergencies, it is of interest to assess the onset of protection and progression of VE in time post-vaccination, e.g. assessment of VE at 7, 14, \geq 21 days following vaccination would be desirable. This information would help, for example, improve emergency preparedness by using these VE parameters to inform simulation models such as those used for the assessment of ToRs 1 and 2 in the Opinion Part 1 (EFSA AHAW Panel, 2023).

To estimate the vaccination effectiveness based on relative risk in the vaccinated population compared with the unvaccinated population, the data needed are the vaccination status of an establishment (Y/N), the date of vaccination, the HPAI infection status (Y/N) and the respective date of detection. Based on this date of detection, inferences on the date of infection could be made (e.g. suspected date of virus introduction).

It is recommended to collect and group the data at the smallest administrative area (e.g. municipality). This is to ensure (as much as possible) that unvaccinated/vaccinated establishments within the geographical region have a similar probability of exposure. In addition, during emergencies, these establishments are likely to be vaccinated practically at the same time. During epidemics, it could be challenging to separate vaccinated from unvaccinated groups geographically. Instead, separation can be made temporally, by comparing the probabilities (or Hazard) of infection before vaccination and at different time points following vaccination (e.g. 7, 14 and > 21 days). Depending on the analytic method, either regression or survival models, RR (Poisson regression), OR (logistic regression) or HR (Cox proportional hazard) could be calculated and VE values could be derived.

In summary, assessing VE following emergency vaccination would help to improve emergency preparedness, to make inferences of the vaccine used and to assess the efficacy of the implemented control strategies and related allocation of resources, and together define more effective interventions for future epidemics.

During preventive vaccination, all establishments of a specific poultry type are likely to be vaccinated, so no comparison with unvaccinated ones is possible. Nevertheless, the occurrence of outbreaks in vaccinated establishments points to vaccination failure. Given the assumption that vaccination is effective in 70% of the establishments (EFSA AHAW Panel, 2023), some outbreaks may be expected if HPAIV exposure from wild birds occurs. Nevertheless, this should not result in extensive between farm transmission (EFSA AHAW Panel, 2023). To detect this as soon as possible, it is important to sequence viruses from the outbreaks to estimate the likelihood of between farms spread in contrast to separate introductions from the wild compartment.

3.1.2.5 Surveillance to assess the level of immune response induced by vaccination (P2)

In the absence of virus circulation, vaccination effectiveness cannot be assessed, but the level of immunity at the establishment level induced by vaccination can be measured using seropositivity, or any other correlate of protection, following the administration of the vaccine. Therefore, in addition to monitoring the level of immunity, there needs to be confirmation that there is no virus circulation in the vaccinated flock, following the surveillance approach discussed in Section 3.1.2.3.

The type of vaccine applied can affect the character (humoral or cell-mediated) and magnitude of immune response, this may in turn determine the most appropriate diagnostic test to monitor the immunity response at the flock level, as thoroughly discussed in Section 3.1.1. To do this, the animals should be sampled at a precise time within the time window for detection of birds with a specific correlate of protection (e.g. 28–35 days post vaccination for HVT-H5 vectored vaccine, not earlier than 21 days post-vaccination or latest booster for inactivated vaccines). However, identifying a correlate of protection, e.g. a given HI titre, which can be used to monitor the level of immunity reached within a flock (referred to as 'herd immunity') remains elusive (EFSA AHAW Panel, 2023). Hence, we suggest a pragmatic approach to assess the percentage (prevalence) of a vaccinated flock showing an 'effective' immune response (as defined by the diagnostic test to be used) as an indication of 'adequate vaccination' and assume that this could also provide an indication of the level of herd immunity reached in the vaccinated flock. First, the desired level of herd immunity level is equal to $1-1/R_0$. Median (range) of experimental and field estimates of R_0 for HPAI H5 viruses (CVI and APHA, 2017; EFSA AHAW Panel, 2023) are 3 (2–6) for chickens and 20 (16–88) for ducks. For turkeys, only one estimation was available, this estimate was 13 (95% Confidence Intervals: 9–17) (Ssematimba et al., 2019). Given these estimates, the desired level of immunity would be \geq 95 (94–99)% for domestic ducks ($R_0 = 24$), \geq 92 (89–94)% for turkeys ($R_0 = 12.8$) and \geq 73 (50–83)% for chicken layers ($R_0 = 3.6$).

A sampling approach could be the one previously adopted in the emergency vaccination plans, set out in Commission Decision 2000/721/EC,¹⁰ applied for the control of LPAI in Italy in 2000 and currently being used to monitor vaccination in France. The approach is based on taking and assessing at least 20 random samples per establishment. In Table 18 sample sizes, prevalence estimates based on the number of positive (for immune response) birds (the 'sample prevalence') and the one-sided 95% lower confidence limit (LCL) of the prevalence are provided. The latter gives an indication of the confidence that the 'true' prevalence of immune response in the establishment is at the required level. For example, if we were monitoring vaccinated chicken layers, the level of immunity we could target is \geq 73%. If 20 samples were taken and 15 are positive, the sample prevalence is 75% but we cannot be 95% confident that the 'true' prevalence in the establishment is \geq 73%, since the 95% LCL of the prevalence is 58%. If, however, 18 samples were positive, then one can be 95% confident that the prevalence in the establishment is \geq 73%, since the 95% LCL of the true prevalence is 74% (Table 18). Similar interpretations can be made for different sample sizes and results combinations, for chickens, ducks or turkeys.

¹⁰Commission Decision 2000/721/EC of 7 November 2000 on introducing vaccination to supplement the measures to control avian influenza in Italy and on specific movement control measures. OJ L 291, 18.11.2000, pp.33–36.

Poultry species	Required prevalence of immune responders	Sample size	Number positives	Sample prevalence	95% lower confidence limit ^a
Chickens	≥73 (50–83)%	20	15	75%	58%
Turkeys	≥92 (89–94)%		16	80%	62%
Ducks	≥95 (94–99)%		17	85%	69%
			18	90%	74%
			19	95%	80%
			20	100%	88%
		30	25	83%	70%
			26	87%	73%
			27	90%	77%
			28	93%	82%
			29	96%	86%
			30	100%	92%
		40	35	87%	76%
			36	90%	79%
			37	92%	83%
			38	95%	86%
			39	97%	89%
			40	100%	94%
		50	45	90%	81%
			46	92%	83%
			47	94%	86%
			48	96%	89%
			49	98%	92%
			50	100%	95%

TABLE 18 Sample sizes, to determine the prevalence of birds showing a positive immune response induced by vaccination and assess whether this prevalence is at the required level.

^aThese are one-sided Wilson exact 95% confidence limits.

3.1.2.6 Surveillance after cessation of vaccination (duration and type)

After cessation of vaccination, i.e. after the last vaccine has been administered and there are no further plans to vaccinate the birds, it is expected that vaccinated flocks will be gradually replaced by unvaccinated flocks at the end of each production cycle. This gradual replacement will end to a time point where all the flocks in the area will be unvaccinated and, therefore, no longer subject to HPAI surveillance to be implemented after vaccination. As assessed in the Opinion Part 1 (EFSA AHAW Panel, 2023), currently, there is no knowledge on the length of protective immunity in vaccinated birds, particularly for long-living birds (e.g. breeders) and on threshold values for correlate of protection. Therefore, an assessment of the duration of the surveillance after cessation of vaccination based on the demonstration that the vaccinated birds are still protected at a certain time point cannot be conclusive. Vaccinated flocks should be under surveillance depending on the context of vaccination (emergency/preventive) up to their replacement, with dead birds being preferred for testing. Meanwhile, passive surveillance should be applied on unvaccinated flocks present in the vaccination area.

3.1.2.7 | Surveillance in vaccinated captive birds

According to Regulation (EU) No 2016/429¹¹ on transmissible animal diseases ('Animal Health Law' – AHL), captive birds¹² are defined as any birds other than poultry¹³ that are kept in captivity for any reason other than poultry, including those that are kept for shows, races, exhibitions, competitions, breeding or selling.

¹¹Regulation (EU) No 2016/429 of the European Parliament and of the Council of 9 March 2016 on transmissible animal diseases and amending and repealing certain acts in the area of animal health ('Animal Health Law'). OJ L 84, 31.3.2016, p. 1.

¹²According to Regulation (EU) No 2016/429, **captive birds** means any birds other than poultry that are kept in captivity for any reason other than those referred to in point (9), including those that are kept for shows, races, exhibitions, competitions, breeding or selling.

¹³According to Regulation (EU) No 2016/429, poultry means birds that are reared or kept in captivity for: (a) the production of: (i) meat; (ii) eggs for consumption; (iii) other products; (b) restocking supplies of game birds; (c) the purpose of breeding of birds used for the types of production referred to in points (a) and (b).

Vaccinated birds in EU-approved confined establishments

Vaccinated birds in EU-approved confined¹⁴ establishments such as zoos, wildlife park or rehabilitation centres, are exempted from the surveillance as set out by Delegated Regulation (EU) No 2023/361 (see Part 5, point 2.3) (e.g. serological or virological testing). This is likely due to the negligible risk of HPAIV transmission from zoos to poultry holdings because of their confinement. Also, active surveillance is impractical, because periodically catching, swabbing and testing live wild birds of high value are difficult. Thus, passive surveillance is deemed more feasible and practical. Regular clinical examinations (e.g. weekly) by zoo personnel and zoo veterinarians should be conducted to early detect and intercept the presence of HPAIV in vaccinated zoo birds. In this setting, any clinical signs can be promptly detected, coupled with prompt analyses of any sick and dead birds, including wild birds found dead on the zoo premises. Moreover, environmental samples collected from each cage regularly (e.g. monthly) from e.g. cage structures, drinkers, feeders, faeces, feathers might improve the sensitivity of the surveillance system. If vaccinated birds are to be moved to another confined establishments, they should be subjected to clinical examination and molecular virological testing with negative results within 72 h before movement. Of note, there is limited knowledge in relation to the dynamic of HPAI infection of vaccinated zoo bird species.

Establishments with vaccinated ornamental birds

For establishments housing vaccinated ornamental birds, the surveillance approach may be tailored to their size and trade activities. In limited size establishments, passive surveillance similar to that proposed for confined establishments could be appropriate. However, in establishments with more extensive trade activities, surveillance strategies as proposed for poultry in Section 3.1.2.1 might be more suitable. Pre-movement clinical examination and molecular virological testing with negative results within 72 h before movement are recommended, considering the potential risks associated with movements of such birds to other establishments or to events (e.g. for trade, markets, exhibitions, shows, etc.) with a gathering of HPAI-sensitive animals. Of note, there is limited knowledge in relation to the dynamic of HPAI infection of vaccinated ornamental birds.

3.2 | ToR 4: Restrictions and risk mitigation measures

3.2.1 | Emergency vaccination

According to Delegated Regulation (EU) No 2020/687, during emergency vaccination authorisation can be granted for the movement of animals and eggs if the risk is considered negligible by the competent authority based on a risk assessment including results of clinical examination, laboratory examination (if necessary) and an official veterinary visit.

It should be considered that, according to Delegated Regulation (EU) No 2023/361, the establishments that have been vaccinated under emergency are already under surveillance, so laboratory examination is ongoing. According to the EFSA assessment, the surveillance to be applied in these establishments is reported in Sections 3.1.2.1 and 3.1.2.2. to early detect HPAI and to demonstrate freedom from HPAI (Tables 10A–12B). Therefore, including the effectiveness of the combined emergency vaccination and surveillance established in this Opinion, the risk of movement of animals and eggs is considered negligible in the following situations described in Delegated Regulation (see Figure 6). Note that, laboratory examinations, differently from what is stated in Delegated Regulation (EU) No 2020/687 where it should be done only if necessary, are here considered as part of the risk mitigation measures to be implemented.

For movement to the slaughterhouse, vaccinated **poultry** is clinically inspected and dead birds for molecular testing should be collected not earlier than 72 h before the movement. Negative test results should be provided before the movement and no clinical signs should be observed. Sampling should focus on the birds that have died on the day of clinical inspection or/ and the day before. This combination of inspection and virological checks results in a negligible risk and, consequently, from the perspective of HPAIV risk, there is no need to slaughter vaccinated animals separately from non-vaccinated animals.

For the movement of **ready to lay** poultry, vaccinated birds are clinically inspected and dead birds for testing should be collected not earlier than 72 h before the movement. Negative test results should be provided before the movement and no clinical signs should be observed. If the recipient establishment is located in the vaccinated zone, the appropriate testing scheme provided in Tables 10A–12B should be followed. If the recipient establishment is located outside the vaccinated zone, the establishment should be under official supervision for at least 21 days, during which the appropriate testing protocol provided in Tables 10A–12B should be followed. In addition, no other animals of listed species should be present in such an establishment.

For the movement of **hatching eggs**, vaccinated poultry in the breeding flock should be clinically inspected and dead birds for testing should be collected no earlier than 72 h before the movement. Negative test results should be provided before the movement and no clinical signs should be observed. Eggs should not come in contact with eggs or day-old chickens from outside the restriction zone.

¹⁴Confined establishments are defined according to Regulation (EU) No 2016/429 Art. 4 as any permanent, geographically limited establishment, created on a voluntary basis and approved for the purpose of movements, where the animals are: (a) kept or bred for the purposes of exhibitions, education, the conservation of species or research; (b) confined and separated from the surrounding environment; and (c) subject to animal health surveillance and biosecurity measures.

For the movement of **day-old chickens**, it should be confirmed that the breeding flock of origin has been free from HPAIV following the appropriate testing protocol provided in Tables 10A–12B. The establishments of destination should be under official supervision for 21 days.

Placing the establishments of destination of day-old-chickens, ready to lay poultry and hatcheries under official surveillance should facilitate proper implementation of biosecurity measures and assure early detection of the disease, thus reducing the potential risk of spread of the infection.

If the process of movement (e.g. loading) is not completed within 72 h, then testing should be repeated every 48 h until the process is finished.

Given the already ongoing surveillance, in all the situation described above, testing could coincide with the sampling session of the surveillance in place after emergency vaccination.

For movement of chicken layers from establishments < 3000 birds, given the reduced confidence of early detection in this flock size of the surveillance applied (Section 3.1.2.1), molecular testing of live birds (as recommended in Delegated Regulation (EU) No 2023/361) should be considered before movement in addition to clinical inspection and molecular virological testing of dead birds (all up to a number of 10 dead birds weekly as indicated in Section 3.1.2.1).

For movement of turkeys from establishments < 6000 birds, in addition to the surveillance strategy based on mortality threshold (see Section 3.1.2.1), it is recommended to combine clinical inspection and molecular testing of live (as recommended in Delegated Regulation (EU) No 2023/361) and dead birds before movements. It should be noted that those flock sizes are likely to contribute less to HPAIV transmission than larger flocks at a regional level.

3.2.2 | Preventive vaccination

According to Delegated Regulation (EU) No 2023/361, safe movements of vaccinated birds and their products are guaranteed if the vaccinated poultry or the day-old chickens and hatching eggs deriving from such poultry were negative to the active and enhanced passive surveillance implemented after preventive vaccination against HPAI (Table 1) and also other mitigation measures are implemented (see Figure 7). In the current Opinion, the surveillance to be applied after preventive vaccination is assessed in Section 3.1.2.3 and prescribes the monthly sampling of all dead birds up to a number of 15 collected in a flock 48 h before the testing to assess a zone as free from HPAIV.

Therefore, for preventive vaccination, where no outbreaks are present, the surveillance as described in Section 3.1.2.3 that is already in place provides a very high certainty that the area is free from infection (see Tables 15A–17B). Accordingly, if no outbreaks are detected and all establishments follow the surveillance protocol, the risk of movement is considered negligible in the following situations.

For movement to the slaughterhouse, vaccinated **poultry** should be clinically inspected not earlier than 72 h before the movement. Negative surveillance test results should be provided, and, before the movement, no clinical signs should be observed. In addition, at the slaughterhouse ante mortem inspection should be taking place. For Anseriformes, the sampling scheme to be applied under surveillance is the one provided in Tables 16A and 16B.

For the movement of **ready to lay** poultry, vaccinated poultry should be clinically inspected not earlier than 72 h before the movement. Negative surveillance test results should be provided and, before the movement, no clinical signs should be observed. If the recipient establishment is located in the vaccinated zone, the appropriate testing scheme provided in Tables 15A–17B should be followed. If the recipient establishment is located outside the vaccinated zone, the farm should be under official supervision for at least 21 days, during which the appropriate testing protocol provided in Tables 15A–17B should be followed. In addition, no other animals of listed species should be present in such an establishment.

For the movement of **hatching eggs** to a hatchery, the breeding establishment should follow the appropriate surveillance protocol given in Tables 15A–17B.

For the movement of **day-old chickens**, it should be clear that the breeding flock of origin has been free from HPAIV following the appropriate testing protocol provided in Tables 15A–17B. The establishment of the destination is under official supervision for 21 days.

If the establishment from where the vaccinated birds are to be moved is not under surveillance, for example when fewer than 100% of vaccinated establishments are subjected to testing (such as 50% or 25%), then vaccinated poultry or day-old chickens and hatching eggs deriving from such poultry are clinically inspected and dead birds for testing, all up to a number of 15, should be collected not earlier than 72 h before the movement. Negative test results should be provided before the movement and no clinical signs should be observed.

All other measures described for preventive vaccination, e.g. disinfection, traceability, are considered useful and suitable to prevent HPAIV spread via the movement of vaccinated birds or their products.

3.3 Sources of uncertainty

The source of uncertainty found in the current assessment are here listed. In relation to the assessment on diagnostic tests used in the context of HPAIV surveillance, there is a lack of information on several aspects including: the impact of vaccination on the performance of molecular tests under field conditions given the expected decrease in viral loads/shedding; the suitability of different environmental samples for surveillance in vaccinated flocks; the usefulness of amplification-based techniques other than PCR-based methods (e.g. LAMP, SDA, CRISPR) for detection of HPAIV; the performance of rapid

antigen detection systems on dead birds for monitoring of HPAIV circulation in vaccinated flocks; on the duration of the immunity elicited by vaccination; and on the correlate of protection which are described in Section 3.1.1.

Regarding the sources of uncertainty for the assessment of the effectiveness of the surveillance strategies at the establishment level, these are mainly related to the within-flock and between-flock transmission models chosen, which assumed values for transmission and clinical parameters such as the transmission rates, duration of latent and infectious periods, case fatality rate and baseline mortality and a given range of values for the between-flock reproduction number R_b based on previously published estimates when available or based on expert opinion (mainly for vaccinated turkeys) which may not be applicable to certain chicken layer, duck and turkey production systems in Europe. Also, there is uncertainty on species different from chicken layers, ducks and turkeys, given that data available for this assessment only relate to those species.

Similarly, for the assessment of surveillance strategies at area/zone level, there is uncertainty related to the parameters used as inputs for the scenario tree modelling, such as the relative risks of virus incursion in a zone/poultry species, which could differ among Member States, the proportion of establishments effectively protected following vaccination and the probability of infection in vaccinated flocks, which were informed by previously published studies which may not be representative of all the scenarios/populations covered by this assessment.

4 CONCLUSIONS AND RECOMMENDATIONS

4.1 **ToR 3: Diagnostic methods**

- A broad spectrum of diagnostics methods is available for selection according to vaccine type, vaccination strategy and surveillance objective.
 - Recommendation: The vaccination plan should already pre-select the most appropriate diagnostic assays based on vaccine type, vaccination strategy and surveillance objective. Recommended diagnostic tests are provided in Tables 2 and 5.
- There is a lack of field experience and validation data for diagnostic methods to be used for active surveillance in vaccinated flocks.
 - Recommendation: The vaccination plan should only set out the use of diagnostic methods compliant with Points a or b of Point 1 of Article 6 of Commission Delegated Regulation (EU) No 2020/689,¹⁵ while Members States are encouraged to conduct additional studies to collect field experience and validation data on alternative diagnostic methods in vaccinated establishments. The Manual of Diagnostic Tests and Vaccines for Terrestrial Animals (WOAH, online-b) (Section 1.1.6. – Validation of diagnostic assays for infectious diseases of terrestrial animals and Section 2.2. – Validation of diagnostic tests), as well as the requirements of the EN ISO/IEC 17025¹⁶ provide guidance for validating diagnostic methods to ensure they are compliant with the purpose they are intended for.
 - Recommendation: Since vaccinated birds, if exposed and infected, are expected to excrete lower levels of the virus than unvaccinated birds, the use of diagnostic methods with high sensitivity is recommended.
- Serological methods used to assess vaccination coverage, which are based on the detection of antibodies, can perform poorly in flocks that have received only replicating-vectored vaccines as these predominantly induce T-cell over humoral immune responses.
 - Recommendation: For each vaccine strategy-specific thresholds to estimate vaccination coverage as well as correlates of protection need to be identified based on experimental studies and field evidence.
- Serological methods usually have a lower specificity than molecular diagnostic methods. This compromises their use as a sole surveillance assay.
 - Recommendation: Serological results when aiming at demonstrating disease freedom must be confirmed with molecular virological investigations (i.e. adequate samples should be collected to allow for virological testing in DIVA-vaccinated flocks with seropositive monitoring results).
- Serological data obtained from laboratory trials performed in the EU demonstrated that accurate selection of antigens for HI tests is important for sensitive detection of vaccine-induced humoral immune responses (i.e. selection of antigens

¹⁶EN ISO/IEC 17025:2017. General requirements for the competence of testing and calibration laboratories. International Organization for Standardization.

18314732, 2024, 4, Downloaded from https://efsa.onlinelibrary.wiley.com/doi/10.2903/j.efsa.2024.8755 by CochraneItalia, Wiley Online Library on [15/11/2024]. See the Terms and Conditions (https://onlinelibrary.wiley.com/terms-and-conditions) on Wiley Online Library for rules of use; OA articles are governed by the applicable Creative Commons License

¹⁵Commission Delegated Regulation (EU) 2020/689 of 17 December 2019 supplementing Regulation (EU) 2016/429 of the European Parliament and of the Council as regards rules for surveillance, eradication programmes, and disease-free status for certain listed and emerging diseases. OJ L 174, 3.6.2020, pp. 211–340.

antigenically close to the one contained or expressed by the vaccine used) and for the differentiation of antibodies stimulated by HPAI infection in vaccinated birds from antibodies elicited by vaccination (i.e. antigens antigenically close to circulating field strains).

- **Recommendation**: Besides data collected in laboratory settings, field data are needed to understand antibody dynamics in the field.
- **Recommendation**: Continuous monitoring and improvement of test antigens tailored to the specific surveillance activities aims is needed to ensure the highest performances of HI assays.
- NP-ELISA assays can be used to detect past infection in vaccinated birds (DIVA approach) starting from 7 days postinfection with Se that range from 60 to 100% depending on the species and if compatible vaccines (i.e. vaccines that do not contain the AI-NP in the formulation) are used.
 - **Recommendation**: Commercially available indirect ELISA have demonstrated superior performance compared with available competitive ELISA assays in both chickens and ducks and thus should be preferred if employed in surveillance activities and if the assay is validated for the target species.
- Reservation to using DIVA vaccination is linked to the low specificity of serological tests which increases risks of falsepositive results.
 - **Recommendation**: Seropositive results in DIVA-vaccinated flocks require independent confirmation by molecular assays carried out on swab samples.
 - **Recommendation**: Continuous monitoring and improvement of serological tests tailored to the specific surveillance activities aims in vaccinated flocks, is needed to ensure the highest performances of these assays.
- Co-circulation of avian influenza viruses of low pathogenicity can interfere with clear-cut molecular and serological diagnostic surveillance measures in HPAI vaccinated as well as in non-vaccinated flocks.
 - **Recommendation**: Molecular methods that allow identifying all avian influenza viruses, including LPAIV, down to the sub- and pathotype levels must be in place in official laboratories.
- Analysis of samples collected from the environment is cost- and time-saving compared with sampling and testing individual birds for HPAIV in premises suspected of infection. However, it is not yet suitable for surveillance, particularly when targeting early detection, as there is currently a lack of validation data and protocols in comparison to diagnostic performances with individual bird-derived samples in vaccinated flocks.
 - **Recommendation**: Validation of environmental sampling methods and associated protocols for vaccinated flocks, in particular for use as part of an early detection strategy, are needed in the field; research studies aimed at filling these knowledge gaps are encouraged also in the context of zoos or confined establishments.
- Field experience in vaccinating countries has demonstrated that the use of unvaccinated fully susceptible sentinels can increase the risk of infection of the flock and that numerous practical problems reduce their advantages or hamper their implementation.
 - Recommendation: Use of sentinels in vaccination plans is not recommended.
- Currently available commercial point-of-care diagnostic methods for HPAIV detection cannot be employed for surveillance in vaccinated flocks due to lack of validation. In general, they have lower performance (in particular lower sensitivity) compared with RT-qPCR.
 - **Recommendation**: Further development and validation of point-of-care assays are needed in the field; research in that direction should be performed. Currently, the use of rapid antigen assays for active surveillance in vaccinated flocks is not recommended.
- Genetic and phenotypic characterisation tools are available for the accurate detection and characterisation of vaccine escape mutants.
 - **Recommendation**: Genetic characterisation of all viruses identified from HPAIV detected in vaccinated flocks is highly recommended and NRLs should be prepared for this activity before beginning the vaccination campaigns.
 - **Recommendation**: Viral isolation followed by phenotypic characterisation (e.g. evaluation of antigenic characteristics by HI and/or V-NT; adaptation to mammalian species; replicative behaviour; etc.) is recommended in a representative number of PCR-positive samples obtained from vaccinated flocks.

4.2 | ToR 3: Surveillance

General considerations

- The epidemiological unit for surveillance is the establishment, which may consist of one or several flocks (comprising all poultry of the same health status kept in the same enclosure).
- Sample sizes are provided at the flock level, meaning that if an establishment includes several flocks, sampling should be carried out within each individual flock. Detection of HPAIV in one flock within an establishment indicates infection for the entire establishment.
- Surveillance strategies for testing dead birds include random sampling of all dead birds up to a number of 5, 10 or 15. If fewer dead birds than this number are available, all available dead birds should be tested.
- We assumed the lowest case fatality rates of 20%, 5% and 20% in infected vaccinated chicken layers, ducks and turkeys, respectively, observed for vaccines considered as not effective in stopping sustained transmission (EFSA AHAW Panel, 2023). However, uncertainty remains in estimating these parameters accurately.
- The model relies on within-flock transmission parameters, yet uncertainty persists regarding these parameters and their accurate determination due to data limitations.
 - Recommendation for research: It is recommended to estimate case fatality in (a sample of) HPAIV outbreaks in vaccinated flocks to reduce the uncertainty of this parameter. Laboratory testing of dead and live birds individually can be used to obtain these estimates.
 - **Recommendation** for research: It is recommended to estimate within-flock transmission parameters during HPAIV outbreaks to reduce the uncertainty of the transmission parameters.

General conclusions

- Surveillance strategies based on sampling and testing dead birds have a higher probability of early detecting HPAIV than strategies based on testing live birds.
- Increasing the sample size above the numbers of 5, 10 or 15 dead birds provided does not increase the probability of HPAIV detection.
- The effectiveness of surveillance is increased by the repeated sampling in time.
- If HPAIV enters a flock between two sampling events, the proportion of infected birds will be highest close to the second sampling event. Consequently, testing birds found dead in the last 48 h yields the highest sensitivity.
- If, at a certain sampling event, no dead birds are present in the flock, it is probable that the infection has either not entered the flock or entered so recently that it cannot be detected yet. In the latter case, the surveillance strategy involving repeated sampling would result in the collection of dead birds in the next sampling event, leading to the detection of the infection entered in the previous sampling interval.

o Recommendation: Sampling and testing dead birds should consist of dead birds found in the last 48 h.

- No definitive conclusion can be drawn regarding the duration of the surveillance after the cessation of vaccination in a flock, due to the current lack of knowledge on the correlate of protection and consequently on the duration of protective immunity in vaccinated poultry, particularly in long-living birds.
 - **Recommendation**: Vaccinated flocks should be under surveillance up to their replacement, with the sampling and testing of dead birds being the recommended option. Meanwhile, passive surveillance should be applied for unvaccinated flocks within the vaccination area.

4.2.1 | Surveillance for early detection at establishments for emergency vaccination

Strategies for early detection were considered effective if they resulted in more than 95% of the simulated outbreaks reaching detection probabilities above 99% (interpreted as leading to early detection). 'Passive reporting' and 'mortality threshold' strategies were considered effective if they stopped sustained between-flock transmission ($R_s < 1$).

Chicken layers

• The surveillance strategy described in Delegated Regulation (EU) No 2023/361, which involves testing 60 **live birds** for virus presence (**virological** testing) every 2 weeks during emergency vaccination to detect at least one infected bird given 5% or higher disease prevalence, was assessed as not effective for early detection since there was < 95% certainty that it would lead to the early detection of HPAIV in vaccinated chicken layer flocks.

- The alternative surveillance strategy of testing 60 **live birds** for antibodies presence (**serological** testing) every 2 weeks during emergency vaccination was also assessed as not effective since there was <95% certainty that it would lead to HPAIV early detection in vaccinated chicken layers.
- The surveillance strategies based on 'passive reporting' or on 'mortality thresholds' were not effective for HPAIV early detection, resulting in sustained between-flock transmission (R_s > 1) and long mean detection times post-introduction (e.g. 31 days (95% CI: 25–43) and 28 days (95% CI: 22–39), respectively, for 20,000 size flock; 21 days (95% CI: 16–33) and 15 days (CI 95%: 10–26), respectively, for 2500 flock size).
- The alternative strategy of testing **dead birds**, when implemented for flocks ≥ **3000** birds, was effective for early detection with >95% certainty (Table 10A).
- For flocks < 3000 chicken layers, none of the surveillance strategies assessed was effective with > 95% certainty for early detection (Table 10B). The maximum certainty that could be reached for early detection, by sampling all up to a number of 10 or 15 dead birds weekly, was 90%.
 - Recommendation: For early detection of HPAIV outbreaks, in flocks ≥ 3000 birds, during emergency vaccination, molecular testing of dead chicken layers in a flock is recommended. A number of effective options have been identified according to flock size testing all up to a number of 5, 10 or 15 dead birds weekly, up to 10 or 15 dead birds every 2 weeks or up to 15 dead birds every 3 weeks and should be selected according to a country's specific circumstances and resources.
 - Recommendation: For flocks < 3000 birds, it is recommended to test all up to a number of 10 dead birds weekly, which combined with molecular testing of live birds every 2 weeks, as recommended in the Delegated Regulation (EU) No 2023/361, would increase the level of confidence of early detection.

Ducks

- The surveillance strategy described in the Delegated Regulation (EU) No 2023/361, which involves testing 60 **live birds** for virus presence (**virological** testing) every 2 weeks during emergency vaccination to detect at least one infected bird given a disease prevalence of 5% or higher, was assessed as effective for early detection since there was > 95% certainty (specifically 97%) that it would lead to the early detection of HPAIV in vaccinated ducks flocks. The average detection time post-introduction was 14 days (95% CI: 11–23) in flocks of 2500 ducks, and 17 days (95% CI: 13–27) in flocks housing 6000 ducks, with these detection timeframes resulting in limited between-flock transmission ($R_s < 1$).
- The alternative surveillance strategy of testing 60 **live birds** for antibody presence (**serological** (DIVA) testing) every 2 weeks during emergency vaccination was also assessed as effective since there was > 95% certainty (specifically 97%) that it would lead to HPAIV early detection in vaccinated ducks. However, diagnostic performance of the serological tests is inferior compared with molecular tests, thus reducing our certainty in this strategy.
- A surveillance strategy based on '**passive reporting**' only was not effective for HPAIV early detection, resulting in sustained between-flock transmission ($R_s > 1$) and a longer mean detection time post-introduction of 21 and 23 days (95% Cl: 17–30; 19–32), depending on flock size (2500 and 6000, respectively).
- A surveillance strategy based on '**mortality thresholds**' was effective for early detection, resulting in limited betweenflock transmission ($R_s < 1$); additionally, in flocks of 2500 ducks, the average detection time post-introduction was the shortest – 10 days (95% CI: 7–18).
- The alternative strategies of testing **dead birds** were proven to be effective with >95% certainty that it would lead to early detection of HPAIV in vaccinated ducks. The average detection time ranged between 14 and 17 days post-HPAIV introduction in the flock, depending on sampling frequency, sample size and flock size and this was accompanied by limited between-flock transmission ($R_s < 1$) (Tables 11A and 11B).
 - Recommendation: For early detection of HPAIV outbreaks during emergency vaccination, molecular testing of dead ducks in a flock is recommended. A number of effective options have been identified: (i) in flocks of 2500 ducks (representing flocks with < 6000 ducks), testing all dead birds up to a number of 5, 10 or 15 weekly, (ii) in flocks of ≥ 6000 ducks, testing all dead birds up to 5, 10 or 15 weekly, testing all dead birds up to 10 or 15 every 2 weeks, or testing all dead birds up to 15 every 3 weeks. These strategies should be selected according to country's specific circumstances and resources.
 - **Recommendation**: As an alternative to testing dead birds, testing of 60 live ducks every 2 weeks (by means of molecular or serological testing) in a flock can also be carried out, but it may present more challenges in terms of welfare consequences and logistics (e.g. catching, handling and restrain) as well as requiring more resources.
 - **Recommendation**: As an alternative to testing dead birds, a surveillance strategy based on mortality threshold can also be carried out, but its effectiveness depends on the availability and accuracy of information about mortality thresholds that could prompt farmers to report suspicions of HPAI.

Turkeys

- The surveillance strategy described in the Delegated Regulation (EU) No 2023/361, which involves testing 60 **live birds** for virus presence (**virological** testing) every 2 weeks during emergency vaccination to detect at least one infected bird given a disease prevalence of 5% or higher, was assessed as not effective for early detection since there was <95% certainty that it would lead to the early detection of HPAIV in vaccinated turkey flocks.
- The alternative surveillance strategy of testing 60 **live birds** for antibodies presence (**serological** testing) every 2 weeks during emergency vaccination was also assessed as not effective since there was < 95% certainty that it would lead to HPAIV early detection in vaccinated turkeys.
- The surveillance strategies based on '**passive reporting**' or on '**mortality thresholds**' were not effective for HPAIV early detection in flocks \geq 6000 turkeys, resulting in sustained between-flock transmission ($R_s > 1$) and long mean detection time post-introduction (e.g. at 27 days (95% CI: 31–38) and 24 days (95% CI: 19–34), respectively, for flock 18,000).
- The alternative strategies of testing **dead birds**, when implemented for flocks of ≥ 6000 turkeys, proved to be effective for early detection with > 95% certainty (Table 12A).
- For flocks < 6000 turkeys, the only effective surveillance strategy for early detection was based on the use of 'mortality thresholds' (0.21%), resulting in limited between-flock transmission (R_s < 1) (Table 12B).
 - Recommendation: For early detection of HPAIV outbreaks, for flocks ≥ 6000 turkeys, during emergency vaccination, molecular testing of dead turkeys in a flock is recommended. A number of effective options have been identified testing up to 5, 10, 15 or 20 dead birds weekly, up to 15 or 20 dead birds every 2 weeks or up to 20 dead birds every 3 weeks and should be selected according to country's specific circumstances and resources.
 - Recommendation: For flocks < 6000 turkeys, the surveillance strategy based on mortality threshold should be carried out which combined with molecular testing of live birds every 2 weeks, as recommended in the Delegated Regulation (EU) No 2023/361, would increase the level of confidence of early detection; the effectiveness of mortality threshold depends on the availability and accuracy of information about mortality thresholds that could prompt farmers to report suspicions of HPAI.

Surveillance in peri-vaccination zone

- Based on the results from EFSA AHAW Panel (2021), it was assessed that in a 'surveillance zone', where there is an unvaccinated poultry population, the probability of transmission of HPAIV beyond a 10-km radius (jump spread) was below 0.04. The same radius can be applied to the peri-vaccination zone as a conservative estimate which increases the confidence that the probability of transmission beyond 10 km is lower than 0.04.
 - **Recommendation**: A 10-km peri-vaccination zone during emergency protective vaccination is recommended, to increase the confidence that the probability of jump spread is lower than 0.04 (4%).
- In relation to the type of surveillance in the peri-vaccination zone, the options depend on the vaccination status of the establishments.
 - Recommendation: For preventively vaccinated farms, it is recommended to apply the same surveillance strategy as for early detection of emergency vaccination. For those farms that are not vaccinated, the recommendations provided in EFSA Opinion on control measures for Category A diseases (EFSA AHAW Panel, 2021) are considered applicable. Therefore, passive surveillance is to be applied for gallinaceous poultry (chickens and turkeys) and bucket sampling for Anseriformes (ducks and geese), carrying out a weekly sampling of all available dead birds, preferably within the last 48 h.

4.2.2 | Surveillance for early detection and demonstrating disease freedom following preventive vaccination

- Molecular virological testing of all dead birds in vaccinated flocks up to a number of 15 every 30 days, coupled with passive surveillance in unvaccinated flocks, would lead to at least 99% confidence in disease freedom in the zone, assuming ≤ 5% prior prevalence of establishments infected. This was confirmed by the model in the three high-risk zones assessed, the Netherlands with vaccinated chicken layers, France with vaccinated ducks and Italy with vaccinated turkeys.
 - Recommendation: Molecular virological testing of up to 15 dead birds every 30 days in vaccinated flocks, while maintaining passive surveillance in both vaccinated and unvaccinated flocks, is recommended to effectively demonstrate disease freedom with > 99% confidence within high-risk zones for HPAIV infection.

- With the above testing scheme, the probability of early detecting at least one infected vaccinated establishment (EDse) in the time frame of 30 days was 92% for vaccinated chicken layers in the Netherlands, 74% for vaccinated ducks in France and 93% for vaccinated turkeys in Italy; reducing the sampling intervals from 30 days to 14 or 7 days increased the EDse to > 98% for chicken layer and turkey flocks and to 100% in duck flocks.
 - **Recommendation**: If the aim is to increase the early detection surveillance sensitivities, then it is recommended to reduce the sampling intervals.
- Reducing the proportion of vaccinated establishments to be tested to 25% would have a limited impact on the probability of freedom, remaining > 96% for chicken layers, 95% for ducks and > 96% for turkeys.
- Reducing the proportion of vaccinated establishments tested markedly decreases the early detection sensitivity (EDse). For instance, reducing the proportion of establishments to be tested from 100% to 50% would result in a decrease on EDse from 92% to 46% in chicken layers, from 74% to 37% in ducks and from 93% to 49% in turkeys.
 - Recommendation: For chicken layers, a number of effective options have been identified to maintain a level of early detection sensitivity > 92%, and probability of freedom > 99%, by testing all dead birds up to a number of 15 in 100% of the vaccinated establishments monthly; in 25% of the vaccinated establishments weekly and should be selected according to country's specific circumstances and resources.
 - Recommendation: For ducks, a number of effective options have been identified to maintain a level of early detection sensitivity > 74% and probability of freedom > 98%, by testing all dead birds up to a number of 15 in 100% of the vaccinated establishments monthly; in 50% of the vaccinated establishments every 2 weeks and should be selected according to the country's specific circumstances and resources.
 - Recommendation: For turkeys, a number of effective options have been identified to maintain a level of early detection sensitivity > 75% and probability of freedom > 98%, by testing all dead birds up to a number of 15 in 100% of the vaccinated establishments monthly; in 50% of the vaccinated establishments every 2 weeks; in 25% of the vaccinated establishments weekly and should be selected according to country's specific circumstances and resources.
- In the high-risk zones, where unvaccinated and vaccinated establishments are present, passive surveillance in the unvaccinated establishments would contribute to increase the overall surveillance system sensitivity, and the confidence of disease freedom.
 - **Recommendation**: Maintaining passive surveillance efforts in unvaccinated establishments in vaccinated zones is recommended to enhance the overall sensitivity of the surveillance system.

4.2.3 | Surveillance for estimation of vaccination effectiveness

- The estimation of vaccination effectiveness (VE) under field conditions is a crucial element for (i) preventing and controlling epidemics, (ii) verifying the measures applied and (iii) justifying the investments. The effectiveness can differ among vaccination programmes.
- The assessment of vaccination effectiveness requires careful planning, a good data collection framework, follow up of all flocks in the vaccinated area as well as sufficient human resources to conduct the study, in order to ensure high quality of data and outcomes from the analysis.
 - **Recommendation**: Data from the vaccinated population in the field should be collected from each epidemiological unit considered, including at least outbreak (events, suspicion and confirmation date) and vaccination (events, date) data, and spatial coordinates of each epidemiological unit. Data should be assessed to estimate vaccination effective-ness. Assessing VE at 7 and 14 days post-vaccination would give an indication of the VE for emergency vaccination.
 - **Recommendation**: In case of HPAI outbreaks following vaccination, it is recommended to sequence the virus to estimate the likelihood of between farm spread in contrast to separate introduction from wild birds.

4.2.4 | Surveillance to assess level of immune response induced by vaccination

- Assessment of the level of immune response following vaccination of an establishment can be performed when virus circulation is absent or expected to be low in the zone where (preventive) vaccination was applied.
 - Recommendation: Assessment of level of immune response following vaccination of an establishment according to the vaccines used and the selected diagnostic test, should be done combined with an assessment of the absence of virus circulation in the flock.

- **Recommendation**: To best detect the proportion of birds with a measurable immune response sampling should be done when the highest immune response is expected, e.g. between 21 and 28 days post primary or booster vaccination. If only vector-based vaccines are used (e.g. HVT-H5), immune response should be assessed at least 4 weeks (e.g. 4 weeks in chickens or 7 weeks in turkeys) after vaccination.
- Based on the R0 for within flock transmission quantified from experimental and field data, the minimum proportion of
 protected birds needed to stop transmission was estimated to be higher than 90% for ducks and turkeys, and 70% for
 chicken layers. Although there are limited estimates of R0 values for ducks and turkeys, transmission in these species is
 expected to be faster and more extensive (with 95% certainty) than transmission between chickens, hence the high level
 (> 90%) of herd immunity required.
 - **Recommendation for research**: Further quantification of R0 values for the currently circulating HPAIV is recommended to have a more accurate estimate of the minimum proportion of birds that should be protected in a flock to stop transmission.
- There is no established correlate of protection, e.g. antibody titre against HPAIV, that could be considered as a proven indicator of protection for poultry species. In the meantime, assessment of the prevalence of birds showing an immune response following vaccination is assumed to provide an indication of sufficient protection and possibly herd immunity. For chickens, taking 20 samples per establishment would allow confirmation of this prevalence being higher than the 70% with 95% certainty. For turkeys and ducks, 40, 50 or higher number of samples might be needed to confirm that the prevalence of birds' immune response is higher than 90% with 95% certainty.
 - **Recommendation**: At least 20 blood samples per establishment, tested with the most suitable diagnostic test depending on the vaccine used, are appropriate to determine whether vaccines are correctly used in the field.
 - Recommendation: As there are no established antibody titres that can be used as a correlate of protection for poultry species, the (sero)positivity in an appropriate test could be used as a proxy measure of protection in vaccinated birds. To estimate the true level of protection (herd immunity), an animal level correlate of protection should be defined. If a certain antibody titre was to be shown as a good correlate of protection, then a serological survey (based on serological test) could be conducted.

4.2.5 | Surveillance in vaccinated captive birds

- In confined establishments such as zoos and in small ornamental bird establishments, the risk of HPAI transmission to poultry holdings is likely limited, warranting passive surveillance. However, ornamental bird establishments with more extensive trade activities may pose increased risks.
- There is limited knowledge about the dynamics of HPAI infection in vaccinated species reared in confined or captive bird species establishments (which are usually different from poultry species).
 - **Recommendations**: Maintaining passive surveillance in vaccinated birds from confined and small ornamental bird establishments is recommended.
 - **Recommendations**: For kept ornamental birds, certain movements (i.e. those with a high risk of HPAIV spread such as for trade, exhibitions, shows, etc.) may represent a risk, and testing should be carried out before the movement takes place, since the premovement clinical examination alone may not be sufficient.

4.3 | ToR 4: Restriction and risk mitigation measures

4.3.1 | Emergency vaccination

- For **emergency vaccination**, where HPAI outbreaks are present in the area, vaccinated establishments are already under surveillance, including laboratory examination. In Delegated Regulation (EU) No 2020/687, clinical inspection in the 72 h before movement and laboratory examination to demonstrate freedom from HPAI is considered part of the risk mitigation measures.
 - Recommendation: To enable safe movement of vaccinated poultry, day-old chickens and hatching eggs deriving from such poultry, birds should be clinically inspected and all up to a number of 15 dead birds should be tested no earlier than 72 h before movement to demonstrate that they are free from HPAIV. Negative test results and the absence of clinical signs should be provided before movement.
 - **Recommendation**: Given the ongoing surveillance applied after emergency vaccination for early HPAIV detection and to demonstrate that the establishment is free from HPAIV, testing could coincide with the sampling session of the surveillance in place.

4.3.2 | Preventive vaccination

- For **preventive vaccination**, in which no HPAI outbreaks are present, the surveillance described in Section 3.1.2.3 (testing all dead birds up to a number of 15 in vaccinated flocks every month, see Section 4.2.2) provides high certainty that the area is free from HPAIV, given negative results. Accordingly, if no outbreaks are detected and all establishments follow the surveillance protocol, the risk of HPAIV spread via birds' movement is considered negligible.
 - **Recommendation**: To enable the safe movement of vaccinated poultry, day-old chickens and hatching eggs deriving from such poultry, negative surveillance test results should be provided, and no clinical signs should be observed before the movement.
 - Recommendation: If the vaccinated establishment is not under surveillance, for example when fewer than 100% of vaccinated establishments are tested, vaccinated poultry, day-old chickens and hatching eggs deriving from such poultry should be clinically inspected and all up to a number of 15 dead birds should be tested no earlier than 72 h before movement. Negative test results and the absence of clinical signs should be provided before movement.

ABBREVIATIONS

Al	avian influenza
AIV	avian influenza virus
AR	adjusted relative risks
CVMP	Committee for Veterinary Medicinal Products
DIVA	Differentiating Infected from Vaccinated Animals
EMA	European Medicines Agency
FPV	Fowlpox virus
GMO	genetically modified organism
HDPA	high-density poultry areas
HI	haemagglutination inhibition
HP	highly pathogenic
HPAI	highly pathogenic avian influenza
HPAIV	highly pathogenic avian influenza virus
HVT	Turkey Herpesvirus
LP	low pathogenic
LPAI	low pathogenic avian influenza
LPAIV	low pathogenic avian influenza virus
NDV	Newcastle Disease Virus
NP-ELISA	nucleoprotein-based ELISA
R _h	between-flocks reproduction number
RR	relative risk
SAN	specific antibody negative
SLR	systematic literature review
SPF	specific pathogen free
TRL	technology readiness level
USDA	United States Department of Agriculture
VE	vaccine efficacy
WOAH	World Organisation for Animal Health

GLOSSARY

CSe _i	component sensitivity for the <i>i</i> -th surveillance system component. It is the average probability that a surveillance system component will return a positive surveillance outcome, given disease is present in the population at a level equal to, or greater than, the specified design prevalence
EDSe	early detection sensitivity of each SSC, which is the probability that the surveillance activity is able to cor- rectly detect among the poultry establishments the epidemiological unit(s) affected by a new incursion within a specific time frame
Pfree	probability that the country is free from disease, given that the surveillance system did not detect the disease. Confidence of population freedom (= negative predictive value)
PriorInf	prior probability (also known as prior prevalence) that the poultry population at establishment level would be infected
SSC	surveillance system components, a surveillance activity that, in itself, can contribute evidence to disease freedom; it has the capacity to detect disease if/when it occurs Five SSCs were identified: broilers, fattening ducks, meat turkeys, breeders and laying hens
TotalSe	surveillance system sensitivity, which combines the EDSe calculated for each SSC

ACKNOWLEDGEMENTS

EFSA and EURL wish to thank the following Member States and their representatives who provided data on HPAI vaccines, poultry population and HPAI outbreaks: Karen Bucher, Bianca Quercetti and Andrea Jimenez Pellicer for France; Andrea Maroni Ponti and Luigi Ruocco for Italy; Borbála Bende and Gábor Wyszoczky for Hungary; Annemarie Bouma and Wim Pelgrim for The Netherlands. As well as thanking the following experts from National Reference Laboratories (NRL) on avian influenza and pharmaceutical industries for providing data on serological studies: Belgium – Fiona Ingrao, Julie Valentin, and Bénédicte Lambrecht, Huvepharma and CEVA; France – Beatrice Grasland; the Netherlands – Kim Bouwman and Nancy Beerens, MSD Animal Health, Huvepharma, Boehringer Ingelheim Animal Health and Ceva; Italy – Francesco Bonfante, Lucrezia Vianello, Sami Ramzi and Calogero Terregino. EFSA and EURL wish to acknowledge Beatrice Grasland as hearing expert from ANSES, Eleonora Bastino as hearing expert from the European Medicines Agency (EMA), Alexandre Fediaevsky as hearing expert from the World Organisation for Animal Health (WOAH), Francesca Scolamacchia as Individual Scientific Advisor (contract EOI/EFSA/SCIENCE/2020/01), the WBVR for the work done on modelling the surveillance strategies under contract OC/EFSA/ALPHA/2020/01, and the project KB-37-003-047-WBVR, and Roxane Delacourt and Gina Cioacata from EFSA for the support provided to this scientific output.

CONFLICT OF INTEREST

If you wish to access the declaration of interests of any expert contributing to an EFSA scientific assessment, please contact interestmanagement@efsa.europa.eu.

REQUESTOR

European Commission

QUESTION NUMBER

EFSA-Q-2023-00056

COPYRIGHT FOR NON-EFSA CONTENT

EFSA may include images or other content for which it does not hold copyright. In such cases, EFSA indicates the copyright holder and users should seek permission to reproduce the content from the original source.

PANEL MEMBERS

Søren Saxmose Nielsen, Julio Alvarez, Dominique Joseph Bicout, Paolo Calistri, Elisabetta Canali, Julian Ashley Drewe, Bruno Garin-Bastuji, Jose Luis Gonzales Rojas, Christian Gortázar, Mette S. Herskin, Virginie Michel, Miguel Ángel Miranda Chueca, Barbara Padalino, Helen Clare Roberts, Hans Spoolder, Karl Stahl, Antonio Velarde, Christoph Winckler and Arvo Viltrop.

REFERENCES

- Bavinck, V., Bouma, A., van Boven, M., Bos, M. E., Stassen, E., & Stegeman, J. A. (2009). The role of backyard poultry flocks in the epidemic of highly pathogenic avian influenza virus (H7N7) in the Netherlands in 2003. *Preventive Veterinary Medicine*, 88, 247–254. https://doi.org/10.1016/j.prevetmed.2008.10.007
- Bouma, A., Muljono, A. T., Jatikusumah, A., Nell, A. J., Mudjiartiningsih, S., Dharmayanti, I., Siregar, E. S., Claassen, I., Koch, G., & Stegeman, J. A. (2008). Field trial for assessment of avian influenza vaccination effectiveness in Indonesia. *Revue Scientifique et Technique*, 27, 633–642. https://doi.org/10. 20506/rst.27.3.1823
- Cameron, A. R., Meyer, A., Faverjon, C., & Mackenzie, C. (2020). Quantification of the sensitivity of early detection surveillance. *Transboundary and Emerging Diseases*, 67(6), 2532–2543. https://doi.org/10.1111/tbed.13598
- Cortes, A. L., Montiel, E. R., Lemiere, S., & Gimeno, I. M. (2011). Comparison of blood and feather pulp samples for the diagnosis of Marek's disease and for monitoring Marek's disease vaccination by real time-PCR. *Avian Diseases*, *55*(2), 302–310. https://doi.org/10.1637/9578-101510-ResNote.1
- CVI and APHA (Central Veterinary Institute (NL) and Animal and Plant Health Agency (UK)). (2017). Data analysis and predictive modelling of HPAI H5 and H7 outbreaks in the EU 2005–2015. EFSA Supporting Publications, 14, 1285E. https://doi.org/10.2903/sp.efsa.2017.en-1285
- Doherr, M. G., & Audigé, L. (2001). Monitoring and surveillance for rare health-related events: A review from the veterinary perspective. *Philosophical Transactions of the Royal Society of London. Series B, Biological Sciences, 356*(1411), 1097–1106. https://doi.org/10.1098/rstb.2001.0898
- EFSA (European Food Safety Authority). (2018). Scientific report on lumpy skin disease II. Data collection and analysis. EFSA Journal, 16(2), 5176. https:// doi.org/10.2903/j.efsa.2018.5176
- EFSA AHAW Panel (EFSA Panel on Animal Health and Welfare), Nielsen, S. S., Alvarez, J., Bicout, D. J., Calistri, P., Depner, K., Drewe, J. A., Garin-Bastuji, B., Gonzales Rojas, J. L., Schmidt, C. G., Herskin, M., Michel, V., Miranda Chueca, M. Á., Pasquali, P., Roberts, H. C., Sihvonen, L. H., Spoolder, H., Stahl, K., Calvo, A. V., ... Aznar, I. (2021). Scientific opinion on the assessment of the control measures of the category a diseases of animal health law: Highly pathogenic avian influenza. *EFSA Journal*, *19*(1), 6372. https://doi.org/10.2903/j.efsa.2021.6372
- EFSA AHAW Panel (EFSA Panel on Animal Health and Animal Welfare) European Union Reference Laboratory for Avian Influenza, Nielsen, S. S., Alvarez, J., Bicout, D. J., Calistri, P., Canali, E., Drewe, J. A., Garin-Bastuji, B., Gonzales Rojas, J. L., Gortázar, C., Herskin, M., Michel, V., Miranda Chueca, M. Á., Padalino, B., Roberts, H. C., Spoolder, H., Stahl, K., Velarde, A., Winckler, C., ... Viltrop, A. (2023). Vaccination of poultry against highly pathogenic avian influenza Part 1. Available vaccines and vaccination strategies. *EFSA Journal*, 21(10), e08271. https://doi.org/10.2903/j.efsa.2023.8271
- EFSA Scientific Committee, Benford, D., Halldorsson, T., Jeger, M. J., Knutsen, H. K., More, S., Naegeli, H., Noteborn, H., Ockleford, C., Ricci, A., Rychen, G., Schlatter, J. R., Silano, V., Solecki, R., Turck, D., Younes, M., Craig, P., Hart, A., Von Goetz, N., ... Hardy, A. (2018). Guidance on uncertainty analysis in scientific assessments. *EFSA Journal*, 16(1), 5123. https://doi.org/10.2903/j.efsa.2018.5123
- Elbers, A. R. W., & Gonzales, J. L. (2021). Mortality levels and production indicators for suspicion of highly pathogenic avian influenza virus infection in commercially farmed ducks. *Pathogens*, 10(11), 13 pp. https://doi.org/10.3390/pathogens10111498

EURL. (online). Diagnostic protocols. https://www.izsvenezie.com/reference-laboratories/avian-influenza-newcastle-disease/diagnostic-protocols/

- Fereidouni, S. R., Harder, T. C., Gaidet, N., Ziller, M., Hoffmann, B., Hammoumi, S., Globig, A., & Starick, E. (2012). Saving resources: Avian influenza surveillance using pooled swab samples and reduced reaction volumes in real-time RT-PCR. *Journal of Virological Methods*, 186(1–2), 119–125. https://doi. org/10.1016/j.jviromet.2012.08.002
- Filaire, F., Lebre, L., Foret-Lucas, C., Vergne, T., Daniel, P., Lelièvre, A., de Barros, A., Jbenyeni, A., Bolon, P., Paul, M., Croville, G., & Guérin, J. L. (2022). Highly pathogenic avian influenza a(H5N8) clade 2.3.4.4b virus in dust samples from poultry farms, France, 2021. *Emerging Infectious Diseases, 28*(7), 1446–1450. https://doi.org/10.3201/eid2807.212247

- Germeraad, E. A., Velkers, F. C., de Jong, M. C. M., Gonzales, J. L., de Wit, J. J., Stegeman, J. A., & Beerens, N. (2023). Transmissiestudie met vier vaccins tegen H5N1 hoogpathogeen vogelgriepvirus (clade 2.3.4.4b). Wageningen Bioveterinary Research.
- Gonzales, J. L., Boender, G. J., Elbers, A. R., Stegeman, J. A., & de Koeijer, A. A. (2014). Risk based surveillance for early detection of low pathogenic avian influenza outbreaks in layer chickens. *Preventive Veterinary Medicine*, *117*(1), 251–259. https://doi.org/10.1016/j.prevetmed.2014.08.015
- Gonzales, J. L., Elbers, A. R., van der Goot, J. A., Bontje, D., Koch, G., de Wit, J. J., & Stegeman, J. A. (2012). Using egg production data to quantify withinflock transmission of low pathogenic avian influenza virus in commercial layer chickens. *Preventive Veterinary Medicine*, *107*(3–4), 253–259. https:// doi.org/10.1016/j.prevetmed.2012.06.010
- Gonzales, J. L., & Elbers, A. R. W. (2018). Effective thresholds for reporting suspicions and improve early detection of avian influenza outbreaks in layer chickens. *Scientific Reports*, 8(1), 8533. https://doi.org/10.1038/s41598-018-26954-9
- Gonzales, J. L., Hennen, W. H. G. J., Petie, R., de Freitas, C. E., Beerens, N., Slaterus, R., Kuiken, T., Stahl, J., & Elbers, A. R. W. (2022). *Risicofactoren voor introductie van HPAI-virus op Nederlandse commerciële pluimveebedrijven, 2014–2022*. Wageningen Bioveterinary Research.
- Gonzales, J. L., Hobbelen, P. H. F., Spek, A. N., Vries, E. P., & Elbers, A. R. W. (2023). Highly pathogenic avian influenza a H5 virus outbreaks in broiler farms in the Netherlands Clinical signs, transmission and identification of reporting thresholds. *bioRxiv*:2023.2001.2005.522008. https://doi.org/10. 1101/2023.01.05.522008
- Graat, E. A., de Jong, M. C., Frankena, K., & Franken, P. (2001). Modelling the effect of surveillance programmes on spread of bovine herpesvirus 1 between certified cattle herds. *Veterinary Microbiology*, *79*(3), 193–208. https://doi.org/10.1016/s0378-1135(00)00356-4
- Grasland, B., Schmitz, A., Niqueux, E., Busson, R., Morin, N., Guillemoto, C., Orosco, A., Souchaud, F., Bougeard, S., Briand, F. X., Martenot, C., Cherbonnel, M., Massin, P., Rose, N., Louboutin, K., Pierre, I., Amelot, M., Delpont, M., Pouvelle, L., ... Eterradossi, N. (2023). Expérimentation de vaccination des canards mulards en élevage contre un virus influenza aviaire hautement pathogène A(H5N1) clade 2.3.4.4b. Rapport intermédiaire: Evaluation expérimentale de la protection clinique et de l'excrétion virale, ANSES, ENVT.
- Harder, T., de Wit, S., Gonzales, J. L., Ho, J. H. P., Mulatti, P., Prajitno, T. Y., & Stegeman, A. (2023). Epidemiology-driven approaches to surveillance in HPAI-vaccinated poultry flocks aiming to demonstrate freedom from circulating HPAIV. *Biologicals*, *83*, 101694. https://doi.org/10.1016/j.biologicals.2023.101694
 Hobbelen, P. H. F., Elbers, A. R. W., Werkman, M., Koch, G., Velkers, F. C., Stegeman, A., & Hagenaars, T. J. (2020). Estimating the introduction time of highly
- pathogenic avian influenza into poultry flocks. Scientific Reports, 10(1), 12388. https://doi.org/10.1038/s41598-020-68623-w
- Hoinville, L. J., Alban, L., Drewe, J. A., Gibbens, J. C., Gustafson, L., Häsler, B., Saegerman, C., Salman, M., & Stärk, K. D. (2013). Proposed terms and concepts for describing and evaluating animal-health surveillance systems. *Preventive Veterinary Medicine*, *112*(1–2), 1–12. https://doi.org/10.1016/j.preve tmed.2013.06.006
- Huynh, H. T. T., Truong, L. T., Meeyam, T., Le, H. T., & Punyapornwithaya, V. (2019). Individual and flock immunity responses of naïve ducks on smallholder farms after vaccination with H5N1 avian influenza vaccine: A study in a province of the Mekong Delta, Vietnam. *PeerJ*, 7, e6268. https://doi.org/10. 7717/peerJ.6268

Keeling, M. J., & Rohani, P. (2008). Modeling infectious diseases in humans and animals. Princeton University Press.

- Klement, E., Broglia, A., Antoniou, S. E., Tsiamadis, V., Plevraki, E., Petrović, T., Polaček, V., Debeljak, Z., Miteva, A., Alexandrov, T., Marojevic, D., Pite, L., Kondratenko, V., Atanasov, Z., Gubbins, S., Stegeman, A., & Abrahantes, J. C. (2020). Neethling vaccine proved highly effective in controlling lumpy skin disease epidemics in the Balkans. *Preventive Veterinary Medicine*, *181*, 104595. https://doi.org/10.1016/j.prevetmed.2018.12.001
- Ladman, B. S., Spackman, E., & Gelb, J. (2012). Comparison of pooling 11 or 5 oropharyngeal swabbings for detecting avian influenza virus by real-time reverse transcription-PCR in broiler chickens. Avian Diseases, 56(1), 227–229. https://doi.org/10.1637/9839-062011-ResNote.1
- Martin, P. A., Cameron, A. R., Barfod, K., Sergeant, E. S., & Greiner, M. (2007). Demonstrating freedom from disease using multiple complex data sources 2: Case study Classical swine fever in Denmark. *Preventive Veterinary Medicine*, 79(2–4), 98–115. https://doi.org/10.1016/j.prevetmed.2006.09.007
- Martin, P. A., Cameron, A. R., & Greiner, M. (2007). Demonstrating freedom from disease using multiple complex data sources 1: A new methodology based on scenario trees. *Preventive Veterinary Medicine*, 79(2–4), 71–97. https://doi.org/10.1016/j.prevetmed.2006.09.008
- Mo, J., Spackman, E., & Swayne, D. E. (2023). Prediction of highly pathogenic avian influenza vaccine efficacy in chickens by comparison of in vitro and in vivo data: A meta-analysis and systematic review. *Vaccine*, *41*(38), 5507–5517. https://doi.org/10.1016/j.vaccine.2023.07.076
- Muñoz-Aguayo, J., Flores-Figueroa, C., VanBeusekom, E., McComb, B., Wileman, B., Anderson, J., Halvorson, D. A., Kromm, M., Lauer, D., Marusak, R., Nezworski, J., Voss, S., & Cardona, C. (2019). Environmental sampling for influenza a viruses in Turkey barns. *Avian Diseases*, *63*(1), 17–23. https://doi. org/10.1637/11892-050418-Reg.1
- R Core Team. (2018). R: A language and environment for statistical computing. R Foundation for Statistical Computing. https://www.R-project.org
- Souvestre, M., Guinat, C., Niqueux, E., Robertet, L., Croville, G., Paul, M., Schmitz, A., Bronner, A., Eterradossi, N., & Guerin, J. L. (2019). Role of backyard flocks in transmission dynamics of highly pathogenic avian influenza A(H5N8) clade 2.3.4.4, France, 2016–2017. *Emerging Infectious Diseases, 25*, 551–554. https://doi.org/10.3201/eid2503.181040
- Ssematimba, A., Malladi, S., Hagenaars, T. J., Bonney, P. J., Weaver, J. T., Patyk, K. A., Spackman, E., Halvorson, D. A., & Cardona, C. J. (2019). Estimating within-flock transmission rate parameter for H5N2 highly pathogenic avian influenza virus in Minnesota Turkey flocks during the 2015 epizootic. *Epidemiology and Infection*, 147, e179. https://doi.org/10.1017/S0950268819000633

Suarez, D. L. (2005). Overview of avian influenza DIVA test strategies. Biologicals, 33(4), 221–226. https://doi.org/10.1016/j.biologicals.2005.08.003

- Tarigan, S., Wibowo, M. H., Indriani, R., Sumarningsih, S., Artanto, S., Idris, S., Durr, P. A., Asmara, W., Ebrahimie, E., Stevenson, M. A., & Ignjatovic, J. (2018). Field effectiveness of highly pathogenic avian influenza H5N1 vaccination in commercial layers in Indonesia. *PLoS One*, *13*(1), e0190947. https://doi. org/10.1371/journal.pone.0190947
- Tatár-Kis, T., Dán, Á., Felföldi, B., Bálint, Á., Rónai, Z., Dauphin, G., Pénzes, Z., El-Attrache, J., Gardin, Y., & Palya, V. (2019). Virus-like particle based vaccine provides high level of protection against homologous H5N8 HPAIV challenge in mule and Pekin duck, including prevention of transmission. *Avian Diseases*, 63(Suppl. 1), 193–202. https://doi.org/10.1637/11882-042718-Reg.1
- Widgren, S., Bauer, P., Eriksson, R., & Engblom, S. (2019). SimInf: An R package for data-driven stochastic disease spread simulations. *Journal of Statistical Software*, *91*(12), 1–42. https://doi.org/10.18637/jss.v091.i12
- WOAH (World Organisation for Animal Health). (online-a). Terrestrial animal health code. https://www.woah.org/en/what-we-do/standards/codes-andmanuals/terrestrial-code-online-access/?id=169&L=1&htmfile=sommaire.htm
- WOAH (World Organisation for Animal Health). (online-b). Manual of diagnostic tests and vaccines for terrestrial animals. (12th ed.). https://www.woah. org/fileadmin/Home/eng/Health_standards/tahm/A_summry.htm

How to cite this article: EFSA AHAW Panel (EFSA Panel on Animal Health and Animal Welfare), European Union Reference Laboratory for Avian Influenza, Nielsen, S. S., Alvarez, J., Bicout, D. J., Calistri, P., Canali, E., Drewe, J. A., Garin-Bastuji, B., Gortázar, C., Herskin, M. S., Michel, V., Miranda Chueca, M. Á., Padalino, B., Roberts, H. C., Spoolder, H., Stahl, K., Velarde, A., ... Gonzales Rojas, J. L. (2024). Vaccination of poultry against highly pathogenic avian influenza – Part 2. Surveillance and mitigation measures. *EFSA Journal*, *22*(4), e8755. <u>https://doi.org/10.2903/j.efsa.2024.8755</u>

APPENDIX A

Mapping of ToRs 3 and 4 and Opinion's output

TABLE A1 Term of references (ToR) as they were received from the European Commission and the assessment methods and results relative to each ToR or sub-ToR.

ToR	Sub-ToR	Assessment method	Assessment results
3. Surveillance in the vaccinated zone and/or vaccinated establishments	 3.1. Assess the suitability and effectiveness of the reinforced surveillance set out in Part 2 and in point 2 of Part 5 of HPAI Annex to the Delegated Regulation (EU) No 2023/361 to early detect infection in the vaccinated flocks and to prevent spread of the HPAI virus by movement of birds and their products from vaccinated establishments/flocks. The above assessment should include scenarios depending on different vaccination strategies. 	 By vaccination strategy: a. To early detect (Section 2.1.5.1) and demonstrate freedom from HPAIV during emergency vaccination: modelling HPAI infection dynamic in vaccinated flocks b. To early detect and demonstrate freedom from HPAIV during preventive vaccination (Section 2.1.5.2): scenario's tree model 	 By vaccination strategy (emergency protective/ preventive): a. R_s values and time between introduction and detection based on virology (prev infectious), serology (prev recovered), mortality (prev dead), given different sampling scheme (frequency and size) (Section 3.1.2.1) b. Sensitivity of HPAIV early detection given different sampling schemes (frequency and proportion of vaccinated establishments under surveillance) and confidence of freedom (Section 3.1.2.3)
	 3.2. Taking into account the possible different surveillance approaches (e.g. serology, virology, conventional, DIVA, passive/active surveillance of vaccinated birds, use of sentinels) explore and provide for alternative suitable surveillance approaches/strategies indicating the minimum level and duration of surveillance required in a vaccinated establishment, including sampling schemes and testing procedures, to: a. ensure early detection of infection with HPAI viruses; b. be implemented as additional guarantees to authorise the movement of vaccinated and non-vaccinated poultry and poultry products within the vaccinated zone and from the vaccinated zone or establishment; c. be implemented after cessation of vaccination as necessary risk mitigation measure to authorise movement of 	 By vaccination strategy (emergency protective/preventive): a. To early detect (Section 2.1.5.1) and demonstrate freedom from HPAIV during emergency vaccination: modelling HPAI infection dynamic into vaccinated flocks b. To early detect and demonstrate freedom from HPAIV during preventive vaccination (Section 2.1.5.2): scenario's tree model c. Qualitative assessment d. Descriptive analysis of diagnostic methods available for surveillance of HPAIV vaccinated birds (Section 2.1.4) 	 By vaccination strategy (emergency protective/ preventive): a. Rs values and time between introduction and detection based on virology (prev infectious), serology (prev recovered), mortality (prev dead), given different sampling schemes (frequency and size) (Section 3.1.2.1) b. Sensitivity of HPAIV early detection given different sampling schemes (frequency and proportion of vaccinated establishments under surveillance) and confidence of freedom (Section 3.1.2.3) c. Timespan of surveillance to be applied after cessation of vaccination (Section 3.1.2.6) d. Indication of appropriate diagnostic methods tailored to the surveillance objectives and sample type (Section 3.1.1)
	 birds and their products from those establishments, taking into account: the relevant factors to influence the level and duration of protection following vaccination with different type of vaccines, simultaneous presence in the establishment of vaccinated and unvaccinated animals. 	Description of vaccination effectiveness (Article 9) based on relative risk measures; assessment of level of immune response induced by vaccination (Section 2.1.5.3)	Indication of options and data needed to assess vaccination effectiveness (Section 3.1.2.4) and level of immune response induced by vaccination (Section 3.1.2.5)
	3.3. Taking into account the WOAH standards (i.e. surveillance to be carried out in all vaccinated flocks) explore and provide for alternative suitable surveillance approaches/ strategies to be implemented in a vaccination zone to demonstrate freedom from HPAI based on representative sampling of the vaccinated (and not vaccinated) establishments within a vaccination zone.	For preventive vaccination: b. Scenario's tree model (Section 2.1.5.2)	 b. Confidence of freedom given different sampling schemes considering different proportion of vaccinated establishments under surveillance (Section 3.1.2.3)

18

20

(Continues)

TABLE A1 (Continued)

ToR	Sub-ToR	Assessment method	Assessment results
4. Restrictions and risk mitigation measures to be applied in a vaccinated establishment or a vaccination zone	 4.1. Assess the suitability of the restrictions and risk mitigation measures set out in Part 3 and in points 3 to 5 of Part 5 of HPAI Annex to the Delegated Regulation (EU) No 2023/361 to prevent the spread of HPAI viruses, enabling safe movement of poultry and their products following emergency and preventive vaccination, respectively. 4.2. Explore and provide for alternative (to those referred in 4.1.) suitable movement restrictions and risk mitigation measures required to prevent the spread of virus by movement of birds or their products from a vaccinated establishment/vaccination zone. 	By vaccination strategy (emergency protective/preventive) (Section 2.2.1): Qualitative assessment of measures currently in place and demonstration of freedom as additional/alternative measure via modelling HPAI infection dynamic into vaccinated flocks (emergency vaccination) and scenario's tree model (preventive vaccination)	By vaccination strategy (emergency protective/ preventive): confidence of freedom given different sampling schemes (frequency, size, proportion of vaccinated establishments under surveillance) (Section 3.1.2.3)



